

Assessing Parental Capacity to Change when Children are on the Edge of Care: an overview of current research evidence

Research report

June 2014

Harriet Ward, Rebecca Brown and Georgia Hyde-Dryden, Centre for Child and Family Research, Loughborough University

Contents

Contents	2
Table of Figures	8
Table of Tables	9
Acknowledgments	10
Note on Terminology	10
Executive Summary	11
Introduction	11
Key findings	11
Background	13
Aim	13
Methodology	14
Findings	14
The impact of risk and protective factors on parenting capability and capacity to change	14
The potential impact on children of abuse and neglect	15
Informing structured professional judgments using standardised measures	15
Useful conceptual models to facilitate analysis and case conceptualisation	16
Elements requiring consideration within the change process: resistance, motivation engagement and relapse	n, 17
Understanding effective interventions – parenting skills, specific and multifaceted interventions	18
The timing of the change process	19
The implications for social work practice	20
Conclusion	21
Chapter One: Introduction and methodology	22

Introduction	22
Children and Families Act 2014 and revised Public Law Outline	22
Assessing parental capacity to change within a wider context	24
Societal factors and their influence on parenting	24
Interplay of factors	24
Social work assessments in the real world	25
Availability and interpretation of information	26
Factors that skew the decision-making process	27
Checks and balances	29
Purpose of this literature review	32
Methodology	32
Conclusion	34
Key points from Chapter One	35
Chapter Two: Risks of Future Harm	36
Introduction	36
Factors which undermine parenting capability	37
Parents' problems	37
Impact of combinations of problems	39
Environmental factors and their impact	39
Parenting children with additional needs	40
Recurrent abuse	40
Protective factors	41
The impact of abuse and neglect on children's subsequent development	44
Attachment styles and their impact	44
The impact of abuse and neglect as children grow	45

Identifying children suffering, or likely to suffer significant harm	49
Tools to assess risk of significant harm	50
Conclusion	52
Key points from Chapter Two	52
Chapter Three: Models of change: their strengths and weaknesses	54
Introduction	54
Case conceptualisation	54
Parents' attachment styles and their implications	55
Models of change	56
The Trans-Theoretical Model of Change (TTM)	56
Adaptations to the Trans-Theoretical Model and alternative models	61
A procedure for assessing parents' capacity to change (Harnett 2007)	62
Conclusion	66
Key points from Chapter Three	66
Chapter Four: Ambivalence, denial and unwillingness to change	68
Introduction	68
How common is denial or unwillingness to change?	68
Disguised compliance, failure to cooperate, and unwillingness to change	69
Factors underlying unwillingness to change	70
Internal factors	71
External factors	73
Support from other agencies	74
Motivational Interviewing: strengths and weaknesses	75
The Family Partnership (FP) Model	77
Family Group Decision-Making	78

Engagement and satisfaction	79
Children's participation and outcomes	80
Engagement of professionals	81
When parents cannot change within a child's timeframe	82
Conclusion	82
Key points from Chapter Four	83
Chapter Five: Motivation and engagement	84
Introduction	84
Motivation: the decisional balance	84
Turning points	86
Engagement	94
Models of engagement	95
Conclusion	98
Key points from Chapter Five	98
Chapter Six: Building Parenting Capability through Evidence-based Interventions	100
Introduction	100
Understanding effective services and interventions	101
General principles concerning evidence of effectiveness	101
Understanding effective social work	104
Barriers to effective social work	105
Intensive Family Preservation Services (IFPS)	106
Outcomes of care	108
Outcomes of kinship care	109
Outcomes of adoption	110
Returning home from care	111

Specific interventions to support parents and children	111
Targeted preventive programmes	112
Specific programmes to address abusive parenting and prevent impairment	113
Specific interventions to address parental problems that increase the risk of maltreatment	116
Intensive, multi-faceted interventions for families with complex needs	121
Conclusion	124
Key points from Chapter Six	124
Chapter Seven: Relapse and the maintenance of change	126
Introduction	126
The impact of different types of problem on sustained change	127
How long is needed to establish whether parents will successfully maintain cha	ange?127
The process of recovery and relapse	128
Virtuous circles	129
Factors supporting or undermining the maintenance of change	129
Examples of factors supporting change	130
Examples of factors undermining change	131
The role of parenthood in supporting and undermining change	132
Relapse and harm to the child	133
Conclusion	134
Key points from Chapter Seven	134
Chapter Eight: Conclusion: Making use of the research findings to inform profess practice and assessments required by the courts	sional 136
Introduction	136
Assessment of parents' capacity to meet needs (parenting capability)	137
Assessments of children's needs	138

Assessing the likelihood of significant harm	139
Analysis of why there might be a gap between parental capability and the child's	needs139
Case conceptualisation	139
Gaps between parenting capability and children's needs	140
Implications for day to day practice	140
Court requirements and implications for social work practice	140
Understanding the process of change	141
Relationships between social workers and parents	143
Relationships with other professionals	144
Assessing parents' capacity to change within a child's timeframe	145
Implications for continuing professional development	147
Conclusion	148
Key Points from Chapter Eight	148
Appendices	150
Appendix One: search terms	150
Table 1: search terms	151
Appendix Two: scientific advisory group and steering group members	154
References	155
List of Cases	192

Table of Figures

Figure 1.1: The Assessment Framework	25
9	_

Figure 5.1: An Integrated Model of Parental Engagement with Child Welfare Services 97

Table of Tables

Table 2.1: Factors associated with future harm	42
Table 2.2: Impact of abuse and neglect during key developmental timeframes for	or children 47
Table 6.1: Hierarchy of levels of evidence	103

Acknowledgments

This literature review was funded by the Department for Education as part of the Childhood Wellbeing Research Centre programme. The project team would like to express their gratitude for the invaluable advice and expertise provided by the scientific advisory group in reviewing the findings of the literature search and advising on the content of the draft chapters: Jane Barlow, Paul Harnett, Judith Harwin, Judith Masson, Duncan McLean, Dendy Platt, and Nicky Stanley. They would like to thank the members of the steering group for their wise counsel and advice provided throughout and for reviewing the draft report: Chris Cuthbert, Jonathan Dickens, Judith Freedman, Liz Gillett, Bridget Lindley, Brendah Malahleka, Judge Lesley Newton, Helen Pustam and Steve Walker. The authors are also grateful for the advice and support in overseeing the project provided by staff in the Department for Education: Louise Bridson, Jessica Dunn, Nicola Gilbert, Catherine North and Richard White.

Note on Terminology

'Parenting capacity' is a phrase deriving from the Children Act 1989 (s.1.3), that refers to the question of whether or not parents are capable of meeting their children's needs (Department of Health, 1989). 'Parental capacity to (or for) change' is the established terminology for describing parents' willingness and ability to overcome risk factors such as alcoholism and domestic abuse that increase the likelihood of abuse and neglect and improve their parenting practices. The two phrases can be confused with one another. There is movement to replace the term 'parenting capacity' with 'parenting capability' to avoid such confusion (see Department for Education, 2014a), and for this reason we have used this new terminology in this report.

Some of the issues covered by this report are difficult to express without appearing to denigrate already vulnerable population groups. Throughout, we have been mindful of the need to use non-discriminatory language; terms that we have used, such as 'denial', 'resistant to change' or 'perpetrators of abuse' are modelled on language commonly used in practice. They may differ from the terms used in other arenas such as within organisations supporting service users or promoting policy issues.

Executive Summary

Introduction

Assessing Parental Capacity to Change when Children are on the Edge of Care is an overview of current research evidence, bringing together some of the key research messages concerning factors which promote or inhibit parental capacity to change in families where there are significant child protection concerns. It is intended to serve as a reference resource for social workers in their work to support families where children's safety and developmental functioning are at risk. Its purpose is also to assist social workers and children's guardians in delivering more focused and robust assessments of parenting capability and parental capacity to change, and assist judges and other legal professionals in evaluating the quality of assessment work in court proceedings. The report brings together research findings from a wide range of disciplines, which are not otherwise readily available in one location for social workers, family justice professionals and other practitioners with safeguarding responsibilities.

The research evidence covered in this report confirms that change is both important and necessary when children are suffering abuse and neglect. However it also makes it clear that change is difficult for everyone, but even harder for those parents who are struggling with an interlocking web of problems. It also takes time. Change is a complex process, and although it can be supported and promoted through effective interagency interventions, it cannot be imposed. It will not happen unless parents are proactively engaged. These are the key messages from the review.

Key findings

- An extensive body of evidence shows how factors such as domestic abuse, substance misuse, mental health problems and learning disability undermine parenting capability and increase the likelihood of significant harm, particularly when they occur in combination. Moreover, parenting does not take place in isolation. Parents are also influenced by stressors within the wider environment and family, such as poor housing, poverty and unemployment that make parenting more challenging and increase the likelihood that difficulties will arise.
- There is an extensive and growing evidence base showing how the experience of abuse and neglect may have a long-term, negative impact on children's physical, cognitive, social, emotional and behavioural development that can last throughout the life course.
- The reliability of practitioners' judgments concerning the risks of significant harm could be improved. Judgments based on experience and intuitive thinking should be supported, but not replaced, by information collected through evidence-based tools and standardised measures to inform structured professional decisionmaking. However, it is important that all tools and measures selected are

- validated within a UK context for use with families with complex problems and needs.
- In order to carry out an accurate analysis when assessing parenting capability and capacity to change, social workers need to formulate a case conceptualisation, setting out the various internal and external factors that influence parents' ability to meet their children's needs. Conceptual models are necessary to provide social workers with a framework to analyse the circumstances of each case and assess the likelihood of change.
- The assessment of parenting capability and capacity to change needs to reflect
 the complex reality of child protection cases, including consideration of the
 individual challenges and wider environmental problems faced by families; how
 multiple problems interlock; and the potential impact of factors such as coercion or
 the pressure on parents to present themselves in a positive light.
- Assessment of parental capacity to change should be undertaken as a dynamic process in which strengths and weaknesses are identified, targets set and agreed, effective interventions identified and implemented and progress monitored over a specific time period.
- The process of behaviour change is well-established, incorporating a number of common elements including resistance, ambivalence, motivation, engagement and action. Lapse or relapse is also viewed as an integral part of the change process.
- False compliance, failure to cooperate and denial are common features of parents involved in the child protection process. Apparent resistance may be the result of fear, stigma, shame, ambivalence, or a parent's lack of confidence in their ability to change. Parents may be resistant to the involvement of social workers rather than resistant to change in itself, particularly where they feel social workers are exercising power *over them* instead of *with them* in a supportive manner. However parental resistance is not necessarily indicative of a lack of skilled social work.
- The Family Partnership Model (FP), Motivational Interviewing (MI) and Family Group Decision-Making (FGDM) offer potential methods of engaging parents who are ambivalent about change, mistrustful of social workers, or not fully ready for change. Such methods can empower parents by giving them an element of control. FGDM also enables families to participate in the decision-making process.
- Social workers need to be mindful that parents may turn to other professionals, who are seen as less threatening, or whose involvement attracts less stigma. This is a strong reason for social workers to develop good inter-agency links with other professionals such as health visitors and practitioners in adult services to maximise engagement with parents.
- Some interventions are effective and others can be harmful. It is important to understand how to interpret evidence of effectiveness, not all of which is of the same quality.

- An increasing range of effective intensive interventions aimed at improving parenting skills or addressing other specific problems, for instance, drug or alcohol misuse, can complement social work support to a family.
- Interventions designed to increase parenting skills can be effective and can have a
 positive knock on impact, reducing other parental problems by increasing selfefficacy and self-esteem. However, in cases where parents are facing complex,
 multi-layered problems, an integrated package of support may be required,
 tailored to meet the needs of each member of the family.
- Many standardised measures and intensive programmes are still relatively new.
 They may well prove to be effective but many have not yet been adequately validated in the UK and are not available in all areas of the country.
- Finally, interventions take time and change may not always be possible within the child's timeframe, particularly where children are very young or vulnerable, entrenched parental behaviour patterns need to be addressed, progress is slow and relapse is frequent.

Background

When there are serious child protection concerns and children are on the edge of care, the most difficult decisions facing social workers concern the capacity of parents to change. Social workers need to be able to make informed decisions about which parents are unable to meet their children's needs and why, what aspects of parents' behaviour need to change and whether parents have the capacity to make such changes within a timeframe that is appropriate for the child. Decisions will be informed by the social worker's understanding of the likely impact of continuing abuse and neglect on children's life chances and by their assessments of parents' motivation and ability to make and sustain changes to behaviour patterns that compromise their children's development. Parenting does not take place in isolation: social workers must also consider the specific needs of the child about whom there are concerns, and take wider environmental factors such as poverty and inadequate housing into account when making their assessments. They also need to consider the availability and effectiveness of interventions designed to support parents through the change process.

Aim

The aim of the overview is to:

- serve as a reference resource to social workers in their work to support families where children's safety and developmental functioning are at risk; and
- assist social workers and children's guardians in delivering focussed and robust assessments of parenting capability and parental capacity to change and assist judges and other legal professionals in evaluating the quality of assessment work in court proceedings.

Methodology

The methodology employed in this review incorporated a search of the peer reviewed literature supported by expert advice. The findings of the literature search were reviewed by a scientific advisory group made up of academics specialising in areas relevant to parental capacity to change, who also advised on specific issues. The content of the draft report was reviewed by the scientific advisers and by members of a steering group consisting of academics and professionals also working in related fields. The overview is not the report of a systematic review, but of an extensive narrative literature review that draws together information from a wide range of sources.

Findings

The impact of risk and protective factors on parenting capability and capacity to change

There is an extensive body of evidence that shows how parental problems such as domestic abuse, substance misuse, mental health problems and learning disability can undermine parenting capability and increase the likelihood of significant harm to children. Parents' unacknowledged experiences of abuse in their own childhoods can also increase the risk that children will be exposed to maltreatment. However, experiencing any one of these problems does not preclude loving and effective parenting. The research suggests that it is where multiple problems interlock and interact that there is a substantially increased risk that children will be exposed to maltreatment and suffer significant harm.

Parents' problems may reduce parental capability in a number of ways: for instance, parents with substance misuse or alcohol problems may use their resources to fund their addiction and have little left over to feed, clothe and house their children; they may also be too preoccupied with their addiction and too much under its influence to be aware that their children's basic needs are not being met (Fraser et al., 2009). Domestic abuse may directly expose children to physical danger and also have a negative impact on their sense of safety and security (Buckley et al., 2007). Impaired personality functioning and some mental health problems such as anxiety disorders, depression and some psychotic illnesses may reduce parents' ability to be reciprocal, involved and encouraging with their children and offer them sufficient emotional warmth (Kohl et al., 2011; Mowbray et al., 2002). Substance misuse and domestic abuse, as well as depression, may engender apathy and listlessness and reduce parents' ability to stimulate their children; such problems are also associated with low self-esteem and unpredictable, inconsistent and ineffective parenting behaviours, reducing parents' ability to provide authoritative guidance and set boundaries. Moreover these parental problems are also all associated with frequent changes of partner and living arrangements, reducing a parent's ability to offer children stability and security. It is also clear that parental problems do not occur in isolation: they are influenced by stressors within the wider environment and family, such

as poor housing, poverty and unemployment that magnify the challenges to parenting and increase the likelihood that problems will arise.

Caring for a child with additional needs can increase parents' stress levels and escalate other problems (Baker *et al.*, 2003). Children with disabilities and health care needs, and younger children are all more likely than others to experience abuse and neglect.

Once abuse has occurred there is a strong possibility of recurrence: 14.7% of children experience further maltreatment within six months of the first episode; 22.6% within eighteen months (Fluke *et al.*, 1999). Factors which have been shown to reduce the likelihood of significant harm include the presence in the household of a non-abusive partner; parents' recognition of their problems and willingness to take responsibility; their engagement with services; and their empathy for the child (see Hindley, Ramchandani and Jones, 2006).

The potential impact on children of abuse and neglect

There is a growing body of research evidence from a wide range of disciplines showing that abuse and neglect are likely to have long-term adverse consequences across children's physical, cognitive, social, emotional and behavioural development (see for instance Norman *et al.*, 2012). Maltreatment has a negative impact at all ages, but is thought to be particularly pernicious in the early years because of the impact on the child's neurobiological development and the attachment process (see for instance Howe, 2006; McCrory, De Brito and Viding, 2012). The negative impact of abuse and neglect has also been shown to continue throughout childhood and adolescence, and into adulthood (see for instance Romer, 2010; Springer *et al.*, 2007). A number of protective factors within the child, the family and the environment may mitigate the impact of abuse and neglect; however the longer maltreatment persists and the more intensive it is, the harder it will be to overcome the consequences (Rutter *et al.*, 2007). Change is therefore necessary in families where abuse and neglect are evident.

Informing structured professional judgments using standardised measures

When social workers assess the risk of harm to children, research evidence suggests that the reliability of professional judgment alone could be improved (Munro, 1999). There is increasing recognition of the need for 'structured professional decision-making' which utilises data collected through evidence-based tools in addition, but not instead of, judgments that can be over-reliant on social workers' intuition and experience (Barlow, Fisher and Jones, 2012). A number of standardised measures have consequently been developed to support professional decision-making in this area. Barlow, Fisher and Jones (2012) undertook a systematic review of tools available to social workers to assist them in deciding whether a child is suffering, or likely to suffer, significant harm. Although the review highlights the need for further piloting and validating of tools within a UK setting,

such tools have the potential to provide practitioners with standardised measures that can support the development of structured professional judgment.

Useful conceptual models to facilitate analysis and case conceptualisation

A number of conceptual models can provide social workers with a useful framework for analysing a family's circumstances and the factors influencing parents' capacity to change in order to develop a case conceptualisation. This is a necessary element in the process of identifying what needs to change and formulating appropriate plans. The different domains of the Assessment Triangle (HM Government, 2013) provide a conceptual map that helps practitioners analyse the strengths and weaknesses in a family and their environment and identify the issues that need to be addressed.

In families where parents' poor personality functioning is an issue, social workers will also need a conceptual framework of the elements of mature adult functioning that can offer secure and responsive parenting. They will also need to be mindful that parents' attachment styles may reflect early childhood trauma and may present an obstacle in assessing and working with families.

A number of conceptual models have been developed to facilitate understanding of behavioural change, the most prominent of these being Prochaska and DiClemente's Trans-Theoretical Model of Change (TTM), incorporating the stages of change (SOC) (see for instance DiClemente and Prochaska, 1982; Prochaska and Prochaska, 2002). TTM incorporates six stages of change: *pre-contemplation, contemplation, preparation, action, maintenance and relapse.* Change is conceptualised as a spiral pattern where people can progress through the stages, but are likely to encounter relapses in which they will regress to an earlier position. The model was originally developed to understand smoking cessation, but has been used more widely including in a child welfare context. However, while it is useful for social workers to have an understanding of the key concepts of the model, they also need to be aware of a number of concerns that have been raised about its applicability to child welfare cases, and about the predictive validity of some of the accompanying materials.

Harnett (2007) offers an example of an alternative model for understanding parents' capacity to change. Rather than relying solely upon a cross-sectional assessment, Harnett's model incorporates assessments of parents' and children's current functioning; specification of targets for change derived from an assessment of current strengths and deficits in the family; implementation of an intervention with proven efficacy for the client group with a focus on achieving clearly specified targets; and objective assessment of measurement of changes in parenting.

Elements requiring consideration within the change process: resistance, motivation, engagement and relapse

The research evidence highlights how social workers need to understand the various elements within the change process, in order to be able to identify where and why parents may face particular obstacles or have certain reactions at different points in time. Real change cannot be imposed on parents, so this understanding will help social workers to support parents and find ways of moving forward. Examples of the elements of the change process that social workers need to understand include unwillingness, ambivalence, motivation, engagement and relapse.

Parents with learning disabilities and/or mental health problems including impaired personality functioning may not understand the impact of neglectful or abusive behaviour on children's welfare. However parents may also deny that there is a problem and appear unwilling to change. Such apparent 'resistance to change' can reflect internal factors such as shame, ambivalence about the need to change, and parents' lack of confidence in their capacity to change. The context of child protection work is in itself likely to create resistance, which may be a response to social worker involvement in the family rather than to change itself (Forrester *et al.*, 2012). Social workers need to be aware of their own role in exacerbating or reducing resistance, particularly through the way in which they approach parents. Confrontational styles and the heavy handed use of authority can lead to both parent and social worker focusing on a power struggle rather than the issues to be addressed (Dumbrill, 2006).

The Family Partnership Model attempts to address the difficulties that practitioners encounter in developing relationships under adverse circumstances, such as when parents are hostile or frightened that their children might be removed. The model was developed to 'provide practitioners working with families with an explicit and detailed understanding of the dynamic processes of helping' (Harnett and Day, 2008, p.81). Research evidence also points to the use of Motivational Interviewing to reduce unnecessary confrontation and help parents move on from a position where they deny that a problem exists and are unwilling to change or engage with services (see Bowen and Gilchrist, 2004; Forrester *et al.*, 2012).

There is considerable evidence to suggest that approaches such as Family Group Decision-Making (FGDM) involving parents and their wider extended family in the decision-making processes can decrease parental resistance to involvement with social workers by reducing their feelings of powerlessness within the context of statutory interventions and court proceedings. Although there is good evidence that FGDM is viewed positively, further research is required with regard to the implementation of plans and outcomes for the children concerned (Barnsdale and Walker, 2007).

Parents become motivated to change for a number of reasons, many of which are unclear. However it does appear that motivation begins when the perceived advantages of changing the *status quo* outweigh the perceived disadvantages, thereby tipping the

'decisional balance' (Prochaska *et al.*, 1994). Events or changes in circumstances such as pregnancy or referral to children's social services can also create turning points in parents' lives that motivate them to make the changes needed to overcome adverse behaviour patterns and improve their parenting. Turning points should be viewed by social workers as opportunities to engage with parents and offer appropriate ongoing support.

Children's views of the situation can also act as a powerful cue for action (Hahn *et al.*, 1996; Stanley *et al.*, 2012a). On the other hand, coercion can push parents who were previously ambivalent about the *need* for change into action, while those who are doubtful of their *capacity* to change may become further entrenched in adverse behaviours. It is also possible that parents facing multiple problems may become motivated to change in one area, but may not necessarily appreciate the need for change in others (see DiClemente *et al.*, 2008).

Despite being motivated, there is no guarantee that parents who are ready to engage with services will succeed in making changes. There are a number of interlocking factors, including internal and external determinants and background issues which influence parental engagement. Using a theoretical model such as Platt's (2012) Integrated Model of Parental Engagement to explore the different factors that promote or inhibit parental engagement with child welfare services can facilitate a greater understanding of the issues that need to be addressed.

Lapse or relapse forms a natural part of the change and recovery process and should therefore be expected. Recovery from problems such as substance misuse is a gradual process and can extend over a period of years rather than being a time limited event. Parents' ability to sustain change in the long term will be affected by the type and number of difficulties they are trying to overcome, and whether these can be fully addressed or only controlled and alleviated. Sustained change will be supported by factors such as demography; self-efficacy; having a 'normal' role in society; having a positive support network; and appropriate ongoing support from professionals. Equally sustained change can be undermined, for instance, by stress, negative emotions, the co-existence of problems, isolation, inadequate support networks and poverty.

Understanding effective interventions – parenting skills, specific and multifaceted interventions

Being able to identify appropriate and effective interventions that can complement social work support is fundamental to supporting parents in changing their behaviours. In order to do this, social workers need to understand how different interventions operate and how their reported impact should be interpreted and understood. The quality and rigour of the evaluation process are important factors to consider in assessing the effectiveness of interventions. Ineffective interventions may not only fail to ensure that children are safe, they may also fail to offer parents adequate support in overcoming destructive behaviour

patterns and demonstrating capacity to change. Moreover some interventions are worse than ineffective in that they have been shown to be harmful (see Jarrett, 2008; Lilienfeld, 2007; Rhule, 2005). Interventions which may be effective in facilitating capacity to change include those designed to improve parenting skills and/or address specific problems such as substance misuse, domestic abuse or impaired personality functioning. However a multi-faceted approach that integrates a range of services for the whole family may prove to be a more effective means of increasing parental capability when parents are facing a complex web of long-term problems which are already impacting on parent-child relationships and children's development. The availability of specific, intensive interventions will vary across the country and provision will be delivered by professionals from a range of agencies, requiring close inter-agency working.

The timing of the change process

No one can predict how long it will take for abusive or neglectful parents to develop sufficient capacity to meet their children's needs. It is, however, evident that this is unlikely to happen overnight, that the process of change may be lengthy and that setbacks are common. We do not know how long it takes for parents with multiple interlocking problems to accept the need for change and to engage with services. However, once they become engaged, intensive interventions to address specific problems take several weeks to deliver, and are often followed by a further period of follow up. Relapse is also particularly common within the first six months of entering treatment. Detailed and dynamic assessment procedures, supported by standardised measures and clear conceptual models will provide social workers with the best indication of the rate at which parents are likely to make changes.

In some cases, an assessment of parenting capability and parents' capacity to change will lead professionals to conclude that sufficient change to reduce the likelihood of significant harm is not possible within the timeframe of the individual child and that a permanent placement away from home will best meet their needs. Of particular concern should be children living in homes where the following factors are present: extreme domestic abuse where the perpetrator shows a pervasive pattern of disregard for and violation of the rights of others (Gondolf, 2002; Scott, 2004); there is both substance misuse and domestic abuse and violence in the home Forrester and Harwin, 2008); children are not protected from perpetrators of sexual abuse; and/or where parents consciously and systematically cover up deliberate maltreatment (Brandon *et al.*, 2008). Maltreated children tend to do better in kinship or foster care or placed for adoption than those who remain with or return to abusive or neglectful birth parents (Biehal, 2010; Wade *et al.*, 2011; Farmer and Lutman, 2012).

The implications for social work practice

The messages from the literature review have clear implications for practitioners' assessments for the courts and for every day social work practice.

The findings have particular relevance for assessments of children's needs, assessments of parental capacity (or capability) and the analysis of why there might be a gap between parental capability and children's needs, all required by the courts. Such assessments should inform decisions to initiate proceedings, decisions in proceedings, and be an integral part of day to day social work with parents whose children are on the edge of care. The process of assessment should be dynamic rather than static, for the findings should form the basis for understanding the strengths and weaknesses in the family, agreeing what needs to change and setting goals and timescales.

Relationships matter in everyday social work practice. Many parents will not develop sufficient capability to meet their children's needs without support. If social workers can establish a good working relationship with parents characterised by honesty about what needs to change and why, sensitivity and a willingness to listen to parents' points of view, respectful uncertainty in the face of dissimulation and supportive use of power, they may be better able to help parents become motivated and engage with services (Forrester *et al.*, 2008a; Dumbrill, 2006; Laming, 2003). This relationship needs to continue after completion of an intervention to ensure that parents are provided with appropriate ongoing support to avoid unnecessary episodes of relapse into old behaviour patterns.

Good working relationships with professionals from other agencies are also of fundamental importance. Where parents are trying to overcome complex and entrenched problems, social work support needs to be complemented by other, more specific interventions, delivered by professionals from a wide range of agencies. Tensions between adult and children's services often exist, particularly when one professional focusses on the parent's needs and another on those of the child. It is important to resolve such tensions (Tompsett *et al.*, 2009; Ward, Brown and Westlake, 2012).

Once the risk of harm has been identified, social workers need to manage cases proactively to minimise the potential impacts of abuse and neglect (see Farmer and Lutman, 2012). Proactive case management needs to include early agreement about the goals to be met and the provision of effective services that will support parents in meeting them. What we know about likely timescales for overcoming destructive behaviour patterns and improving parenting capability makes it clear that this work needs to begin as early as possible after the risk of harm is identified in order to ensure that parents have the best opportunities to make necessary changes within a child's timeframe.

Conclusion

The overview of the literature highlights the complexity involved in assessing whether parents have the capacity to change within a child's timeframe. Many of the issues it raises will already be familiar to social workers and other practitioners. The review should strengthen practitioners' understanding of parental capacity to change by bringing together empirical evidence from a wide range of sources, highlighting what research tells us about the process of change and by introducing theoretical models available to social workers to underpin the assessment and decision-making process.

Chapter One: Introduction and methodology

Introduction

In families where there are significant child protection issues, well-informed social work assessments concerning the ability of parents to keep their children safe and meet their developmental needs are essential to making appropriate decisions about whether children can be safely supported at home; whether they should be looked after by the local authority while parents increase their ability to respond to their needs; or whether they should be permanently placed away from home.

High levels of uncertainty and anxiety run through decision-making when there are significant child protection concerns. A better understanding of the lived experience of parents whose children are on the edge of care, together with knowledge of those external and internal factors that influence and inhibit parental change, will help social workers provide more focussed support for families to assist them to meet the needs of their children. Where children cannot safely remain at home, accurate and relevant evidence presented to the courts will help to reduce the need for additional assessments of parenting capability and capacity to change and avoid unnecessary delays in reaching decisions concerning permanence.

Judgments about parents' ability to meet their children's needs, which are based on assessments made at one point in time, are often prone to error. In contrast, an assessment of parental capacity to change within a timescale that is appropriate for the child, based on a systematic evaluation of the extent to which parents are overcoming problems and becoming more able to meet children's needs, with the support of evidence-based practice, provides the clearest evidence for decision-making by social work practitioners and by the courts. This paper brings together the research evidence that can inform such assessments.

Children and Families Act 2014 and revised Public Law Outline

New provisions introduced by the Children and Families Act 2014 have led to a renewed focus on social work assessments. The Act provides the legislative structure for introducing recommendations made by the *Family Justice Review* (Ministry of Justice, 2011), and is supported by a revised Public Law Outline (PLO) (Ministry of Justice, 2014) and updated guidance for local authorities on court orders and pre-proceedings (Department for Education 2014a).

The new measures include the introduction of a 26 week time limit for completing care and supervision proceedings. The strict new timeframe is intended to reduce delays in court proceedings, which have increased in length substantially over the last 15 years.

The new PLO has been piloted in order to assess the use of new practices and procedures to secure case resolution within this timeframe and ensure that proceedings are concluded justly and swiftly. One implication is that there will be a greater onus on local authorities to provide the court with evidence not only that the threshold is met but of a thorough assessment of the child's needs and the available options for meeting them so that the court can make decisions without seeking further expert assessment. There are concerns that this may simply lead to additional delays before taking legal action (McKeigue and Beckett, 2010; see also Masson and Dickens, 2013); in order to counteract this, it is therefore essential that at the outset thorough assessments are made, appropriate timescales for the child are identified and within these, goals are set and progress monitored.

A key focus of the PLO is an expectation that local authorities will submit less extensive but more analytical material to the courts on application than has been the case in the past. Such material must analyse all realistic options for the child, stating why a particular course is being recommended and the others discounted. The President of the Family Court has specifically confirmed that the new evidential requirements should not detract from local authorities' ability to meet the 26 week timeframe (see *Re B-S*, 2013). Social workers have a key role in achieving timely outcomes to care proceedings (see CAFCASS/ADCS undated).

Section 13 of the Children and Families Act 2014 also introduced measures to control the use of expert evidence and assessments in family proceedings concerning children by ensuring that these are permitted only when necessary to resolve a case justly. This is in response to research evidence that indicates that the indiscriminate use of expert witnesses and the commissioning of repeated assessments have played a major role in extending the length of court proceedings (Masson *et al.*, 2008; Ofsted, 2012; Ward, Brown and Westlake, 2012). An intended consequence of these measures is to reposition social work as a trusted profession that has a central role in care proceedings. The President of the Family Court believes that social workers have been overshadowed by the use and reliance upon other experts in the past.

Social workers are experts. In just the same way, I might add, CAFCASS officers are experts. In every case we have at least two experts – a social worker and a guardian – yet we have grown up with a culture of believing that they are not really experts and we therefore need experts with a capital E. The plain fact is that much of the time we do not (*View [2]* Sir James Munby 2013a, pp.7-8).

However, in order to provide evidence of the quality required by the courts, social workers need high quality supervision and support from managers in the local authority responsible for bringing the case (see Department for Education, 2011), and expertise in a number of areas. Among the core skills are: 'a strong grounding in child development and the impact of parental difficulties on that development; an understanding of effective, evidence-based interventions and how they can be used with families, with progress

effectively monitored and recorded; assessment and analytical skills' (CAFCASS/ADCS, undated, p.4). A key element of assessment is to undertake an 'analysis of the parenting capacity gap, with the reasons for professional pessimism if the judgment is that the gap either cannot be bridged at all, or cannot be bridged in the child's timescale' (CAFCASS/ADCS, undated, p.2).

This report is intended to provide a summary of the research evidence that can be drawn upon in making such assessments. It provides information that can underpin the assessment of parents' ability to meet children's needs, facilitate the analysis of the parenting capability gap and support those who are responsible for making judgments concerning the extent to which it has been bridged (or is likely to be bridged) within a child's timescale (see Brown and Ward, 2012).

Assessing parental capacity to change within a wider context

This review focuses on research evidence concerning parental capacity to change when children are on the edge of care. It explores how parents whose children may be placed away from home because of serious child protection concerns can be helped to become ready, willing and able to make sufficient changes to ensure that they are adequately safeguarded from harm. However it is important to note at the outset that parenting does not take place within a vacuum, and that the capacity of parents to overcome those problems that place their children at risk of significant harm is influenced both by internal factors within themselves and external factors within their families, social networks and the wider environment.

Societal factors and their influence on parenting

The functioning of individual parents is influenced by broader societal factors that are associated with neglect and abuse. It is beyond the scope of this report to explore these factors in any detail, but there is substantial evidence that social and economic inequalities, the impoverishment of communities, low pay and casualization of labour and widespread poverty and debt increase the stressors in families and communities and provide a context in which children are less likely to be safeguarded (see Jack and Jordan, 1999; Jack and Gill, 2003; Jack and Gill, 2010; Wilkinson and Pickett, 2010). This is one reason why indications of increasing inequality and widespread poverty should be viewed with concern and action should be taken to reduce them (see Brewer, Browne and Joyce, 2011; MacInnes *et al.*, 2013).

Interplay of factors

The interplay of factors which influence parents' ability to protect their children from harm and to develop the capacity to overcome abusive or neglectful behaviour patterns is acknowledged in the statutory guidance on interagency working to safeguard and promote the welfare of children. *Working Together to Safeguard Children* (HM

Government, 2013) has been radically revised following the recommendations of the *Munro Review on Child Protection* (Department for Education, 2011). However the revised version continues to ascertain that good assessments are those which take a systematic approach to enquiries, using an ecological model as set out in the *Framework for the Assessment of Children in Need and their Families* (Department of Health, Department for Education and Home Office, 2000). In considering what action to take, practitioners should investigate the interaction of factors within the following three domains:

- the child's developmental needs, including whether they are suffering or likely to suffer significant harm;
- parents' or carers' capacity (or capability) to respond to those needs; and
- the impact and influence of wider family, community and environmental circumstances (HM Government, 2013, p.19)

The different dimensions for consideration within each of these domains are set out in the Assessment Framework diagram (Figure 1.1), and are referred to in subsequent chapters of this report.

Health Basic Care Education Ensuring Safety Emotional Warmth Emotional & Behavioural Development CHILD Stimulation Family & Social Safeguarding Guidance & Boundaries Relationships and promoting Social Presentation welfare Stability Self care skills **FAMILY & ENVIRONMENTAL FACTORS**

Figure 1.1: The Assessment Framework

(Reproduced from HM Government , 2013, p.20)

Social work assessments in the real world

While a theoretical model of decision-making assumes that practitioners will work with parents and other professionals to access information across these domains, analyse it and make rational decisions concerning the most appropriate and cost-effective plan to ensure that children's needs are met, the reality is rather different, for social workers are faced with often chaotic human situations that do not fit neatly into a carefully constructed

paradigm. Information is rarely complete, and can be interpreted differently by each of those involved; there is often little time available for thoughtful analysis; and plans are constrained by the availability of resources. There is also considerable evidence that professional assessments and judgments in social work and other disciplines are influenced not only by the availability of information and resources, but also by a wide range of philosophical, psychological and organisational factors that have a powerful impact on decision making.

Availability and interpretation of information

Studies of social work practice show that, following referral, a substantial amount of time is taken up with collating information from a wide variety of formal and informal sources (Kirkman and Melrose, 2014). Much of the information from professionals is incomplete: practitioners will have focussed on different aspects of a situation and been involved at different time points, and memories and records are not always reliable. Information collected from different sources may not be adequately collated, and there may be errors in communication (Munro, 1999). All of these issues underline the importance of close cooperation between agencies. However, even when inter-agency working is strong, when decisions have to be made within a specific deadline, there may be insufficient time to gather comprehensive information. Kirkman and Melrose (2014) found, for instance, that in a busy referral team, decisions might be based on photocopied reports from other professionals that had been prepared for a different purpose; in these circumstances salient information was not prominently displayed and could be easily overlooked (p.31).

The available information may also be inaccurate. Munro (1999) points out that:

There are many reasons why people lie or distort the facts when talking to a social worker. Parents who are actually harming their child have powerful motives for concealing this. Children who are being abused can be scared to say so. Neighbours and relatives can be malicious and exaggerate or falsify what has happened in order to get the family into trouble. Even when not being deliberately dishonest, people tend to be biased in judging what seems significant and worth reporting. Neighbours who dislike a family find it easier to think of examples of their faults than of their virtues (p.752).

This description is supported by data from empirical studies which show that parents do indeed consciously or unconsciously distort or conceal information that might cast themselves in a negative light (see Forrester *et al.*, 2012), and that some professionals in adult services, who are focussed on protecting parents, will occasionally do the same (Ward, Brown and Westlake, 2012).

Not only may the information collected during assessment be incomplete and inaccurate, it is also subject to different interpretations. Professional assessments are informed by the values of an organisation and its practitioners and by the social and political context in which they are working. For instance, practitioners, teams and social work agencies can all hold subtly different positions on the ethics of care and adoption, and the extent to

which they are in tune with the dominant political agenda in this area; these positions will colour their interpretations of information and the decisions they make concerning families (Ward, Brown and Westlake, 2012). Cultural expectations and personal factors may also influence interpretations of some situations, including those involving domestic violence or neglect (see Holland, 2000; Ward, Brown and Westlake, 2012).

Factors that skew the decision-making process

Some commentators argue that, since participants in an assessment may hold competing interpretations of a situation, all based on necessarily partial or biased information, it may be futile to look for any one external reality. Practitioners are dealing with complex, and often chaotic, human situations, where nothing is entirely predictable, and decisions can rarely if ever be based on certainty (see Holland, 1999).

Nevertheless, when children are likely to suffer significant harm, decisions have to be made, and those taken formally by professionals need to be based on the best information available, and to be as objective and fair as possible. Research into social work decision-making has identified a number of factors that introduce bias into assessments and skew the process. These include human behavioural factors that are known to reduce objectivity; tendencies to favour some types of evidence over others; excessive and stressful workloads that leave little room for reflection; limited and diminishing resources that restrict the options for action; and organisational contexts that do not facilitate careful, reflective practice. The impact of all these factors is magnified by the absence of checks and balances that might redress skewed perceptions and a culture that favours intuitive over analytical thinking (Munro, 1999; Munro, 2005; Kirkman and Melrose, 2014).

Human behavioural factors

An extensive body of research on human reasoning suggests that there are two types of decision-making – a measured, analytic approach requiring careful analysis of the information available, and faster, intuitive thinking that uses experience and heuristics (short cuts) to reach conclusions (Hammond, 1996; Kahneman, 2011). There is a tendency for social work decisions to rely heavily on intuitive reasoning, partly because this aligns more closely with a practice that focuses on relationships and empathy, and partly because pressures on time and resources leave little room for measured, analytical deliberations (Munro, 1999; Holland, 1999). In fact when most people make decisions they tend to rely on experience and short cuts 'rather than face the laborious task of sifting through all the evidence and reaching a rational conclusion' (Kahneman, Slovic and Tversky, 1990).

However this 'everyday' approach is prone to a number of errors. When people rely too much on intuitive thinking they become attached to their initial impression and are slow to revise their judgment even when new and challenging information comes to light (Sutherland, 1992). They tend to base their decisions on the most readily available evidence – for instance events that are most dramatic or recent – rather than exploring

the full range of information available (Tversky and Kahneman, 1973); and they may be guided by their emotions rather than the evidence before them (La France and Hecht, 1995). Group decisions can be dominated by a desire to avoid conflict rather than a determination to establish the basic facts. People who have to make difficult or emotionally charged decisions may cope by postponing them or avoiding making them at all (Anderson, 2003; Redelmeier and Shafir, 1995) (for further information see Munro, 1999; Kirkman and Melrose, 2014).

This body of research undertaken by behavioural psychologists has provided a valuable context for studies of social work decision-making in child protection work. Both Munro (1999) and Kirkman and Melrose (2014) argue that many of the recurrent mistakes in social work decision-making are 'due to the bias introduced by using everyday habits of reasoning in assessing and reviewing cases' (Munro, 1999, p.747). For instance, the tendency to become attached to one's first impression means that some practitioners are reluctant to alter their initial decisions, even after further allegations of maltreatment have been made: this has major implications in serious cases of abuse that are first assessed erroneously as having a low risk of significant harm (see Farmer and Owen, 1995; Munro, 1999). The tendency to base decisions on the most recent or dramatic evidence means that some practitioners do not take sufficient account of the parent's history (see Munro, 1999; Kirkman and Melrose, 2014), although previous maltreatment has been shown to be a powerful indicator of subsequent abuse (see Hindley, Ramchandani and Jones, 2006; White, Hindley and Jones, in press). This has particular implications in cases of neglect where it is the long-term, chronic exposure that places the child at risk of significant harm (Egeland, Stroufe and Erickson, 1983; De Bellis, 2005). Studies of child protection conferences show high levels of consensus, suggesting that 'group think', or the desire to avoid conflict may skew decisions (see Farmer and Owen, 1995). And a number of studies have found that practitioners and their managers frequently postpone or avoid difficult decisions which will have a long-term impact on the lives of vulnerable children and their families (see Kirkman and Melrose, 2014; Ward, Munro and Dearden, 2006; Ward, Brown and Westlake, 2012).

Factors that favour some types of evidence

Studies of social-work decision-making have also identified how some types of evidence are given greater weight than others. The source of the evidence may skew the response it receives. More attention may be paid to concerns made by other professionals than to those made by neighbours and relatives (see Munro, 1999; Kirkman and Melrose, 2014). Children's allegations of abuse do not always raise appropriate concerns. They are sometimes ignored (see Farmer and Lutman, 2012); or their testimony may be accepted if it matches the social worker's assessment, but regarded as unreliable if it is at odds with it (Munro, 1999; see also Holland, 2000).

The type of evidence may also have an undue impact on decisions. Munro (1999) found that less attention is paid to written than to verbal information, a point that was illustrated by Holland (2000), who asked social workers to describe the main factors that influenced

their decisions in assessing parental capability and capacity to change. The core types of evidence cited were:

parent-related factors, including parenting skills and the relationship between parents; the ability of parents to change their behaviour and lifestyle within an acceptable timescale; and the verbal interactions between the assessing social worker and the parent being assessed' (Holland, 2000, p.152).

However the verbal interaction was frequently given the highest status, and the assessment of practical parenting skills, including emotional engagement with the child, appeared to have had little impact on decision-making.

This is an important point because it indicates how an over-emphasis on intuition and empathy, which lie at the heart of social work practice, can obscure the need for analysis and objectivity in making decisions in highly complex child protection cases. Verbal interactions are inextricably bound up with the relationship between the social worker and the parent. Relationships are thought to be key components of effective social work practice (see Trevithick, 2003; de Boer and Coady, 2007); however if the nature of the relationship is allowed to have an overriding influence on the outcome of the assessment while other information is discounted, decisions then tend to be biased in favour of those parents who are plausible and articulate, and to disfavour those who are uncommunicative or who appear passive (see Holland, 2000).

Excessive workloads and diminishing resources

Research on social work practice in assessment provides graphic data concerning the sheer volume of work with which practitioners and their managers are struggling. Social workers are faced with making innumerable decisions under extreme pressure each day, and many of these are very complex and have far-reaching consequences (Kirkman and Melrose, 2014). Frequent, sequential decision-making is known to lead to depletion of mental resources, and this in itself can lead to poor quality decisions, or to inaction (Danziger *et al.*, 2011). In such circumstances there is often little room for proactive social work or creative thinking (see Farmer and Lutman, 2012).

Moreover, social workers often have very little room for manoeuvre. Decisions can be constrained both by the actions of external bodies such as the police and the stipulations of the local authority concerning, for instance, use of resources and application of thresholds. The latter may be driven by diminishing resources following cuts in local authority spending (see Hastings *et al.*, 2013). Finally, and most importantly, parents and children will have a major influence on professional decision-making. Practitioners' options are constrained if they cannot reach an agreement with parents concerning the severity of a situation or appropriate plans for the future (see Holland, 2000).

Checks and balances

The pressures and constraints detailed above show how difficult it is in the everyday world of social work practice to make objective decisions based on a careful analysis of

all the available information. Nevertheless, a number of checks and balances could be introduced to guard against some of the pitfalls and improve the quality of decisions, particularly in the most complex cases, when children are on the edge of care.

Firstly, children's social work services are often organised in such a way that practitioners receive insufficient feedback concerning the consequences of their decisions. When cases are passed on from one team to another as their complexity becomes more evident, practitioners have little opportunity to find out what happens to the children and families with whom they are concerned. Kirkman and Melrose (2014) advocate the introduction of stronger feedback loops to help practitioners develop the experience that enables them to make more reliable decisions.

Secondly, social work assessments tend to be undertaken as private transactions between practitioners and parents; there need to be opportunities for sharing information and reflective discussion with others in order to identify whether the focus on the relationship is obscuring other indicators that should influence decision-making. Encouraging practitioners to play devil's advocate and present the arguments for an opposing point of view is an effective strategy for helping people to question their first judgments (see Koriat, Lichtenstein and Fischoff, 1980). Offering supportive one-to-one supervision, in which common errors of reasoning are acknowledged and practitioners have opportunities to reflect and identify potential mistakes in a safe environment may also improve decision-making (Munro, 1999, 2012).

Standardised, evidence-based tools

Commentators argue that the insights that come from intuitive reasoning and experience need to be counterbalanced and tested out by an exploration of more objective data (see Munro, 1999; Holland, 1999, 2000). There is therefore a strong case to be made for making greater use of standardised, evidence-based tools to support social work decision-making, and check some of the common errors of human reasoning discussed above. Validated, actuarial scales have been found to be more accurate in predicting behaviour, including the likelihood of future abuse and neglect, than clinical judgment alone (see Shlonsky and Wagner, 2005). Psychometric instruments and scales can also collect empirically valid information concerning, for instance, strengths and difficulties of children's functioning, parenting, and the individual and family environment (see Bentovim and Elliot, 2014).

The reader should be aware that although standardised, evidence-based tools can provide rigorous, valid data to support social work decision-making, there are a number of caveats. The issues are more complex than can be reflected in a numerical score or a simple actuarial table, for the range of external and internal factors interact in complex and fluid ways and the complexity of the interactions and feedback loops creates an often chaotic process of change that confounds accurate predictions of future adult behaviour or harm to a child. Moreover each case will have its own particular circumstances and unique pattern of interaction between risk and protective factors. Estimates of probability concerning *groups* of parents or children can provide a context which practitioners should

take into account, but they cannot predict what will happen to an *individual* child or their parents. Numerical data from inventories and scales can improve understanding but not provide answers to questions concerning whether parents are ready and able to change or whether children are likely to be harmed. Such instruments can inform, but not replace structured professional decision-making.

Understanding evidence based interventions

Finally, Kirkham and Melrose (2014) found that the many psychological and resource factors that can compromise effective decision-making are compounded by social workers' poor grasp of the evidence base relating to effective practice: 'stronger emphasis is placed upon the 'experience' and 'expertise' of an individual, than an understanding of which interventions are likely to have the most positive effect' (p.34). Poor understanding and use of research evidence is identified by many other commentators as an impediment to effective practice (see Munro, 1999). Further information concerning how interventions are evaluated and how to identify 'what works' is given in Chapter Six.

Effective assessments

Intuitive, relationship-based assessments produce valuable hypotheses concerning family situations and the ways in which children might better be protected. However these then need to be rigorously tested out, using a wide range of resources. Children's views should be given due weight as a fundamental element of an assessment. Historical information, written records, observation and scores from validated instruments and scales need to be fully utilised, in order to counterbalance what is learnt from verbal discussions. Assessments of individual children and families should also be informed by a thorough understanding of up to date research on parents' problems and their likely impact on children, on processes and indicators of change, and on effective strategies and interventions to support change. An important skill for social workers is applying what is known about child development, family functioning and 'what works' to test out intuitive hypotheses, to make sense of the available information about an individual family, and to support them through the process of change. This overview of the research evidence is intended to provide a resource for promoting understanding and developing these skills.

It is important to note that while this overview can identify a number of factors that should help practitioners in making assessments of parental capacity to change, these are indicators that change may (or may not) occur, and not determinants of specific actions. A significant element in the process of behavioural change is the part played by parents themselves who will exercise their free will to act idiosyncratically within the constraints of the circumstances in which they find themselves. This is one of many reasons why change in human behaviour is not a mechanistic process that can be accurately predicted.

Finally, some consideration should be given to the purpose of assessing parental capacity to change. While the primary objective is to ensure that children are safe, ethical

practice in a humane society demands that parents are given opportunities to change, and support in doing so. So assessment is not only about making a relationship with a parent and understanding their situation, or collecting and analysing information concerning risk and protective factors and the effectiveness of interventions, but also about balancing the requirement to treat parents fairly and give them sufficient opportunities for change against the demands of a timeframe that is appropriate for the child.

Purpose of this literature review

There is an extensive body of research that focusses on the risk factors that can inhibit parenting capability and the protective factors that mitigate their impact. A wide range of interventions has also been developed to help parents improve their ability to meet their children's needs. There is also a substantial body of literature on the process of change, the factors that enhance or inhibit motivation to change and engagement with services. and the timescales for change and relapse. However, this literature comes from a variety of sources and is not readily available to social workers, family justice professionals and other practitioners with safeguarding responsibilities. This narrative review of the literature is intended to produce a distillation of existing research findings and knowledge in this area and present them in a manner that is accessible to practitioners. It is not a comprehensive, systematic review of the extensive evidence base: multiple systematic reviews would have been required to cover the range of subject matter to be included in the report, and these would have taken substantially more time and resources than were available. It should also be noted that although specific intervention programmes and assessment tools are referred to in this report, they are included in order to demonstrate the link between research theory and practice. It is beyond the scope of this review to produce a comprehensive list of interventions and tools that might be used by practitioners in this field.

The purposes of the review are to:

- serve as a reference resource for social workers in their work to support families where children's safety and developmental functioning are at risk; and
- assist social workers and children's guardians in delivering focussed and robust assessments of parenting capability and parental capacity to change, and assist judges and other legal professionals in evaluating the quality of assessment work in court proceedings.

Methodology

The methodology employed to produce this evidence paper includes a literature search of peer reviewed research papers supported by expert advice from a scientific advisory

group. The literature review involved a search of relevant databases using terms including: parent, intervention, treatment, substance, domestic abuse, alcohol and mental health. A full list of the databases used and search terms is included in Appendix One. The search was limited to papers published in English within the last ten years, although earlier key papers were included in the review where they were extensively referenced by published, peer reviewed articles, or recommended by scientific advisers.

The initial search returned 16,364 results. The relevance of the results was then considered against the aims of the review. This was initially achieved by excluding papers where the title clearly bore no relevance to the review subject. The abstracts of the remaining papers were then scrutinised in order to identify those which appeared to contribute to the review aims. Using this process, 343 papers were identified as relevant to the subject of parental capacity to change and were examined in detail. The researchers focussed on papers based on empirical data, but did not apply rigid inclusion/ exclusion criteria as it was important to gain an overview of the range of literature available.

A specific form of evidence frequently referred to in this report is the systematic review. These differ from more general reviews of the research by involving a particularly structured and robust review process. Conducting a systematic review involves formulating a review question, defining the inclusion criteria for studies, developing search strategies and terms to identify all eligible studies, reviewing those studies, extracting and analysing the relevant data and assessing study quality (Uman, 2011). Systematic reviews often incorporate a meta-analysis, which uses statistical analysis to combine findings. The evidence provided using systematic review and meta-analysis is therefore particularly robust as it involves a rigorous review process. Where both were available we have given priority to systematic reviews and meta-analyses over individual papers.

The selected papers were complemented by scrutiny of a number of recent research reports that focus on these topics but which would not have been included in the databases. Those included met the study criteria and, before publication, were subject to an equivalent rigorous peer review process as academic journal articles.

A scientific advisory group made up of academics specialising in fields including child protection, socio-legal issues, social work, psychiatry and psychology were asked to review the findings of the literature search and advise whether there were any additional peer reviewed papers that had not been identified. The group also provided expert advice on specific issues and reviewed the content of the draft chapters.

A steering group, consisting of professionals from local authorities, the judiciary, organisations promoting children's and parents' rights, and academics was consulted throughout the review to ensure the relevance of this paper to social workers undertaking work in this area and/or presenting evidence of parents' capacity-to- change to the

courts. At the request of the steering group the authors have taken particular account of literature reporting parents' views. The steering group also reviewed the draft report.

Written comments from the scientific advisers and the steering group were collated and drawn upon in producing the final version of the report.

Conclusion

The results of this review of research evidence are presented in the following chapters.

- Chapter Two briefly summarises those risk factors that undermine parents' ability
 to respond to children's needs, the mitigating value of protective factors and the
 evidence concerning the impact of abuse and neglect on the developing child.
- Chapter Three considers methods of assessing the risk of future harm, and models of parental change.
- Chapter Four explores why some parents may find it difficult to accept that there is a problem and appear to be 'resistant to change'. It explores those barriers that are known to exacerbate the difficulties that some parents have in accepting that a problem exists and considers ways of overcoming them.
- Chapter Five considers how parents become motivated to change and engage with services.
- Chapter Six sets out how interventions intended to support parents in overcoming
 problems that pose a risk to their children can be evaluated, and how information
 concerning impact can be interpreted. It explores the outcomes of social work
 support and placements in care and adoption, and presents some examples of
 specific interventions designed to improve parents' ability to respond to children's
 needs that are currently being piloted or implemented in the UK.
- Chapter Seven considers the evidence concerning timescales for parental change and its sustainability. It explores how change can best be supported.
- Chapter Eight considers the implications for practice of the research findings
 covered in this report. It draws together practice messages concerning early
 support work with families, essential decision-making around thresholds for action
 including instituting legal proceedings and the collation and presentation of
 evidence to the courts.

Key points from Chapter One

- New provisions introduced by the Children and Families Act 2014 have led to a renewed focus on social work assessments. This report is intended to provide a summary of the research evidence that can be drawn upon in making such assessments. It focuses on research evidence concerning parental capacity to change, and explores how parents whose children may be placed away from home because of child protection concerns can be helped to make sufficient changes within a child's timeframe.
- Parenting does not take place within a vacuum, and assessments of parental capacity, and capacity to change, need to take an ecological approach, exploring the interrelationships between factors within the three domains of the Assessment Triangle.
- Social work assessments tend to be informed by intuitive reasoning, partly
 because this aligns more closely with a practice that focuses on relationships and
 empathy, and partly because pressures on time and resources leave little room for
 measured, analytical deliberations.
- Such intuitive, relationship-based assessments produce valuable hypotheses
 concerning family situations and the ways in which children might better be
 protected. However they are prone to bias and need to be rigorously tested out
 and counterbalanced by information from other sources including children's views,
 historical information, written records, observation, scores from validated
 instruments and scales, and research evidence concerning 'what works'.
- This wide ranging narrative literature review is intended to provide a distillation of up to date research on parents' problems and their likely impact on children, on processes and indicators of change and on effective strategies and interventions to support change that should inform such assessments.

Chapter Two: Risks of Future Harm

Introduction

The Children Act 1989 is based on the principle, also enshrined in the United Nations Convention on the Rights of the Child, that 'children are generally best looked after within the family, with both parents playing a full part' (Department of Health 1989, p.1). This principle is reflected in the duty laid on local authorities to provide services and interventions to safeguard and promote the welfare of children in their area who are in need. The local authority should promote the upbringing of such children by their families so far as is consistent with that duty (our italics) (Children Act 1989, s.17.1). The argument that, where there is no risk of significant harm, some children might nevertheless be better looked after elsewhere, is not a sufficient basis for care proceedings (see L (Children), 2006; Re L (Care: Threshold Criteria), 2007).

There is an extensive body of evidence that shows how parental problems such as domestic abuse, substance misuse and mental health problems undermine parenting capability and increase the likelihood of significant harm, particularly when they occur in combination. There is also an extensive and growing evidence base showing how the experience of abuse and neglect may have a long-term, negative impact on children's physical, cognitive, social, emotional and behavioural development that can last throughout the life course. However parental problems do not occur in isolation. They are influenced by stressors within the wider environment and family such as poor housing, poverty and isolation that make parenting more difficult and increase the likelihood that problems will arise. Parents' own attributes, and their perceptions of self-efficacy and confidence will also affect the ways in which they respond to difficult circumstances.

In families where children are on the edge of care, social work decisions should be supported by an assessment of the risk of significant harm and of parental capacity to change: that is, the assessment of a parent's capacity to improve their ability to respond to their children's needs sufficiently to ensure that they are safeguarded from significant harm within an appropriate timeframe for the child. Knowledge of the extensive research evidence summarised in this and subsequent chapters should inform these assessments. Social work practice is known to focus too often on the mother to the exclusion of the father (Brown *et al.*, 2009; Scourfield, 2006); wherever possible assessments should include both parents and the strengths and weaknesses within their support networks.

It should also be noted that, although this review focuses on parental capacity to change, such assessments are not undertaken as an isolated exercise. Parenting capability and capacity to change need to be assessed within the context of a child's needs and wishes, and decisions will also need to take account of the strengths and limitations of the other options available: a parent with a serious alcohol problem might be assessed as being unable to protect an infant from significant harm within an appropriate timeframe, but

might nevertheless be considered to be an acceptable carer for a fifteen year old who desperately wants to stay at home.

Risk factors do not in themselves determine whether a child will be neglected or abused, but they are important indicators of potential problems that practitioners need to be aware of when assessing parenting capability in families where children are on the edge of care. This chapter sets out those parental factors that are known to be associated with a risk of significant harm to a child, factors that can reduce the risk of harm, and the likely nature of that harm. Subsequent chapters then set out the research evidence concerning parental capacity to change.

Factors which undermine parenting capability

Parents' problems

There is an extensive body of research which shows that a range of problems can impair parents' ability to meet the needs of their children. These include, but are not restricted to, poor mental health, problem drug and alcohol use, learning disability and domestic abuse. Serious case reviews continually emphasise the importance of understanding and acting on concerns about children's safety and welfare when living in households where these types of parental problems are present (Brandon *et al.*, 2008; 2009; 2010).

The research literature highlights how problems such as these can impact on parenting capability in each of the dimensions identified in the Assessment Framework. The following are some examples, though not a comprehensive list (for a fuller discussion see Cleaver, Unell and Aldgate, 2011, pp.66-74). They show how problems which diminish parents' ability to respond to their children's needs in each of these dimensions lead to situations in which abuse and neglect are likely to occur.

Basic care: Domestic abuse, mental health problems, learning disabilities and problem drug and alcohol use can diminish parenting skills and mean that parents have difficulty in organising day to day life. Parents with substance misuse or alcohol problems may use all their resources to fund their addiction and have little left over to feed, clothe and house their children; they may also be too preoccupied with their addiction and too much under its influence to be aware that their children's basic needs are not being met (Fraser et al., 2009). Some mental health problems may also reduce parents' awareness of their children's basic needs or their ability to meet them. Parents with learning disabilities may want to meet their children's basic care needs but need additional support to understand what is required and learn appropriate skills (Booth and Booth, 1993; MacIntyre and Stewart, 2011).

Ensuring safety: Parents' problems may distort their perceptions of the world, diminish their ability to control their emotions and prevent them from ensuring that children are safe. Domestic abuse may literally expose children to physical danger, but it may also have an impact on 'children's sense of their own safety and security and the fear and

dread that it instil[s] in them' (Buckley *et al.*, 2007, p.300). Similarly drug taking, drinking and some mental health problems may distort parental perceptions of reality or diminish their self- control, reducing their ability to ensure that children are safe (Fraser *et al.*, 2009). Used needles and drug paraphernalia may pose a risk to young children, while substance misusing parents may also resort to theft, drug dealing and prostitution to feed their habit, and thereby potentially expose their children to a range of criminal activities and inappropriate sexual behaviour (Barnard and McKeganey, 2004; Magura and Laudet, 1996; Powis *et al.*, 2000).

Emotional warmth: Domestic abuse, mental health problems and substance misuse can have a negative impact on parent-child relationships. A parent who is the victim of domestic abuse may experience excessive stress, resulting in their becoming emotionally unavailable or distant from their children and unable to provide emotional warmth (Holt *et al.*, 2008). Impaired personality functioning and mental health problems such as anxiety disorders, depression and some psychotic illnesses may reduce parents' ability to be reciprocal, involved and encouraging with their children (Kohl *et al.*, 2011; Mowbray *et al.*, 2002). Parents whose 'principal attachment is to a substance' may have difficulty in forming attachments with their children (Kroll and Taylor, 2003).

Stimulation: Parents' problems can reduce their ability to provide adequate stimulation for their children. Substance misuse and domestic abuse, as well as mental health problems such as depression, may engender apathy, listlessness and poor self-esteem, thereby reducing parents' capacity to play or interact with their children or take an interest in their activities (see Barnard, 2007; Oyserman *et al.*, 2000; Stanley *et al.*, 2009). Parents with learning disabilities may require additional support and training to understand how to provide adequate stimulation (see Cleaver and Nicolson, 2007).

Guidance and boundaries: Parents' problems are also often associated with low self-esteem, and unpredictable, inconsistent and ineffective parenting behaviours. Children may not have clear boundaries, consistent guidance or adequate supervision (see Barnard, 2007; Cleaver and Nicolson, 2007; Oyserman *et al.*, 2000; and Stanley *et al.*, 2009). For example, domestic abuse has been found to have a negative impact on the level of control and authority an abused parent has over their child (Holt *et al.*, 2008; Ulman and Strauss, 2003). Parental substance misuse may lead to even very young children being unsupervised (see Ward, Brown and Westlake, 2012).

Stability: Parental problems also have a number of psychosocial consequences that can have a negative impact on their ability to provide children with adequate stability. For example, maternal mental illness can diminish children's sense of stability when everyday care becomes disrupted by emergency hospital admissions, or when panic attacks or depression leave a parent feeling unable to play with their child (Kohl *et al.*, 2011; Stallard *et al.*, 2004). Impaired personality functioning is associated with hostility and difficulty in controlling anger (Morse *et al.*, 2005), and with disharmonious personal relationships (Reder, Duncan and Lucey 2003). Parents with these problems may

experience multiple changes of partner. Domestic abuse, substance misuse and mental health problems are all associated with frequent changes of address, homelessness and separation through children's entry to care (Anderson and Christian, 2003; Baker *et al.*, 2003).

Impact of combinations of problems

Many parents who suffer from these problems do *not* present a risk to their children's wellbeing. Cleaver and colleagues' (2011) review of the evidence concluded that: 'much research indicates that, with adequate support, parents who are experiencing a single disorder are often able to be effective and loving parents and present little risk of significant harm to children' (p.63). However, as is evident from the above, the review also found that these factors often interlock in complex combinations and are compounded by the age or previous experiences of the parents. Interlocking, multiple problems substantially increase the likelihood that children will be exposed to maltreatment (p.65). Dixon and colleagues' (2005) found that the risk of children being abused within the first thirteen months of life is fourteen times higher when parents have been abused themselves as children, are under 21 years old, have a history of mental illness or depression and are living with a violent adult.

Environmental factors and their impact

Parents' ability to care for their children adequately is also affected by the presence of wider deprivation including poverty, poor or overcrowded housing and unemployment. It is important to note that the vast majority of poor families do not neglect or abuse their children (see Connell-Carricks, 2003; Sedlak et al., 2010; Slack et al., 2004). However as well as providing a context that makes parenting more difficult, poverty also increases the stressors in families that make abuse more likely, and there is a particular association between poverty and child neglect (see Chaffin et al., 1996). When combined with specific issues such as mental health problems or domestic abuse, poverty magnifies the challenge to parenting (Fernandez, 2007). Moreover there is an intricate, symbiotic relationship between parental problems and socio-economic circumstances (Jack and Gill, 2013). For example, mental health problems such as depression may reduce a parent's ability to find or hold down employment, thereby restricting the household income. The stress of living on limited income will potentially exacerbate a parent's mental health problems, thus creating a vicious circle. Brandon and colleagues (2009) describe poverty as one of the life stressors which affect parents' 'states of mind and the way they understand and interpret the needs and behaviour of any children in their care' (p.110).

Parenting children with additional needs

Parents face particular challenges when their children have health problems, disabilities, or emotional or behavioural difficulties. Caring for a child with additional needs can increase parents' stress levels and escalate other problems (Baker *et al.*, 2003). Parents of children with disabilities may find it difficult to access appropriate information or consistent services (Redmond and Richardson, 2003). Children with disabilities and special health care needs are 3.4 times more likely than others to experience abuse or neglect (Sullivan and Knutson, 2000). Younger children are also at greater risk of recurrent abuse, and more likely to suffer significant harm (see Hindley, Ramchandani and Jones, 2006; White, Hindley and Jones, in press).

Recurrent abuse

Once abuse has occurred there is a strong possibility of recurrence. Fluke and colleagues (1999) found that 14.7% of children experience further maltreatment within six months of the first episode, and 22.6% within eighteen months. Hindley, Ramchandani and Jones (2006) found that the risk of recurrence is highest within 30 days of the index episode of maltreatment and then begins to diminish. These researchers undertook a systematic review of cohort studies investigating factors associated with substantiated maltreatment recurrence in children (Hindley, Ramchandani and Jones, 2006). White, Hindley and Jones (in press) updated the original review to include studies published up to 2009 and to widen the original scope by including, in addition to fully substantiated cases, studies examining cases with varying degrees of substantiation. Table 2.1 below sets out 'those factors that were found to be associated with an increased likelihood of future harm, contrasted with those where the likelihood is decreased following identification of significant harm to an index child' (see Jones, Hindley and Ramchandani, 2006, p.276). Items in italics were most strongly associated with recurrent maltreatment; the other factors were identified by the studies in the reviews but were less strongly associated with recurrence. These findings have important implications for decisions concerning returning abused and neglected children home from care and the nature of support required in these circumstances (see Chapter Eight).

Risk factors that were most strongly associated with recurrence across both reviews included: maltreatment involving neglect; cases where the child had suffered more than one previous episode of abuse; and cases involving very young and/or disabled children. Parental factors included: personality disorder (impaired personality functioning); learning disabilities when accompanied by mental health problems; paranoid psychosis; substance misuse and inter-parental conflict and violence. A wide range of family and environmental factors were also associated with higher rates of recurrence; these included family stress; a lack of social and family support networks; parent-child relationship difficulties and living in a violent or unsupportive neighbourhood (Hindley, Ramchandani and Jones, 2006; White, Hindley and Jones, in press).

Some factors were not found to be significant in terms of the recurrence of maltreatment. These included parental marital status, caregiver age, employment status and ethnicity (Hindley, Ramchandani and Jones, 2006; White, Hindley and Jones, in press).

Protective factors

The reviews also identified factors that can interact positively with parental problems, mitigating their impact, and reducing the likelihood of maltreatment recurrence. These include the presence of a non-abusive partner; the presence of a supportive extended family; parents' adaptation to their own experience of childhood abuse; parents' recognition that there is a problem and their willingness to take responsibility for it; and parents' willingness to engage with services (Hindley, Ramchandani and Jones, 2006; White, Hindley and Jones, in press). These factors have been found to protect children from initial experiences of abuse and neglect as well as from recurrences (see Cleaver *et al.*, 2007; Cleaver and Nicholson, 2007; Kroll and Taylor, 2003; Smith, 2004; Somers, 2007).

Table 2.1: Factors associated with future harm NB Items in italics most strongly associated with maltreatment

Factors	Future significant harm more likely	Future significant harm less likely
Abuse	Severe physical abuse including burns/scalds Neglect Severe growth failure Multiple types of maltreatment More than one affected child in the household Previous maltreatment Sexual abuse with penetration or repeated over a long duration Fabricated/induced illness Sadistic abuse	Less severe forms of abuse (defined in terms of harm, duration and frequency)
Child	Developmental delay with special needs Child's mental health problems Very young child – requiring rapid parental change	Healthy child Child does not blame themselves for sexual abuse and recognises that it caused harm Later age of onset One good corrective relationship
Parent	Personality disorder (anti-social, sadistic, aggressive) Lack of compliance Denial of problems Learning disabilities plus mental illness Substance abuse Paranoid psychosis Significant parental mental health problems Abuse in childhood – not recognised as a problem History of violence or sexual assault	Non-abusive partner Willingness to engage with services Recognition of problem Responsibility taken Mental disorder responsive to treatment Adaptation to childhood abuse
Parenting and parent/ child interaction	Disorganised; severe insecure patterns of attachment Lack of empathy for child Own needs before child's Parent-child relationship difficulties	Secure attachment; less insecure attachment patterns Empathy for child Competence in some areas

Family	Inter-parental conflict and violence	Absence of domestic abuse
	High stress (associated with family stress; parental stress; large	Non-abusive partner
	family size, poor home conditions and housing instability)	Capacity for change
	Power problems: poor negotiation and expression of emotions;	Supportive extended family
	poor sense of autonomy	
	Children not visible to the outside world and continuing perpetrator	
	access	
Professional	Lack of resources	Therapeutic relationship with child
	Poorly skilled professionals	Outreach to family
		Partnership with parents
Social setting	Social isolation	Social support
	Lack of social and family support networks and lone parenthood	More local child care facilities
	Violent, unsupportive neighbourhood	Volunteer network
		Involvement of legal or medical services

(compiled from Hindley, Ramchandani and Jones, 2006; White, Hindley and Jones, in press)

The impact of abuse and neglect on children's subsequent development

There is a strong and increasing evidence base showing that the early childhood environment, and the first three years of life in particular, play a major role in shaping children's neurobiological, cognitive, socio-emotional and behavioural development (for summaries see Barlow and Scott, 2010; Brown and Ward, 2012). Abusive and neglectful parenting can have severe negative consequences for all aspects of children's future learning, behaviour and health, and these may persist well into adulthood. Norman and colleagues (2012) have recently completed a systematic review and meta-analysis examining the long-term health consequences of child physical abuse, emotional abuse, and neglect. They found robust evidence showing causal relationships between non-sexual child maltreatment and a range of mental disorders, drug use, suicide attempts, sexually transmitted infections and risky sexual behaviour.

Attachment styles and their impact

Abuse and neglect can have a negative impact at all ages and stages of development, but are thought to be particularly pernicious in the early years because of the impact on the child's neurobiological development and on the attachment process. The developing child will respond as much to negative as to positive stimuli, and many of the *sequelae* of abuse and neglect can be interpreted as ways of adapting to a hostile environment (see McCrory, De Brito and Viding, 2012).

The key feature of an infant's early environment is the relationship with the primary carer, usually the mother, whose initial role is to ensure the infant survives by responding to their basic needs. Howe (2006) argues that 'the goal of the attachment system is protection from danger.... and attachment behaviours are triggered whenever a highly vulnerable human infant experiences anxiety, fear, confusion or feelings of abandonment' (p.128). Children whose parents or other primary carers are responsive to their needs and sensitive and emotionally attuned to their mental states are likely to develop an internal working model of themselves as loveable, valued and socially effective and others as positively available, understanding and interested. These children develop secure attachments, and these form the basis for 'healthy psychosocial development, improved social cognition, and raised levels of resilience based on high self-esteem, self-efficacy and coping capacity' (*ibid.* p.128).

A number of aspects of early parent-child interactions have been identified as laying the foundations of secure attachments. These include: <code>sensitivity/attunement</code> (the use of eye contact, voice-tone, pitch and rhythm, facial expression and touch to convey synchronicity with the infant); <code>mind-mindedness/reflective function/ mentalisation</code> (a parent's capacity to experience their baby as an intentional being with their own personality traits, strengths and sensitivities); <code>marked mirroring</code> (when a parent shows a

contingent response to an infant such as looking sad when the baby is crying); containment (when a parent uses touch, gesture, and speech to take on an infant's powerful feelings and make them more manageable); reciprocity (turn-taking); and continuity of care (providing infants with sufficient continuous caretaking from a small number of carers to enable them to become securely attached) (Tronick, 2007).

However, parents and other primary carers differ in their ability to help children develop secure attachments. Children whose carers are less available, sensitive or responsive may develop insecure attachments. Children with *insecure/avoidant* attachments have carers who become anxious or rejecting when others place emotional demands on them. These children adapt by over-regulating their emotions and are anxious that any display of need or vulnerability may drive their carers away. More extreme avoidant strategies are found in children who have experienced rejection, physical abuse or emotional maltreatment (Howe, 2006).

Children with *insecure/ambivalent* attachments have caregivers who are unable to respond consistently or to recognise their needs and those of other people. These children respond by maximising their distress and attachment behaviour in order to gain attention. They develop 'a passive and fatalistic attitude to events; an anxious preoccupation with other people's inconsistent emotional availability; and an angry, demanding, dissatisfied, needy, pleading and provocative approach to relationships' (Howe, 2006, p.129). Pronounced versions of this strategy are found in some children who have experienced chaotic neglect (Howe, 2006).

Children with *disorganised* attachments are unable to develop an attachment strategy because their carers are both the source of their distress and fear and the solution to it. These children have caregivers who may be frightening in that they menace, abandon or physically or sexually abuse them; or carers who may be frightened in that they may behave in a helpless and emotionally dysregulated way when faced with their needs. These children do not know what response to expect from their caregivers: sometimes they may be picked up and cuddled, but at other times they may be shouted at or smacked. In such circumstances children may not be able to develop a strategy to maximise their carer's availability. Children with disorganised attachments have considerable difficulty regulating their emotions and will develop highly negative and inconsistent internal working models in which they perceive others as not to be trusted. Up to 80% of children brought up in neglectful or abusive environments develop disorganised attachments (Van Ijzendoorn *et al.*, 1999), and these are strongly associated with later psychopathology (Green and Goldwyn, 2002). (For further details on the development of attachment styles, see Howe, 2006).

The impact of abuse and neglect as children grow

Young children are entirely dependent on their carers for survival, therefore the attachment formed with the primary caregiver(s) shapes the way in which the child

develops, particularly their capacity to regulate their emotions and stress levels. As a child grows, the impact of abuse and neglect will continue to be felt across different aspects of their development. They may exhibit behavioural difficulties such as aggression or cognitive issues including delayed language skills.

Once children reach middle childhood, behavioural and developmental issues become apparent as they start school and interact with an increasing number of other children and adults. Trickett and McBride-Chang's (1995) review of the developmental impact of abuse and neglect found that from early childhood through to adolescence, neglected children consistently show poorer cognitive development and school performance than their peers. Abuse and neglect continue to impact on children's emotional development including the ability to regulate their feelings and responses and the ability to recognise expressions of emotion in others (Pollak *et al.*, 2001).

As children become older and more socially aware, the difficulties faced by their parents will impact upon them in new ways. Those in middle childhood will become increasingly aware of the stigma attached to certain conditions such as schizophrenia or drug dependence and may try to keep their families together by hiding problems from the outside world (Barnard and Barlow, 2003; Somers, 2007). Young people can do this so effectively that they miss out on the professional support they need (Velleman *et al.*, 2008). This can be a sources of stress, which despite employing a range of coping strategies, can leave young people feeling 'extremely angry, frustrated and very sad' (Velleman and colleagues, 2008, p.402).

Adolescence is the point where young people are becoming more socially independent and peer relationships become as important, or more important, than those with family. The formation of these relationships, however, can be compromised where young people have developed insecure attachment styles (Levendosky, Huth-Bocks and Semel, 2002). Abuse and neglect have also been linked to antisocial and risky behaviours in adolescents (Romer, 2010; Scaramella *et al.*, 2002).

Even when young people reach adulthood, the impact of abuse and neglect can be seen in a number of negative outcomes including mental and physical health problems, unemployment and antisocial or criminal behaviour. Where adults have not resolved their own insecure attachment issues, when they become parents themselves, this can negatively impact on the attachment relationship with their own children.

Table 2.2 sets out examples of the potential impact of abuse and neglect from birth through to adulthood.

Table 2.2: Impact of abuse and neglect during key developmental timeframes for children

	Impact of abuse/neglect	
Before birth	Exposure to drugs/alcohol <i>in utero</i> can increase the risk of preterm birth, low birth weight and foetal alcohol syndrome (Autti-Rämö, 2002; Pinto <i>et al.</i> , 2010). Pre-natal exposure to domestic abuse can be fatal; it can also damage a child's future ability to cope with stress (Shonkoff <i>et al.</i> , 2012). The process of attachment begins before birth (Benoit <i>et al.</i> , 1997; Canella, 2005).	
0-3 years	Maltreatment during early childhood can cause functional and structural changes to brain development which can predispose to psychiatric vulnerability in adulthood (McCrory, De Brito and Viding, 2012). Child neglect can have severe, deleterious short- and long-term effects on children's cognitive, socio-emotional, and behavioural development. Consistent with attachment and related theories, neglect occurring early in life is particularly detrimental to subsequent development (Hildyard and Wolfe, 2002).	
	Infants and very young children are dependent on caregivers for survival. Feelings of hunger, cold or discomfort indicate a threat to survival and generate stress. Infants cannot regulate their own stress response systems and are therefore dependent upon caregivers to re-establish their equilibrium by meeting their needs and soothing away their stress (Hofer, 1995). The stress response system begins to self-regulate at around six months, but does not become fully established until a child is around four years old. It will develop atypically in response to aggressive, hostile or neglectful parenting. Children and young people with maladaptive stress responses will find it difficult to control their behaviour, or to regulate their emotions (Tarullo and Gunnar, 2006).	
	The attachment formed with the primary caregiver(s) shapes the way in which the child develops, and may be influenced as much by negative as by positive parenting behaviour: up to 80% of children brought up in neglectful or abusive environments develop disorganised attachments, and these are strongly associated with later psychopathology (Van IJzendoorn, Schuengel and Bakermans-Kranenburg, 1999). Children are more likely to demonstrate early developmental delay if their mothers demonstrate avoidant attachment styles and depressive symptomatology including postpartum depression (Alhusen, Hayat and Gross, 2013).	
	Physically neglected toddlers have been found to cope less well with problem-solving tasks, reacting with frustration or anger (Hildyard and Wolfe, 2002). A systematic review by Naughton and colleagues (2013) identified negativity in play, reduced social interactions and deficits in memory performance as potential developmental outcomes of neglect and abuse.	
3-5 years	Between the ages of three and five there is a dramatic spurt in the development of executive function skills such as working memory, inhibitory control and cognitive flexibility. The sequential nature of childhood development requires earlier stages to be completed before more complex skills can be acquired (Blair, 2002; Knudsen, 2004).	
	As maltreated children start pre-school and school they are likely to have fewer positive social interactions than their peers, occurring as a response to previous insensitive care giving (DiLalla and Crittenden, 1990). They show an increased risk of aggression and other conduct problems, find it difficult to recognise that others will help them, and complex language skills may be developmentally delayed (Naughton	

	et al., 2013).	
	Studies suggest that early negative emotional experiences can alter the way maltreated children recognise and process emotions such as anger (Pollak <i>et al.</i> , 2001). Deficits have been identified in the moral development of neglected and abused school aged children, which have been associated with an increase in negative behaviours including cheating and stealing (Koenig <i>et al.</i> , 2004).	
5-10 years: middle childhood	In middle childhood maltreated children may experience delayed cognitive and educational development. For example, their complex language skills may continue to be delayed (Naughton <i>et al.</i> , 2013). They may exhibit emotional or behavioural problems including fear, anxiety, anger or aggression (Barnard and Barlow, 2003; Covell and Howe, 2009). The may have difficulty understanding emotional expression and not recognise others as a source of help (Naughton <i>et al.</i> , 2013). Children at this age are starting to become aware of the stigma attached to their parents' issues (Somers, 2007). They may also begin to take on caring responsibilities both for their parents and their siblings, and consequently miss out on school and social activities (Aldridge and Becker, 2003).	
Adolescence	Adolescence is the point where young people are becoming more socially independent and peer relationships become as important, or more important, than those with family. The formation of these relationships, however, may be compromised where young people have not developed secure attachment styles (Levendosky, Huth-Bocks and Semel, 2002). The ability of adolescents to form social networks can also be hampered where they have to assume a caring role for a parent or sibling because of their parent's problems (Aldridge and Becker, 2003). Caring responsibilities or anxiety about their parent's wellbeing during school hours may also detrimentally affect young people's education (Aldridge and Becker, 2003; Barnard and Barlow, 2003). However children who experience domestic abuse may find that school can offer a 'safe haven' for several hours a day (Buckley <i>et al.</i> , 2007).	
	Adolescents who have not experienced nurturing and involved parenting are more likely to become involved in antisocial behaviour and delinquency (Scaramella <i>et al.</i> , 2002). Early stressors such as abuse and neglect have been linked to adverse adolescent outcomes and risky behaviours including drug taking, addiction, teenage pregnancy and suicide (Herrenkohl <i>et al.</i> ,1998; Romer, 2010).	
Adulthood	Adults with unresolved attachment issues may find it difficult to open up and trust others or develop supportive social networks (Anders and Tucker, 2000). Unless parenting skills are learnt and attachment issues resolved, there is a risk that parents' unresolved or disorganised attachments will impact on the nature of the relationship with their own children (Hesse and Main, 2000).	
	Childhood abuse and neglect has been linked to negative physical and mental health outcomes (Springer <i>et al.</i> , 2007); lower educational attainment, employment outcomes and earnings (Currie and Widom, 2010; Perez and Widom, 1994); and an increased likelihood of revictimisation in adulthood (Widom, Czaja and Dutton, 2008).	

The research evidence shows how abuse and neglect in the early years can have an adverse impact on children's development, although it does not imply that this will always be the case. Such changes are not irreversible, and there are many factors both within the child, the family and the environment which can mitigate the developmental consequences of maltreatment. Nevertheless, the more prolonged and persistent the abuse or neglect, the more difficult it is to overcome the impact. For example, a longitudinal study is comparing outcomes of children who were removed from Romanian orphanages and placed for adoption in England with those of children born and adopted in this country. The study has explored the impact of severe institutional deprivation across cognitive, intellectual and behavioural domains, among others, and concluded that there has usually been a large improvement in functioning following adoption, although 'significant problems continued in a substantial minority of the children placed after the age of six months' (Rutter et al., 2007).

It is therefore imperative that professionals with responsibilities for safeguarding children make and implement effective and timely decisions both to identify those likely to suffer significant harm and to take action to prevent its occurrence (or recurrence). Understanding how far parents are capable of addressing the difficulties which render their children more likely to suffer maltreatment, within a timeframe which recognises children's developmental needs and that is sustainable in the long term, is fundamental to effective safeguarding.

Identifying children suffering, or likely to suffer significant harm

Many parents will be capable of change and, with appropriate support, will be able to provide a safe and nurturing family context for their children within a timeframe which is in their best interests, and that is sustainable in the long term. Ward and Brown's prospective longitudinal study of infants suffering or likely to suffer significant harm has identified a number of parents who succeeded in overcoming significant adverse behaviour patterns before their children's first birthdays and have now provided nurturing homes for them for at least seven years (see Brown, Hyde-Dryden, Thomas and Ward, forthcoming; Ward, Brown and Maskell-Graham, 2012; Ward, Brown and Westlake, 2012). However, not all children can be safely looked after within the family. Social workers' assessments, analyses and judgments about the impact of abuse and neglect on children's development; parents' understanding of the need to make changes; and their readiness, motivation and ability to change will be crucial components of decisions about whether it is in the best interest of the child to remain with parents, or to be placed away from home on a temporary or permanent basis. The child's welfare should remain central to this decision-making process; the identification of families where children are exposed to ongoing maltreatment with little prospect of change should occur as soon as

is practicably possible, supported by robust assessments prepared for the courts. As Jones (2009) points out:

We have to acknowledge that some situations cannot be changed for the better, and that some families are simply untreatable. These situations are major challenges for children's social care and other services, but must be faced and responded to by front-line workers and their supervisors. These cases do not represent failure, but in fact successful professional practice, to the extent that a sustained focus on child welfare has been achieved (p.302).

Tools to assess risk of significant harm

Chapter One discussed how difficult it is for social workers to make objective decisions based on careful analyses of all the available information, within the pressures and constraints of everyday practice. It also explored some of the pitfalls of relying too heavily on intuitive, relationship based assessments, and how these might be addressed, particularly through testing out initial hypotheses with more objective data collected by using validated evidence-based tools. Social work assessments when children are on the edge of care involve both an assessment of child and family functioning and parents' capacity to change, and an assessment of the risk of future harm, both of which are key elements in decision-making (Shlonsky and Wagner, 2005). Research from at least 100 comparative studies of practice in several disciplines within the social sciences has found that professional observation and clinical judgments are less accurate in predicting future behaviour than the actuarial methods upon which validated risk assessment tools are based (Dawes, Faust and Meehl, 1989; Shlonsky and Wagner, 2005). It should be noted. however, that the complexities of cases where there are serious child protection concerns mean that no tool will be 100% reliable, and standardised actuarial instruments are only 70%-80% accurate in identifying risks of future harm (Leschied et al., 2003). Such tools are a valuable aid to structured professional decision-making, but they cannot replace professional observation and judgment.

All assessment tools designed to identify whether children are suffering or likely to suffer significant harm need to have been formally evaluated in the context within which they will be used, and many of those currently available still require further validation in the UK before they can be relied upon. Only a limited number of standardised tools are routinely used in a small number of local authorities in England. Nevertheless they are increasingly being introduced across the country, and it is important for practitioners to understand their strengths and weaknesses.

Barlow, Fisher and Jones (2012) undertook a systematic review of tools available to social workers to assist them in deciding whether a child is suffering, or likely to suffer, significant harm. Sixteen tools met the criteria for inclusion. These included:

Risk assessment tools that measure a small number of historical and static risk factors that research has shown to be strongly associated with future harm. The evidence

supported the use of one actuarial risk assessment tool 'in some contexts, as part of the assessment process in order to classify the presenting nature and severity of any harm' (p10). This was the California Family Risk Assessment Tool included in the Children's Research Centre Structured Decision-Making System (CRC-SDM). However the review suggests that further evidence is needed about its validity, impact, rigour and acceptability in a UK context.

Strengths and Needs Assessment tools that typically measure dynamic factors that are often defined as needs and which, if remedied, can reduce the risk of harm posed (p.33). The review identified two tools developed in the UK: the Graded Care Profile and the Safeguarding Assessment and Analysis Framework which, when compared with the other tools reviewed, were more consistent with the Assessment Framework, included assessment over a wide range of domains, and provided clear guidance for social workers on use and analysis of the data. Barlow and colleagues consider that these tools 'could potentially improve both the assessment and analysis of data about children in need and children with complex needs in terms of the likelihood of them suffering significant harm'(p.10). However, the review again points to the need for formal piloting of both tools in the UK to test for reliability, validity, impact and acceptability.

The Signs of Safety tool (Turnell and Edwards, 1997), used by a number of local authorities across the UK, focuses on the joint setting of goals and parents' strengths and resources rather than deficits. However, Barlow, Fisher and Jones (2012) found that it 'has a limited number of assessment domains, none of which focus on children's development, limited consistency with the Assessment Framework (HM Government, 2013) and provides limited guidance in terms of analysing and making sense of the data vis-a-vis other tools such as the Safety Assessment and Analysis Framework (SAAF)' (p.73). Its strengths are that it can be used to map evidence in the process of making sense of it and that it provides a visual means of supporting partnership working with children and families to help them understand strengths as well as weaknesses.

Response Priority Decision Trees: tools that are used to 'improve the consistency across workers and to prioritise decisions about initial reports of abuse and neglect, in order to focus the workload on the most relevant cases and aid decision-making about the rapidity of response that is needed' (Barlow, Fisher and Jones, 2012, p.6). The CRC-SDM Response Priority Decision Trees meet this requirement but again the review indicates that they would require testing in a UK setting.

Permanency/Placement and Reunification Checklists have been developed as part of the CRC-SDM structured decision-making system and 'focus explicitly on the likelihood of recurrence of harm in relation to decisions about permanency/ placement and reunification' (Barlow, Fisher and Jones, p.6) but once again would require testing out in a UK setting.

Audit Tools, which are similar to the risk assessment tools, but have been used to date as a means of auditing retrospectively whether cases have been classified accurately.

Ward, Brown and Westlake's (2012) risk classification is based on the work of Hindley, Jones and Ramchandani (2006). It has been developed from an audit tool into a more dynamic methodology for identifying risks and setting goals and timescales in consultation with parents when reunification is being considered. This is currently being piloted and evaluated by the NSPCC.¹

Conclusion

This chapter has explored the evidence base upon which social workers should be able to draw in making decisions about whether abused and neglected children who are on the edge of care can safely remain with birth parents without being placed at continuing risk of significant harm. It has considered the risk factors that make continuing abuse and neglect more likely, and the protective factors that can mitigate their impact; it has also considered the research evidence concerning the continuing impact of abuse and neglect across the dimensions of children's development. Finally it has explored a number of standardised tools that may be usefully implemented to support structured professional decision-making concerning the risk of significant harm. It has, however, focussed on baseline indicators of abuse and neglect, their potential impact and the likelihood of future harm and presented research evidence that demonstrates why change is necessary when children are living with carers who abuse or neglect them. It has not considered how parents' (and children's) circumstances might change, and how positive change might be facilitated and sustained. The remainder of this review focusses on these issues.

Key points from Chapter Two

- Parental problems such as domestic abuse, mental health problems and substance misuse may increase the likelihood of children being maltreated.
 Parents experiencing a single such problem can often provide effective and loving care. However, the risk of maltreatment substantially increases where parents are experiencing interlocking combinations of problems.
- Parents' ability to care for their children adequately is also affected by environmental factors, for example, the presence of wider deprivation including poverty, poor or overcrowded housing and unemployment. Parents also face particular challenges when their children have health problems, disabilities, or emotional or behavioural difficulties.

¹ See NSPCC *Taking Care* project: http://www.nspcc.org.uk/what-we-do/the-work-we-do/priorities-and-programmes/looked-after-children/reunification-of-maltreated-children/reunification-of-children_wda87216.html

- A number of factors associated with an increased risk of harm or the reoccurrence of harm have been well established, in addition to protective factors which decrease that risk.
- Abuse and neglect can have negative consequences across the whole spectrum of children's development, and can continue to have an impact into adolescence and adulthood.
- A range of materials have been developed to assist practitioners in assessing the likelihood of current or future significant harm. Most of these need further validation and/or translation and piloting in a UK context. Once properly validated they should provide practitioners with standardised measures that can support the development of structured professional judgment.

Chapter Three: Models of change: their strengths and weaknesses

Introduction

An individual's readiness and motivation to change is influenced by several interrelated factors, and the assessment of change is likely to be a complex process. In addition, in families where there are serious child protection issues, parents may need to undertake a considerable number of changes in order to ensure that their children are adequately safeguarded. With support from professionals, they may need to address the underlying factors in their own lives that make maltreatment more likely; they may need to address the abuse and neglect itself, and understand the manner in which their children are likely to be harmed; and they may need to develop more effective parenting skills, that are designed to promote children's satisfactory development, improve family relationships and reduce tensions within the family that may be the consequence of past abusive behaviour patterns. A key element of assessment is to 'undertake an analysis of the parenting capacity gap, with the reasons for professional pessimism if the judgment is that the gap either cannot be bridged at all, or cannot be bridged in the child's timescale' (Cafcass/ADCS, 2012, p.2). This chapter considers how social workers can map out and analyse the circumstances of individual cases in order to understand the factors promoting or impeding parental change of behaviour. The chapter then discusses the strengths and weaknesses of models for assessing parent's readiness and capacity for change.

Case conceptualisation

Social workers need to be able to assess whether parents have the capacity to make changes in many interlocking areas. In order to do this they need to formulate a case conceptualisation, setting out the various external and internal factors that impact on parents' ability to meet their children's needs, and identifying strengths as well as weaknesses. The social worker's ability to form a relationship with the parent and explore with them their perceptions of their current situation and how it has arisen is an important element in case conceptualisation (see Holland, 2000), but the process also needs to be supported by more formal, methodical collection of and analysis of relevant information (see Kirkman and Melrose, 2014). The different domains of the Assessment Triangle (HM Government, 2013), covering the developmental needs of children, the capacity of parents or caregivers to respond appropriately to those needs and the impact of wider family and environmental factors on parenting capability and children's progress provides a conceptual framework for reaching a structured professional judgment concerning the strengths and weaknesses in a family and their environment and the issues that need to be addressed (for further information see Sperry, 2005). This type of conceptual map,

that helps practitioners make sense of the interrelationships between factors concerning the child, the family and the environment, is sometimes described as an integrated theoretical framework (see Harnett, 2007; Harnett and Dawe, 2008; Harnett and Day, 2008).

In families where parents' poor personality functioning is an issue, social workers will also need a conceptual framework of the elements of mature adult functioning that can offer secure and responsive parenting; without this they are unlikely to identify the salient features that underlie the deficits in parenting capability, or to assess parents' capacity to act as beneficial attachment figures. There is no universally agreed framework for mature adult functioning, but it will include: parents' ability to recognise emotions in others; their ability to control and manage their own emotions; their relationship to authority; their ability to trust; their capacity for intimacy; and their capacity to take responsibility for the impact of their actions (Dimitrova *et al.*, 2010; Marganska *et al.*, 2013; Shean and Meyer, 2009).

Parents' attachment styles and their implications

Parents' experience of childhood trauma is likely to have a negative impact upon their own internal working models and may continue to affect their attachment style into adulthood. Childhood experiences may therefore impact on a parent's relationships with their own children, with partners and with peers (Styron and Janoff-Bulman, 1997; Turney and Tanner, 2001). Practitioners therefore need to be mindful that parents' attachment styles may present an obstacle when assessing and working with families and take this into account when formulating a case conceptualisation (see Holland, 2000).

Where parents have unresolved or disorganised attachments, their resultant behaviours will impact on the nature of the relationship with their own children. Hesse and Main (2000) suggest that parents with unresolved or disorganised attachments may exhibit inconsistent behaviours or ones which frighten or alarm children who look to them for safety and care. The authors describe children in these circumstances as being confronted with a 'biologically channelled paradox: the simultaneous needs to approach, and take flight, from the parents' (p.1118) and suggest that this leads to disruptions in the child's behaviour.

Insecure parental attachment style also poses potential challenges for professionals seeking to improve parenting capacity. Parents with an avoidant attachment style may appear distant, believe others are unreliable and distrust close relationships (Riggs and Kaminski, 2010). This can result in parents being suspicious of those in authority and unwilling to build relationships with professionals. Relationships with social workers or other practitioners and engagement in parenting programmes are therefore likely to be affected by an insecure attachment style. Anders and Tucker's (2000) study of social support networks found that adults with avoidant attachment styles tended not to open up and disclose to others, and this 'inhibits the possibility for close, supportive relationships

to develop and endure' (p.387). This study also found that individuals with avoidant attachment styles have lower levels of support satisfaction, associated with their tendency not to open up to others and their lack of assertiveness. The authors suggest that this lack of assertiveness may, in part, be due to adults' expectations of others' indifference towards them. An avoidant style may make it difficult for social workers to discuss issues with parents fully and gain a true picture of whether parents are coping. Holland (2000) found that parents' verbal interactions with social workers frequently carried the greatest weight in assessments of parental capability and capacity to change, with the result that those whose attachment styles led them to be inarticulate or passive may have been disadvantaged. When assessing parental capacity to change, social workers therefore need to recognise that exhibiting an avoidant, ambivalent or disorganised attachment style is a foreseeable, although not inevitable, reaction from parents who have themselves experienced childhood trauma.

This chapter now considers a number of conceptual and theoretical frameworks of behavioural change that should help facilitate social work understanding of parental readiness and capacity to overcome those difficulties which place their children at a greater risk of significant harm.

Models of change

When adverse behaviour patterns are entrenched, the process of change can be complex. Psychological models of change can offer helpful insights into behaviour patterns when parental capacity is being assessed. Without a conceptual framework of the process of change, practitioners may fail to appreciate its complexity and rely on compliance as a proxy indicator of success. For instance, parents who attend parenting programmes or drugs counselling are sometimes viewed as making sufficient changes even if their adverse behaviour continues (Brandon *et al.*, 2008). However, compliance with voluntary or statutory interventions does not in itself constitute psychological readiness or provide evidence of genuine behaviour change (see Prochaska and Prochaska, 2002), and will do little to protect the child from future harm.

The Trans-Theoretical Model of Change (TTM)

A number of psychological models have been developed that are designed to facilitate understanding of the processes of behavioural change (see Hegarty *et al.*, 2008; Higginson and Mansell, 2008; Littell and Girvin, 2005; Prochaska and DiClemente, 1983, and discussion below). Many of these models originated in attempts to understand addictive behaviours that have a long-term impact on health, such as smoking or alcohol abuse (see Prochaska and DiClemente, 1982; Tober and Raistrick, 2007).

The most prominent of these is Prochaska and DiClemente's Trans-Theoretical Model of Change (TTM), incorporating the stages of change (SOC) (DiClemente and Prochaska,

1982; Prochaska and DiClemente, 1983; Prochaska, DiClemente and Norcross, 1992; Prochaska and Prochaska, 2002). The model was originally developed in the United States to understand the process of change for smoking cessation (DiClemente and Prochaska, 1982; Prochaska and DiClemente, 1983). However, it is a generic model, that has since been widely used in the treatment of a range of problems, including addictions (see for instance Snow, Prochaska and Rossi, 1994; Tober and Raistrick, 2007); domestic abuse (see for instance Frasier *et al.*, 2001); adult sex offenders (see for instance Ginsberg *et al.*, 2002); and HIV prevention (see for instance Aggleton *et al.*, 1994). The TTM has been developed and refined on the basis of empirical evidence from this wide range of populations (Prochaska *et al.*, 1992). It dominates much of the literature relating to behavioural change and has become a prominent and widely accepted feature within the health promotion field (Whitelaw *et al.*, 2000).

The TTM has also been applied specifically in child welfare cases (Prochaska and Prochaska, 2002; Tuck, 2004). In the UK it features prominently in the social work literature on assessment, where it has been developed as a framework to assist practitioners in conceptualising processes of change in relation to work with families where children are suffering, or likely to suffer, significant harm, and in the preparation of reports for the courts (see Horwath and Morrison, 2001; Morrison, 2010). While it is useful for social workers to have an understanding of the key concepts of the model, they also need to be aware of a number of concerns that have been raised about its applicability to child welfare cases, and the predictive validity of some of the accompanying materials (see below).

Prochaska, DiClemente and Norcross (1992) and Prochaska and Prochaska (2002) identify six psychological stages through which individuals advance during intentional behavioural change: *pre-contemplation, contemplation, preparation, action, maintenance and relapse.*²

Pre-contemplation: describes the stage where the individual has no intention of changing their behaviour and is unaware of their problems or will not acknowledge them. Pre-contemplators may feel coerced or threatened to change their behaviour, and this can be particularly pertinent in child welfare cases. These individuals may demonstrate some change, as long as there is pressure to do so. However, once this pressure no longer exists, they are highly likely to revert to their previous behavioural patterns.

Contemplation: during this stage individuals are aware that a problem exists and are seriously thinking about addressing their difficulties, but have not made any commitment to take action. The contemplation stage can last for long periods and involves weighing up the disadvantages of the problem behaviour *versus* the amount of effort, energy, and

² The following descriptions of each stage have been adapted from Prochaska and colleagues (1992 and 2002).

loss it will cost to overcome it. Parents in this stage may agree with statements such as: 'I have a problem and I really think I should work on it' or 'I have started to think I have not been caring for my children as well as I should'. Individuals in the contemplation stage give serious consideration to problem resolution.

Preparation: this stage combines intention and behavioural criteria. For instance, individuals in this stage may make or support statements such as the following: 'I have questions for my social worker about caring for my children' or 'If I do not change I will not be the type of parent my children need'. These parents may report making some small behavioural and cognitive changes, such as reducing their intake of alcohol or drugs; however, they have not yet met the criteria for effective action and may continue with their adverse behaviour.

Action: individuals in this stage have made modifications to their behaviour, experiences, or environment in order to overcome their problems. These modifications tend to be most visible and receive the greatest external recognition. However, 'people, including professionals, often erroneously equate action with change. As a consequence, they overlook the requisite work that prepares changers for action and the important efforts necessary to maintain changes following action' (Prochaska *et al.*,1992, p.1104). In relation to addiction, this would involve reaching abstinence. Reducing intake of drugs or alcohol would not satisfy the criteria for the action stage. Parents in this stage may agree with the following types of statement: 'I am really working hard to change' or 'I am doing things about my problem that got my social worker involved'. The hallmarks of the action stage include modification of the problem behaviour and substantial overt efforts to address it.

Maintenance: during this stage individuals work towards preventing relapses and consolidating the gains attained during the action stage. However, this stage can go on to last a lifetime. Parents in the maintenance stage may make statements such as: 'I may need a boost right now to help me maintain the changes I have made' or 'I sometimes feel nervous that when my social worker is out of my life I will fall back to my old behaviour'. Parents in this stage will be working hard to prevent relapse and to stabilise their behavioural change.

Relapse: in response to their research on addictive behaviours, Prochaska and colleagues (1992) added *relapse* as an additional stage in the model. The Trans-Theoretical Model of Change was originally conceptualised as a linear model, within which individuals would successfully progress from one distinct stage to another. However, Prochaska and colleagues have subsequently argued that it would be better conceptualised as a spiral pattern that illustrates how people can progress through the stages, but are likely to encounter relapses in which they will regress to an earlier position.

Most people taking action to modify chronic dysfunctional behaviour do not successfully maintain their gains on their first attempt. Relapse is the rule rather than the exception

across virtually all chronic behavioural disorders (Prochaska and Prochaska, 2002 p.380). Individuals may need particular support through the relapse stage, as those who have been able to progress from contemplation to action to maintenance may feel as though they have failed when they return to their previous adverse behaviour patterns. They may then resist consideration of subsequent behavioural change and regress to the pre-contemplation stage.

According to its developers, the TTM offers a 'more reliable, valid, and complex assessment of behaviour change than simple recording of compliance' (Prochaska and Prochaska, 2002, p.379). It has been used as a theoretical construct to understand the structure of intentional behaviour change, as well as a tool to help practitioners understand what will be the most effective intervention to bring about change. For instance, the developers argue that action-oriented therapies will be effective if a person is in the preparation or action stages, but will be ineffective or detrimental for individuals in the pre-contemplation or contemplation stages (Prochaska *et al.*,1992).

Validated assessments and scales

The TTM is supported by a number of questionnaires and scales that have been developed to determine an individual's stage of change readiness or motivation and therefore facilitate an understanding of the type of intervention that would be most beneficial either to improve their readiness, reduce resistance to change or to address the underlying issue. These are mostly in the public domain and free to use without permission. These measures include the University of Rhode Island Change Assessment Scale (URICA), a measure of motivational readiness to change that can be used to determine where an individual is along the stages of change. There are also a number of self-report measures which can be used in a variety of circumstances to assess, for instance, an individual's confidence in their ability to abstain from substance use and how tempted an individual is to engage in problematic health behaviours.

However, caution is needed in using such measures. These types of scales should inform rather than replace professional judgment, and those judgments should also be informed by a thorough knowledge of processes and indicators of change and an understanding of how these relate to the particular individual. They may help to assess a parent's psychological readiness to change, but they cannot predict whether that parent will or will not eventually succeed in making and sustaining change, particularly in the long term. Moreover, assessments based on self-reports are appropriate in a therapeutic relationship, but may not be so valuable in situations such as care proceedings where parents have a strong incentive to present themselves in the best possible light.

In addition, a number of systematic reviews have questioned the strength of the evidence in support of the TTM, particularly in relation to the use of a stage based approach to implementing interventions as a basis for behaviour change (see Bridle *et al.*, 2005; Littell and Girvin, 2005; Whitelaw *et al.*, 2000). Therefore caution is needed in considering the use of this model, particularly as the systematic reviews cited above show that the

standardised scales show no clear evidence of predictive validity. Moreover, the model is not informed by goal setting theory (discussed later in this chapter), which appears to provide a valuable basis for engaging parents in the process of identifying appropriate targets and monitoring the progress of change (Dawe and Harnett, 2007).

Issues concerning the use of the Trans-Theoretical Model of Change as a conceptual framework in child welfare cases

Morrison (2010) presents the TTM in a UK context as a practice tool designed to offer a 'common language about motivation for use between practitioners and service users, between different agencies or between supervisors and staff to understand and assess progress towards change' (Morrison, 2010, p.312). He argues that it can be applied in three practice contexts: first, on a voluntary basis, where parents request help with problems and engage with services themselves; second, where there are concerns about the welfare of a child and the family are aware that if they do not engage with services, statutory action will commence; and third, where statutory proceedings have already been decided upon.

There are, however, difficulties in translating a model developed in response to evidence concerning how individuals *intentionally* overcome adverse behaviour patterns into a context in which coercion plays a part. Parents in both Morrison's second and third categories may well perceive their engagement with services as an involuntary process, and this may lead them to respond in a conflicting manner, moving between withdrawal, disguised compliance and aggression (see Brandon *et al.*, 2008; Morrison, 2010). Girvin (2004) argues that some parents will be aware that being labelled 'unmotivated' can have serious consequences and may result in the removal of their children; therefore they may learn that it is advantageous to assert that they are working on their problems, when in reality this is not the case. In Girvin's view: 'The SOC [stages of change] model does not distinguish between responses that reflect genuine interest in behaviour change and those that are intended to appease caseworkers' (2004, p.900).

Parents who maltreat their children are also likely to have complex and multifaceted problems that may not fit into a single-stage classification such as that proposed in the TTM. Therefore over-reliance on the TTM classification system with its focus on readiness for change might offer a limited view of the challenges to be faced. Littell and Girvin (2005) argue that it is likely that readiness for change varies across problems, and parents who are ready to change one problem behaviour may not be as ready to change another.

The TTM remains a valuable framework for understanding the process of behavioural change as part of a broader assessment; however there is a danger that it could be too simplistic, or used too simplistically in child welfare cases. A model for smoking cessation involves only the individual concerned. In care cases at each stage it is the interrelationship between parent and child and the impact of the parental problems upon the child which are central; change will have much more far-reaching implications, and

relapse will impact not only on the parent, but also on a child who may have begun to trust in a safer environment.

To reflect the complexity of child welfare cases, a number of additional constructs have been proposed, specifically to facilitate understanding of parental engagement with services and readiness to change during the early stages of the process. These are discussed in the following section.

Adaptations to the Trans-Theoretical Model and alternative models

Hegarty and colleagues (2008) utilised and adapted the stages of change theory for work with female victims of domestic abuse and proposed a psychosocial model of readiness to change. The psychosocial model takes into account internal factors, including acknowledging abuse, perceived support from others, and self-efficacy or power (Cluss et al., 2006). It also takes into account external factors, including knowledge and skills of professionals supporting women, community and financial resources (Hegarty et al., 2008). The model involves the rating of these internal and external factors along a continuum.

Humphreys and colleagues (2011) explored how the psychosocial model of readiness to change could be applied to strengthening the relationship between mother and child following domestic abuse. The authors argued that the concept of readiness to change should include not only an individual's motivation to change, but also external factors such as the layers of intervention that are required by organisations, workers, mothers and children to lay the foundations that support the change process. The authors concluded that it was 'clear that there were a number of different dimensions to the change process for women that included: being beyond the immediate crisis; recognising that there was value in talking about aspects of the past with their children; and being in a position to be able to refocus on their children's needs. Children also have their own views and needs that should be independently heard as part of their change process' (Humphreys *et al.*, 2011, p181).

Littell and Girvin (2005) argue that the predictive validity of the stages of change theory is weak, and propose that motivation or readiness for change should be explored using a two dimensional model consisting of problem recognition and intention to change, rather than a single continuum or multidimensional construct. They examined the predictive validity of a number of domains associated with readiness for change among caregivers receiving home services following reports of child abuse or neglect, and found that initial problem recognition and intentions to change predicted 'a few improvements in individual and family functioning, along with significant reductions in the likelihood of additional reports of child maltreatment within one year' (Littell and Girvin, 2005, p.59). However the authors concluded that there were few advantages of an overall readiness score; that it is unlikely that greater problem recognition or intention to change results in better outcomes; and that it should not be 'assumed that initial problem recognition, intentions

to change, or apparent readiness for change determine who is most likely to benefit from treatment' (Littell and Girvin, 2005, p.76). Instead, the concepts should be used as an indication of the issues as part of a broader assessment process. Social workers should therefore conduct comprehensive assessments of parents' views about their problems; goals and values; levels of discomfort with the status quo; hopes about whether the situation can improve; and their opinions about available alternatives (Littell and Girvin, 2006).

A procedure for assessing parents' capacity to change (Harnett 2007)

Harnett (2007) develops some of these theories further and proposes a procedure to assist social workers in assessing parents' capacity to change, which addresses some of the limitations of current assessment. The limitations he identifies include the narrow focus on particular aspects of family functioning; the absence of social workers' views about the reliability and credibility of parents' reports in light of the potential bias towards responses which are perceived as being desirable; and the absence of observations of parent-child interaction. Where the evidence of parental capacity to change is equivocal, Harnett identifies reliance upon cross-sectional assessment as a serious weakness because it only provides a snapshot of family functioning at one point in time. To address these issues, Harnett proposes a procedure for assessing parents' capacity to change over a period of four to six months during which they are engaged in a brief, intensive intervention.

This extended procedure directly assesses parents' motivation and ability to change and requires practitioners to consider any further interventions that might be necessary and identify the level of support required for change to be maintained. This model of 'capacity to change' is less concerned with a parent's report of their intentions to change and more concerned with the direct assessment of actual change. The assessment model considers the attainment of goals as evidence of parents' capacity to change, and it is that allows for a better prediction of future family functioning.

Harnett's procedure for assessing parents' capacity to change involves four elements which are described below:

- conduct of a cross-sectional assessment of parents' current functioning;
- specification of targets for change derived from an assessment of current strengths and deficits in the family;
- implementation of an intervention with proven efficacy for the client group with a focus on achieving clearly specified targets for change; and
- objective measurement of changes in parenting

(descriptions adapted from Harnett, 2007).

The procedure does not therefore exclude the use of cross-sectional assessment of family functioning, but incorporates it as one element within an extended process.

Cross-sectional assessment of parents' current functioning

This stage involves a cross-sectional assessment following best practice guidelines to provide a baseline from which to assess change, and includes an assessment of problems in child and parent functioning and child-parent interaction. The cross-sectional assessment is based on an integrated theoretical framework of child development and family functioning. First the child has an assessment which covers physical problems and developmental delays in daily living skills, together with emotional, cognitive, social and behavioural functioning. Next, drawing on attachment theory, the parent-child relationship is assessed, with a focus on parents' capacity to be emotionally available to the child. An assessment of the parents includes gathering historical information about their own experience of abuse and the impact of this on current parenting practices; knowledge, attitudes and beliefs concerning childrearing; and their intellectual functioning. The ability of the parent to regulate their own emotional state is seen to be a key factor influencing their capacity to be emotionally available, to implement non-punitive discipline, and maintain regular family routines. Finally, the social ecology of the family is assessed, in particular an assessment of stressful life events, external demands and availability of support.

The cross-sectional assessment is used as a basis for developing a case conceptualisation. Social workers employ standardised psychological tests and scales as part of the cross-sectional assessment to provide a baseline on which to formulate an objective measure of parental change.

Specifying targets for change using goal setting theory

The next step is to work with parents to identify achievable goals that are meaningful in the context of the particular family undergoing assessment. Within this context, goal setting theory can help social workers understand how best to support parents to change behaviours. It can also help to focus parents on changing particular aspects of their behaviour and reduces ambiguity as to the extent to which they need to change, by establishing specific targets (Locke and Latham, 2002).

Key to the theory is that goals need to be meaningful to parents and perceived by them as being manageable. In this way, goal setting can bring parents 'on board' and give them a sense of being an active participant in the change process rather than having change thrust upon them. Dawe and Harnett's (2007) study of the Parents under Pressure (PUP) programme for methadone users highlights how setting meaningful goals is more likely to result in parents feeling committed to attempting change, whereas imposing goals on them may lead to frustration and resistance. A goal set tersely by someone else without any explanation will result in lower performance (Latham, Erez and

Locke,1988). Thus the model blends the collection of more objective data by which change can be measured with a strong relationship-based element.

Although the ultimate goal for parents may be to overcome their difficulties so that their child is no longer at risk of harm, goal setting theory suggests that parents will be more successful if smaller, interim goals are set in addition to an end goal (Latham and Locke, 2007). In this way, parents have manageable targets and are not set up for failure. Providing feedback to parents may also motivate them, helping them feel they are progressing towards each goal; the combination of goal and feedback has been found to be more effective than goal setting alone (Locke and Latham, 2002). In addition, the feedback process will give social workers the opportunity to assess progress for themselves as part of the assessment of parents' capacity to change.

A collaborative approach to goal setting is also important because individuals with low levels of self-efficacy (belief in their ability to succeed) have been found to be more likely to set lower goals for themselves than those who strongly believe they can be successful (Latham and Locke, 2007). Parents whose children are on the edge of care are often attempting to overcome multiple, entrenched difficulties and may have low levels of self-efficacy. By agreeing goals collaboratively, social workers can ensure that the goals are realistic yet not set too low. Low goal setting may mean that change is too slow to be sufficient to meet the timescale for the child.

Harnett proposes the use of Goal Attainment Scaling (GAS) (Kiresuk *et al.*, 1994; Ogles *et al.*, 2002) as a useful tool for identifying appropriate goals and monitoring parents' achievements, using a five point outcome rating system which permits quantification of the change achieved.

Implementing an intervention with proven efficacy for this client group with a focus on identified targets for change

Harnett states that the outcome of a capacity to change assessment will be significantly influenced by the delivery of an effective intervention, which addresses the multiple needs of the particular family being assessed. Parents whose children are on the edge of care are frequently faced with a complex web of problems; single issue programmes, which focus on one specific area of concern are less effective than multifaceted programmes that address the interlocking domains of family functioning that may need to change (see Dawe and Harnett, 2007). Chapter Six explores further some of the evidence-based interventions that have been shown to be helpful to parents and children in such circumstances.

Assessment of change

Following completion of the intervention, the social worker will evaluate the results of the standardised psychological tests administered pre-and post-intervention, the structured observation of parent-child interaction, using methods such as the Emotional Availability

Scales (Biringen, 2000) and the results of the GAS procedure. Changes on the standardised assessment measures and the extent to which goals have been attained provide the required evidence of capacity to change.

Employing different types of assessment to understand parents' capacity to change

Using the multiple types of assessment proposed by Harnett enables social workers to assess motivation, readiness and ability to change from a number of perspectives and across a period of time. Harnett's procedure for assessing parental capacity to change recognises that social workers are faced with the difficult task of producing sufficient evidence for the courts concerning families with complex needs who may be living chaotic lives, in circumstances where multiple factors will potentially impact upon parenting capability. By using different types of assessment over an extended period, this procedure for assessment provides a framework for considering capacity to change across a number of domains. The procedure applies a layer of formal objective assessment to highly individual circumstances and, as such, provides social workers with a framework for providing hard data that will support evidence based decision-making, and help counteract some of the recognised pitfalls of relying on professional judgment and experience alone (see Chapter One). Importantly, the assessment model is based on a theoretical framework that is derived from evidence-based models of child development and family functioning.

Limitations of the procedure for assessing parents' capacity for change

It is possible that different elements of the assessment procedure will provide conflicting evidence of capacity to change. For example, an interview will provide the parent's views on their ability and motivation to change, whilst an observation of parent-child interaction using validated scales may tell a different story. The evaluation of the evidence therefore needs to be clearly set out in the social worker's assessment. Where the practitioner considers that the evidence from validated scales should be disregarded, reasons should be given.

It is important to appreciate that although the tools used in Harnett's procedure for assessing parents' capacity for change can help to provide a broader evidence base for social worker decisions, no assessment procedure can provide conclusive evidence of a parent's motivation, willingness or ability to change. Moreover parents are free agents in the process of change, and may not follow anticipated pathways. Unforeseen events may also impact on the change process. The procedure should not be used mechanistically in a tick box approach, and should inform rather than replace structured professional judgment concerning whether parents are able to make sufficient changes within a child's timeframe.

Conclusion

Practitioners will find it difficult to understand whether parents are able and ready to overcome adverse behaviours unless they have some means of conceptualising what needs to change, how such change might be achieved, and the factors that promote or facilitate it. The best known models of change have been developed within health promotion services to facilitate understanding of single issues such as overcoming addiction to tobacco or alcohol. While these may provide a foundation for conceptualising how change might be modelled in child welfare, they need to be modified to reflect the complexities of child protection cases, including the potential impact of coercion and the pressure on parents to present themselves in a positive light. Harnett's dynamic model of assessing parents' capacity to change, through collecting baseline data, case conceptualisation, and then setting mutually agreed goals and monitoring progress towards them should provide the necessary evidence to inform structured professional judgments. Such judgments will focus on whether parents have made sufficient progress to demonstrate that there is no longer a likelihood of significant harm within a timeframe that is appropriate for the child.

Key points from Chapter Three

- In order to assess parental capacity to change, practitioners first need to formulate
 a case conceptualisation, mapping out the various external and internal factors
 that impact on parents' ability to meet their children's needs, and identifying their
 strengths as well as their weaknesses.
- Practitioners need to be mindful that parents' prior experiences may have led to severely insecure attachment styles and that these can affect their relationships with professionals. They may, for instance, be unwilling to open up or be suspicious and distrustful of authority.
- Models of change can offer helpful insights into how parents overcome adverse behaviour patterns, particularly where these are entrenched and the change process is complex. Models of change can provide useful conceptual frameworks for appreciating this complexity and prevent practitioners from relying on compliance as a proxy indicator of success.
- Although models of change can provide an enhanced understanding of the
 processes and mechanisms involved in behavioural change, they cannot predict
 whether an individual parent will or will not change or whether they will sustain
 progress in the long term.
- The best known models of change, such as the Trans-Theoretical Model (TTM) have been developed in response to evidence concerning how individuals *intentionally* overcome adverse behaviour patterns. There are difficulties in translating these into a context in which coercion plays a part.
- Parents who maltreat their children are likely to have complex and multifaceted problems that may not fit into a single-stage classification. Over-reliance on

- classification systems that focus on readiness for change alone might offer a limited view of the challenges to be faced.
- The procedure for assessing parents' capacity to change proposed by Harnett
 offers a useful model by which change can be assessed over time. This involves
 the use of standardised screening tools before and after an intervention, together
 with identification of goals to be agreed between the social worker and the parent
 and met within a specific timeframe.

Chapter Four: Ambivalence, denial and unwillingness to change

Introduction

Many parents in families where there are significant child protection issues and children are on the edge of care may deny that problems exist. They may be overwhelmed by the struggle to cope with issues such as poverty and poor housing, which provide a context that increases the likelihood of abuse and neglect (Brandon *et al.*, 2012; Merritt, 2009). They may not perceive that personal changes are necessary and can vehemently defend adverse behaviour patterns (Prochaska and Prochaska, 2002; Ward, Brown and Westlake, 2012). Assessing parental capacity to change in these situations requires empathy and relationship skills (Forrester *et al.*, 2012; Holland, 2000); it also requires professional objectivity and understanding of the common pitfalls of intuitive reasoning (Munro, 1999), and a working environment in which decisions can be openly discussed and tested out with colleagues and staff receive supportive supervision in which mistakes can be acknowledged without fear of censure (Munro,1999,2005; Kirkman and Melrose,2014; for further discussion see Chapter One).

This chapter explores the factors underlying denial or unwillingness to change, and discusses how parents might be helped to accept that there is a problem and move forward. The following chapter looks at the more positive side of the picture and explores how parents can become motivated to change and engage with the services designed to support them through this process.

How common is denial or unwillingness to change?

Parents who appear to deny that that there is a problem or to be unwilling to make changes that might protect their children from future harm are sometimes described as 'resistant, or highly resistant to change' (Fauth et al., 2010, passim). A recent knowledge review concluded that there were no clear or consistent statistics on the numbers who fit this description (ibid., p.27); moreover prevalence rates will be highly variable as behaviour patterns will change as parents move in and out of this stage. However, Farmer and Lutman's (2012) study of neglected children returning home from care found that 39% of parents actively resisted or attempted to sabotage interventions from professionals, and Dawe and Harnett found that 36% of families remained at high risk of child abuse and neglect after receiving support from the Parents Under Pressure (PUP) programme (2007, p.1185). Moreover, Brandon and colleagues (2008, 2009) found that between 66% and 75% of parents involved in serious case reviews had shown disguised compliance, unwillingness to change and/or ambivalent or selective cooperation with services. Cases before the courts are more likely to involve parents who are unable or unwilling to change; where previous involvement with children's services under a child protection plan or the formal pre-proceedings process has helped parents make and

sustain changes that ensure their children are now adequately safeguarded from harm, applications to court are unnecessary.

Disguised compliance, failure to cooperate, and unwillingness to change

Brandon and colleagues' studies of serious case reviews, where children have been seriously injured or died, found extensive false compliance. In these instances, on the surface parents appear to engage with services to avoid raising suspicions, but in reality are not making progress towards improving their capacity to provide a nurturing home for their children:

Apparent or disguised cooperation from parents often prevented or delayed understanding of the severity of harm to the child and cases drifted. Where parents engineered the focus away from allegations of harm, children went unseen and unheard (Brandon *et al.*, 2008, p.4).

Forrester and colleagues (2012) argue that cases in which parents consciously and systematically cover up deliberate abuse are relatively rare. However, this is more likely to occur in situations of such severity that they are followed by a serious case review, and indeed false compliance and deliberate concealment were factors in the recent high profile deaths from abuse and neglect of Victoria Climbie (Laming, 2003), Peter Connelly (Haringey LSCB, 2010) and Daniel Pelka (Coventry LSCB, 2013).

While systematic and conscious concealment of deliberate abuse may be rare, denial of adverse behaviour patterns and/or their impact on children is a recognised response of parents and other adults who have not yet understood or acknowledged the need to change, or who are overwhelmed with their difficulties and fearful about their ability to do so. It may manifest itself in anger or hostility and/or refusal to admit that there may be a problem. For instance, perpetrators of domestic abuse may avoid taking responsibility for their behaviour by denying or minimising its severity, or by blaming their partner or external circumstances beyond their control. As such, 'the batterer is preoccupied with his own needs and wants, and views his wife as another "object" to control and manipulate for his own advantage' (Gondolf, 1987, p.341). For these individuals there is no problem, and therefore no reason to change or to engage with services. Parents with mental health problems or learning disabilities may also refuse to accept that there is a problem because of the stigma attached to their condition (see Booth and Booth, 1993; Hinshaw, 2005). Within certain black minority ethnic and refugee (BMER) communities there can be acute shame in admitting difficulties such as domestic abuse or mental health problems and stigma attached to seeking the assistance of social services (Bhardwaj, 2001; Qureshi et al., 2000). This may lead families to hide their problems from the outside world, making it difficult for social workers to assess how parents are coping.

Denial may also show itself in more passive resistance, for instance in a failure to undertake the actions specified in child protection plans, to attend appointments with professionals, or to complete a programme of treatment. Ward and colleagues found that written agreements made between parents or carers and agencies, and intended to formalise safeguarding arrangements, could be an ineffective means of protecting children from harm because their terms were frequently broken, often without consequences (Ward, Brown and Westlake, 2012, pp.168-169; see also Farmer and Lutman, 2012). Daly and Pelowski (2000) found that at least one in five perpetrators of domestic abuse fail to complete treatment, and that the drop-out rate is as much as 99% for some programmes; Damashek and colleagues (2011) calculate that between one in five and one in three participants fail to complete home based programmes designed to prevent child abuse and neglect (p.9).

Adults who deny that there is a problem or that their behaviour has an impact on their children, may exhibit false compliance, and go through the motions by keeping appointments, but fail to collaborate by genuinely participating in treatment (see Dawson and Berry, 2002). Ward and colleagues found that abusive and neglectful parents who denied the impact of their behaviours had often had long-standing involvement with children's social care and were familiar with child protection processes; as a result, they were better able to persuade practitioners that they could safeguard their children, regardless of their readiness or ability to change their lifestyles (Ward, Brown and Westlake, 2012, p.135). Ekendahl (2007) found that adult substance misusers who generally spent their time on the streets might occasionally go through the motions of entering treatment if they needed to do so to fulfil the criteria for obtaining help with housing (p.250).

Denial and passive resistance are not the only reasons why individuals do not seek or use available help, fail to attend appointments or drop out of programmes (see below), although they can be a major contributing factor. It is also always important to bear in mind that denial may also mask a genuine lack of insight and understanding of concerns that require greater clarification.

Factors underlying unwillingness to change

We have already seen that a small number of parents feign compliance with services in order to deflect suspicions and conceal systematic and deliberate abuse. However, in the majority of cases, false compliance and feigned engagement are thought to be more subtle – and more complex. A number of authors have argued that combinations of interlocking factors contribute to parents' denial that a problem exists or unwillingness to make changes, and that these broadly fall into two groups:

internal factors present within the individual and their family; and

• external factors present within parents' social contexts, covering not only their immediate environment, but their relationships with the external world, including their encounters with social workers and other professionals (see Forrester *et al.*, 2012; Gladstone *et al.*, 2012; Randolph *et al.*, 2009).

The range of interlocking factors that underlie an individual's motivation to change and engage with services can be seen as presenting the obverse but more positive side of this picture, and are discussed in the next chapter.

Internal factors

Failure to understand that a problem exists

Parents living with learning disabilities or with mental health problems, including impaired personality function, may not deny the need for change. They may simply not recognise or understand the types of behaviours they are exhibiting, or why they can have a damaging impact on children (Booth and Booth, 2004; Brophy *et al.*, 2003). Booth and Booth (2004) state that parents with learning disabilities are 'anywhere between 30 and 60 times more likely to be the subject of a care order application than their numbers in the general population would warrant' (p.10). Likewise, Brophy and colleagues (2003) found that up to 43% of social work cases that lead to care proceedings involve parents with mental health problems.

Increased isolation

Taylor and colleagues (2008) explored some of the challenges of engaging and accessing parents and children in families experiencing interlocking problems, in which the key components are alcohol or substance misuse. Their focus was parents who drop out of alcohol misuse programmes, but many of these issues are also common to parents struggling with mental health problems, substance misuse and domestic abuse, partly because these often co-exist, but also because the barriers to acceptance and engagement are often the same.

The fear of stigma experienced by parents facing these problems, together with the potential child protection implications, may lead them to close off contact with those outside the immediate family, including professionals. A vicious circle may then develop, in which alcohol or substance misuse may become a way of coping with the pressures of maintaining secrecy, but also exacerbate the problems. Poor mental health may be a compounding factor (see Hinshaw, 2005), with existing feelings of depression or paranoia magnified by the fear of stigma and the perceived need for secrecy. In such families secrecy may become the dominant concern: the family may be split between those who know what is happening and those who do not, and children may feel unable to talk to others about their experiences (Taylor *et al.*, 2008).

Barnard and Barlow (2003) interviewed 36 children whose parents misused substances and found that they experienced a 'weight of forced silence' (p.54), in which they

struggled with the contradictory experiences of knowing that there was a problem, yet being unable to raise questions at home because the subject was never discussed, or outside the home because of the unspoken need to maintain family loyalty. Very similar dynamics have also been found in families in which domestic abuse (Humphreys *et al.*, 2011) or sexual abuse (Bradley and Wood, 1996) is an issue, for the same forced silence often exists between family members (perpetrators, victims and non-victimised children), and the same perceived need to protect the family by keeping its secrets hidden from the outside world. Taylor and colleagues (2008) argue that the need for secrecy alters the family dynamics, because it undermines trust and leads to fluctuating emotional relationships between parents and children.

Shame, ambivalence and a lack of confidence

Forrester, Westlake and Glynn (2012) identify three further internal factors that contribute to denial and unwillingness to change within individuals: shame, ambivalence and a lack of confidence. Shame is related to the stigma surrounding many of the behaviours, past experiences and current situations of parents who have dealings with child protection services. Ambivalence refers to the conflicting emotions that parents may feel when they perceive both positive and negative consequences of overcoming behaviour patterns such as alcohol misuse, which professionals may view in terms of the negative impact on their children, but which they may see as also offering some comfort and protection from the reality of their situation. Parents who experience acute ambivalence may not yet have accepted a need to change. Some authors have suggested that the threat of legal proceedings may reduce ambivalence and act as a positive driver of change (see for instance Gregoire and Burke, 2004; Hiller et al., 1998; Joe et al., 1999). This, in part accounts for the effectiveness of the formal pre-proceedings process in diverting cases from care proceedings (Masson and Dickens 2013). On the other hand, parents who lack self-confidence may be ambivalent about their ability to change and this may be at the root of their denial (see also Forrester et al., 2012; Saint-Jacques et al., 2006; Taylor et al., 2008).

Morrison (2010) argues that practitioners need greater understanding of the internal factors that may lie behind the apparently perverse responses of some parents when involved in child protection interventions:

Much is at stake for these families: fear of exposure, stigma, removal of their children or even prosecution. Fearful and anxious at the intrusion of external agencies, they may be defensive, angry and unwilling to acknowledge the reality of any difficulties. Negative previous experiences of statutory agencies may exacerbate this. Blame and responsibility may be externalised in terms of 'we were fine until you lot came along' (pp.314-5).

Alternatively pre-contemplation may be expressed in a more passive or helpless response, where parents cannot comprehend what has happened to their child, and seemingly do not react to professionals' high levels of concern. Such responses may

result from shock, depression, mental illness or learning disabilities, all of which limit the parent's capacity to understand the concerns (Morrison, 2010).

External factors

Resistance to the involvement of social workers

Resistance to the involvement of social workers is not the same as resistance to change, though the two are often confused. Forrester and colleagues (2012) explored the factors that underlie parental resistance to the involvement of child and family social workers in families. They conceptualise resistance to involvement as an active behaviour that encompasses: 'any form of non-co-operation from parents, including apparent co-operation that masks issues of concern, not engaging, violent or threatening behaviour and other manifestations of non-engagement' (p.118) and argue that 'parental lying -whether conscious and systematic or simply minimizing the extent or impact of an issue - is almost omnipresent in child protection work' (p.123). In their view 'the context of child protection involvement is ...in itself likely to create resistance' (p.120). This is because there is a particular imbalance of power in child protection work in that the parent is aware that the social worker's role is, at least in part, to assess and make judgments upon their parenting capacity and that ultimately social workers have the authority to remove the children.

Parents' responses also need to be understood in the light of their own history of engagement with professionals, their early life experiences of shame and anger, their current social context and the ways in which they habitually relate to authority figures. Experiences of discrimination and disadvantage in the past and in the present will be powerful factors behind parents' responses to social work interventions. Past experiences of involvement with children's social care may reawaken feelings of powerlessness, and lead to greater resistance to change.

Dumbrill (2006) explored the part that power relationships between parents and social workers played in the development and maintenance of parents' resistance to social work involvement. He used a grounded theory approach to analyse in-depth interviews with eighteen parents who had been involved in child protection interventions and to build a model representing how parents perceived and reacted to such interventions. The power imbalance was central to this model, but Dumbrill found that parents might perceive such power as being exercised *over* them, in a coercive or penalising manner, or *with* them as a form of support.

Many parents may be mindful of the power imbalance and approach a relationship with social workers with caution; however, those who perceive social workers as exercising power *with* them are more likely to respond by 'working with services in what appear to be genuine and collaborative relationships' (Dumbrill, 2006, p.33). If power is perceived as supportive, then parents can respond positively to small words of encouragement (*ibid.*, p.31). They also appreciate it when workers exercise 'power with them through

advocacy' by, for instance, supporting their case against landlords who are unresponsive to their needs (*ibid.*, p.31).

On the other hand, those parents who perceive the relationship as characterised by power *over* are more likely to be fearful of social workers whom they perceive as quickly categorising their cases to fit pre-established plans. In such circumstances there is little room for dialogue, and parents feel that they have little opportunity to challenge workers' opinions and plans even if these seem illogical and against the best interests of their children. Parents tend to respond to perceptions of power being used over them by 'fighting' through openly challenging or opposing workers in court or by 'playing the game' by feigning cooperation. In these circumstances the power struggle becomes the focus of the relationship with the social worker, and gets in the way of achieving the objectives of the intervention (Dumbrill, 2006).

It is apparent from this and other studies (see Forrester *et al.*, 2012; Forrester and Harwin, 2011; Humphreys *et al.*, 2011; Miller and Rollnick, 2013; Platt, 2008; Scott, 2004) that practitioners need to be aware of their own role in exacerbating or reducing resistance to engage with children's social services. Several writers have argued that the way in which the social worker approaches the family can increase or decrease the parent's resistance. Forrester's studies of social workers' communication styles found that the majority adopted 'a highly confrontational' approach in child protection work, and that confrontation tended to create high levels of resistance (see Forrester and Harwin, 2011; Forrester *et al.*, 2008a). Dumbrill (2006) found that descriptions of power being used *over* parents were 'far more evident' than descriptions of power being used *with* them (p.32). Forrester and colleagues (2012) argue that social workers need to learn how to interact with parents in such a way that they do not exacerbate or cause resistance, their first duty being to cause no harm (p.124); they point out that an approach which underlines parental deficits is more likely to lead to resistance than one which focuses on and reinforces positive behaviours and acknowledges the challenges parents are facing.

Support from other agencies

In times of need, other professionals may be seen as less threatening than social workers, and parents and children may choose to circumvent children's social services by engaging with them (see Stanley *et al.*, 2003). The evidence concerning parents' resistance to social work involvement, and the tendency of some to seek support from elsewhere, also emphasises the importance of social workers developing strong interagency links with professionals such as health visitors and practitioners in adult services (see Ward, Brown and Westlake, 2012).

Advocacy is a promising approach being implemented to promote the active engagement of parents whose children are on the edge of care (see Department for Education, 2014a). Holt and colleagues (2013) suggest that effective advocacy provides a means of moderating the 'imbalances of knowledge and entitlement' faced by parents involved in

formal child care proceedings (p.166). There is also some evidence to suggest that increased participation by parents in decision-making processes leads to improved engagement (Featherstone *et al.*, 2011; Featherstone and Fraser, 2012; Healy and Darlington, 2009). Advocacy may therefore provide a means of reducing resistance and may re-focus parent-social worker interaction on the substantive problems which need addressing rather than the problematic relationship between the parties.

Motivational Interviewing: strengths and weaknesses

A number of authors argue that motivational interviewing can reduce unnecessary confrontation and help parents move on from a position where they deny that a problem exists and are unwilling to change or engage with services (see Bowen and Gilchrist, 2004; Forrester *et al.*, 2012; Zalmanowitz *et al.*, 2013). As Morrison (2010) states: 'change does not start when parents enter the action stage; it starts when we enable an anxious, fearful, or angry parent to make the first steps along the pathway of [contemplating the need for change]' (p.321). Harnett (2007) argues that clear specification of goals that are meaningful to parents and a promise that progress will be acknowledged will facilitate this process.

Motivational interviewing (MI) is a counselling method, originally developed in response to the treatment of problem alcohol users. It focuses on exploring and resolving a person's ambivalence about change, and accepting that ambivalence is a normal part of the change process. The core value of MI is that it does not impose change; rather it supports change in a manner which is congruent with the person's own values and concerns (Miller and Rollnick, 2013). MI was developed as a clinical tool for individuals who are not yet ready for change, to help them move forward (Miller and Rollnick, 2013). MI is characterised by a particular approach, based on three key elements: collaboration between the therapist and client; evoking or drawing out the client's ideas about change; and emphasising the autonomy of the client (Miller and Rollnick, 2013).

There is an extensive and increasing evidence base relating to the effectiveness of MI. A meta-analysis of 72 clinical trials across a range of target problems found some evidence that MI could be an effective method of approaching addictive and health-related behaviours (Hettema, Steele and Miller, 2005). The strongest support for the efficacy of MI was in reducing substance misuse, with slightly larger average effect sizes³ (0.51) for illicit drug use than for alcohol abuse (0.41) (*ibid.*, p.102). However there was a wide variability in effect sizes across studies, even within the same problem areas, indicating that the way in which MI is delivered can have a substantial impact on outcomes.

_

³ An effect size of 0.2 is considered to be small; an effect size of 0.5 is medium and one of 0.8 or greater is large (Cohen, 1969) see Chapter Six.

Across all the studies covered by the meta-analysis, the average effect size of MI was relatively high (0.77) within the first month after treatment, but it diminished fairly rapidly, to 0.30 within six months and 0.11 after a year (*ibid.*, p.100). The exception was in those studies where MI was used in promoting intervention engagement, retention and adherence at the beginning of an active treatment programme. In these circumstances the effect was maintained or increased over time, and averaged 0.6 (Hettema, Steele and Miller, 2005, p. 101).

There is also some slight evidence showing that MI could be useful for men undertaking court mandated treatment for domestic abuse. For instance, Zalmanowitz and colleagues (2013) undertook a large quasi-experimental study⁴ comparing 211 men who attended a domestic violence prevention programme in a large Canadian city. About half of them (105) received MI before attending the programme, and about half (106) did not. All of the men in this study showed positive change following therapy, and those who received MI did better than those who did not. However the differences were not sufficient to rule out the possibility that they had occurred by chance, and none of the participants in this study had initially shown extreme levels of violence and distress.

Individuals who respond to MI may have more discrete and less entrenched difficulties than the multi-layered problems that can overwhelm parents whose children are on the edge of care. Di Clemente and colleagues (2008) argue that motivational interviewing may not be so effective in a child welfare context because: 'brief motivational interventions prior to treatment have not always improved treatment engagement and outcomes in populations of drug-abusing individuals who are poor, minority and less educated with multiple problems' (p.27). Nevertheless, parallels have been drawn between the values of MI and child and family social work (Forrester *et al.*, 2008a). Forrester and colleagues (2012) argue that in child welfare cases, MI can be used to increase parental engagement and to address the factors underlying parental resistance to involvement with child and family social workers. The authors discuss five principle causes of parental resistance: social factors; individual and family factors; shame; ambivalence; and lack of confidence. They propose MI as a useful skill for reducing social worker contribution to resistance as well as other causes, and identify three strategic aims in working in child protection using MI:

- a focus on the child's welfare and possible harm to the child;
- a focus on engaging the parent;
- a focus on eliciting 'change talk' to resolve ambivalence about behaviour change.

A randomised controlled trial of MI in child protection work in the UK is currently underway. The study explores whether training and supervision in MI helps social

⁴ A two-group non-randomised comparative trials with a relatively high level of evidence (Level B, see Chapter Six).

workers to engage parents, and whether this is related to improved outcomes for children. The project is due for completion in 2014 (see Forrester, forthcoming).

The Family Partnership (FP) Model

A number of studies have identified how the professional relationship between the social worker and the service user is central to effective practice (see for instance Barlow and Schrader McMillan, 2010: Sudbery, 2002). The Family Partnership Model attempts to address the difficulties that practitioners encounter in developing such relationships under adverse circumstances, such as when parents are hostile or frightened that their children might be removed and are reluctant to engage with social workers. The model was developed to 'provide practitioners working with families with an explicit and detailed understanding of the dynamic processes of helping' (Harnett and Day, 2008, p.81). Partnership is at the heart of these processes, and this will only be successfully established if practitioners are able to communicate 'genuine respect for, interest in and commitment to parents, regardless of the conditions that have brought about statutory involvement' (Harnett and Day, 2008, p.81). Effective partnership working is characterised by: working closely together with active participation and involvement: sharing decision-making power; recognition of complementary expertise and roles; sharing and agreeing aims and how to achieve them; negotiation of disagreement; mutual trust and respect, openness and honesty and clear communication (Harnett and Day, 2008).

Barlow and colleagues (2007) evaluated the effectiveness of home visiting delivered by health visitors using the Family Partnership Model⁵. The study involved a sample of 131 women recruited through GP practices, who had been identified as vulnerable and specifically at risk of abuse and neglect (e.g. with mental health or housing problems), 67 of whom received the home visiting programme and 64 standard services. Visits took place from 6 months antenatally to 12 months postnatally and were designed to promote parent-child interaction. At 12 months small differences favouring the home visited group were observed on an independent assessment of maternal sensitivity (p <0.04) and infant cooperativeness (p<0.02), but this was no longer apparent at 3 years (Barlow et al., 2008). No differences were identified on measures of maternal psychological health attitudes and behaviour, infant functioning and development at either 12months or 3 years. Slightly more of the intervention group infants became the subject of child protection proceedings, or were removed from home; this was thought to relate to improvements in health visitors' sensitivity to abuse and neglect (Barlow et al., 2007). Kirkpatrick and colleagues (2007) conducted 20 interviews with women who had completed this programme and found that, although they had initially had negative

⁵ This was a multicentre randomised controlled trial with a high level of evidence (Level A, see Chapter Six).

expectations of health visitors and social workers, they greatly valued the relationships that had developed and thought that they had benefited by 'increased confidence, improved mental health, better parenting, improved relationships and changes in their attitudes toward professionals' (p.32).

The Family Partnership Model was also evaluated as part of the European Early Promotion Project (EEPP)⁶ (Davis *et al.*, 2005; Puura *et al.*, 2005). Similar improvements were found in mother-child interactions, particularly in Greece (Davis *et al.*, 2005). In the UK, mothers who participated in the programme were found to be 'more responsive towards their children, provided more appropriate play material, had a better relationship with the children, were more involved and used less control than comparison mothers', although the average effect size was small (0.27) and they did not show significant changes on all the measures used (Puura *et al.*, 2005, p.90).

The Family Partnership Model focuses on the development of a collaborative partnership with parents which is perceived as inextricably bound up with the specific tasks of the helping process. These include 'exploration, understanding, goal setting, strategy planning, implementation, review and ending' (Harnett and Day, 2008, p.81). These tasks are integral to Harnett's (2007) procedure for assessing parental capacity to change (see Chapter Three). The Family Partnership Model offers practitioners an evidence-based approach for developing the effective relationships with parents that are a key element in such assessments (see Harnett and Day, 2008). However, although the trials identified by our literature search show some encouraging results in terms of improved relationships with professionals, they do not show a strong or enduring impact on parent-child interactions.

Family Group Decision-Making

There is considerable evidence to suggest that involving parents and their wider extended family in the decision-making processes can decrease parental resistance to involvement with social workers by reducing their feelings of powerlessness within the context of statutory interventions and court proceedings. Family Group Decision-Making (FGDM), sometimes referred to as 'family group conferencing', was introduced in the UK in the 1990s, with the support of the Family Rights Group. It represents a departure from traditional decision-making models in child welfare, where there can often be an emphasis on expert knowledge and skills within an adversarial context (Barnsdale and Walker, 2007). FGDM was first developed in New Zealand, where it is now mandatory,

.

⁶ This was a large multi-centre non-randomised comparative trial in five European countries with a relatively high level of evidence (Level B, see Chapter Six).

and has since been applied in a child welfare context in more than 150 communities worldwide (Crampton, 2007). FGDM is defined as 'a clear process bringing together professionals and the family, including extended family, in a meeting to resolve issues of child care and protection' (Morris and Connolly, 2012, p.42). Its core components are:

- the family has the services of a co-ordinator, independent of their case management, to assist in the facilitation of the meeting;
- the family has private family time during the meeting when they can make decisions and plan for the care and safety of their child without professional influence; and
- there is a commitment to respecting the family plan unless a child is placed at risk of harm (*ibid.*, p.42).

Our search identified two extensive literature reviews, but no rigorous systematic reviews of studies reporting on the impact of FGDM. Barnsdale and Walker (2007) undertook a comprehensive literature search and interview study to examine the use and impact of FGDM within a UK context. Morris and Connolly (2012) reported on an extensive review of international literature on FGDM that forms the basis of an annotated bibliography, collated for the American Humane Association (Burford *et al.*, 2009). Both reviews appear to have been wide-ranging, but neither gives details of exclusion (or inclusion) criteria.

Engagement and satisfaction

Both reviews found considerable evidence to show that FGDM is effective in engaging the wider family in decision-making; that families value the opportunities to become involved and feel positive about their experiences; that they become more committed and engaged with plans for the child; and that FGDM opens up new possibilities for the care and protection of children within family networks. They also found evidence that FGDM can increase the role of fathers and male and paternal relatives (Barnsdale and Walker, 2007; Morris and Connolly, 2012; see also for instance Holland *et al.*, 2005; Titcomb and LeCroy, 2005). A further study evaluating a project in North West England, and not covered by either of these reviews also shows that FGDM can be effective for BME[R] families as it is: 'respectful of family culture and prioritises the voice of the family over the voice of the professional, thereby dealing with many of the issues raised by BME[R] families' (O'Shaughnessy *et al.*, 2010). The emphasis of FGDM on communication is also viewed as being important to those BMER families requiring interpreters, who are not always available when other mainstream services are used (*ibid.*).

Although there is good evidence that FGDM is viewed positively, is effective in engaging with family members and in facilitating the preparation of plans, there is less research evidence concerning successful implementation of plans and outcomes for the children concerned (Barnsdale and Walker, 2007, p.102).

Children's participation and outcomes

Concerns continue to be raised about whether children's voices and those of vulnerable family members such as victims of domestic abuse are adequately heard in an FGDM context which can be easily dominated by professionals or adults from the extended family. The studies reviewed by Barnsdale and Walker (2007) suggest that 'children and young people frequently attend FGDM conferences; however, a 'significant minority' can find themselves excluded from the process itself' (p.40). Morris and Connolly's (2012) analysis is more positive, and points to the value of an advocate being present at decision-making meetings, to support the child (p.47).

Although the research evidence is limited, a number of the studies reviewed by Morris and Connolly (2012) report a range of positive outcomes for children and young people following FGDM. These include increased support by family members and better family relationships (Staples, 2007) and increased rates of relative care: for instance, Titcomb and LeCroy's (2005) study of 291 families who completed the programme in Arizona found that the numbers of children placed with relatives increased from 47% at referral to 77% after the FGDM meeting (see also Morris and Connelly 2012, p.46).

However two studies which have compared the outcomes of family group conferencing with traditional child protection investigations and child welfare services have produced less positive findings. A randomised controlled trial⁷ undertaken in America (Berzin, 2006) which compared 197 families receiving FGDM with 126 families assigned to traditional services, found no significant differences in outcome, although the results showed 'trends that suggested higher rates of maltreatment, more placement moves and higher rates of service refusal' for children in families receiving FGDM (Berzin, 2006, pp.1455). Similarly, a three year follow-up of 97 children involved in family group conferencing in Sweden⁸ who were compared with a random sample of 149 children from families that received traditional services, found that young people involved in FGDM were more likely to have further substantiated re-referrals for maltreatment (60% vs. 40%); were more frequently placed in out-of –home care (42% vs. 21%); and spent more time in care (205 days vs. 103 days) (Sundell and Vinnerljung, 2004).

Outcomes such as these may appear disappointing, but they may be indicative of greater involvement and concern by members of the extended family following FGDM (see Berzin, 2006, p. 1456), in much the same way that the evaluation of the Family Partnership Model found increased sensitivity of professionals to abuse and neglect (Barlow *et al.*, 2007). The Swedish study found that children in the FGDM group more often placed with relatives (22% *vs.* 3%) and that 'the extended family members seem to

_

⁷ A study with a high level of evidence (Evidence Level A, see Chapter Six)

⁸ A two group comparative study with a relatively high level of evidence (Evidence Level B, see Chapter Six)

have submitted knowledge of the parents and children that the CPS did not have before..., leading to proposals in the plans for services on parental substance abuse and children's problems in school' (Sundell and Vinnerljung, 2004, p.281). There is also evidence in this and other studies that, on average, families referred to FGDM have more serious problems than those referred to traditional services: 71% of the FGDM group and 51% of the comparison group had previously been referred to children's services because of child protection concerns (Sundell and Vinnerljung, 2004, p.274. See also Cashmore and Kiely, 2000; Lupton and Nixon, 1999). Greater levels of need may be one reason for negative outcomes.

Engagement of professionals

A number of research studies confirm that the vast majority of plans made in FGDM meetings are approved by professionals. However, only about two thirds appear to be fully implemented (Barnsdale and Walker, 2007, pp.42-3). One Australian study (Cashmore and Kiely, 2000) which found that most of the plans made during New South Wales FGDM conferences were only partially carried out, identified the main reasons for partial (or non-)completion as: failures among family members to change unhelpful behaviours (particularly in relation to drug and alcohol use); changes in circumstances; and statutory agencies' slowness or inability to perform their duties. Both social workers and families viewed a joint lack of commitment to plans and the resulting failures in delivery as the main problem with the FGDM approach. A number of 'key stakeholders' interviewed by Barnsdale and Walker (2007) confirmed this point, indicating that there could be difficulties in implementing plans because either the family or social work services did not keep to the agreement.

This issue is indicative of one of the major difficulties identified concerning FGDM: resistance from professionals to the participative practice it encompasses, and difficulties in introducing innovative service developments into a risk averse, professionalised system (Morris and Connelly, 2012, p.47; see also Brown, 2007). Marginalisation and poor implementation of the FGDM model may be behind other problems noted such as: inadequate preparation, limiting family attendance, professionalisation of the conference, inadequate resourcing and information provision, and reluctance to refer families (Barnsdale and Walker, 2007). An important issue to resolve is the inherent 'tension between the central tenets of the FGDM model and the role of social care professionals and traditional decision-making processes' (Barnsdale and Walker, 2007, p.23), and difficulties in bringing it into mainstream practice. Barnsdale and Walker (2007) point out that 'when FGC becomes procedurally or legally required, the paradoxical prospect is raised of families being required to take part in an 'empowering' process which aims to increase self-determination' (p.3).

When parents cannot change within a child's timeframe

Where parents appear to be denying the need for change and/or are ambivalent about its benefits and there are concerns that children are suffering or likely to suffer significant harm, assessments of parenting capability should be used to explore their current functioning, identify what approaches have been employed to engage, motivate and help them to change, and show what approaches have been made to the extended family.

A range of interlocking factors lie behind some parents' refusal to accept that a problem exists and their apparent resistance to change when faced by child protection concerns. Internal factors within the parent may be exacerbated by external factors, including the approach taken by professionals. Changes in the way that these issues are approached, perhaps through improving practitioner understanding of the dynamic processes of helping, and/or through further development of motivational interviewing and greater involvement of the extended family in participative decision-making may prove to be fruitful ways forward.

Nevertheless, in child protection work, the first responsibility is to ensure that children are safe. Not all resistance to social work involvement is engendered by poor relationships between the parent and the social worker, or by the failure of professionals to accept participative decision-making, and not all parents are able to move on from denying that a problem exists, or being unwilling to change it, to taking action to resolve it within a child's timeframe. Some authors argue that a small number of parents are so far from accepting the need to change entrenched and damaging behaviour patterns that separation is likely to be the only option for the present. Gondolf (2002) and Scott (2004) put forward this argument in cases of extreme domestic abuse where the perpetrator shows a pervasive pattern of disregard for and violation of the rights of others (an antisocial personality disorder). Forrester and Harwin (2008) draw similar conclusions in cases of substance misuse where there is also violence in the home.

Conclusion

There is insufficient research concerning how many parents whose children are on the edge of care persistently deny that a problem exists or how long it may take them to acknowledge the need for change. Their situations, and those of their children, are amongst the most complex, and are likely to take up a disproportionate amount of practitioners' time and energy (Kirkman and Melrose, 2014). Practitioners who are assessing parental capacity to change in these circumstances require opportunities for reflective discussion with others and extensive support from supervisors, designed to help them explore and test out their case conceptualistion (see Chapter One).

The process by which parents overcome unwillingness to change and resistance to the involvement of social workers may prove to be lengthy, and should not become a reason

for delaying decisions concerning alternative permanence plans if these may ultimately be necessary to the child's long-term wellbeing (see Brown and Ward, 2012). Nevertheless, there is evidence to show that some parents with extensive and complex problems who have previously denied that change is necessary, do eventually succeed in overcoming their difficulties and go on to provide nurturing homes for their children (see Ward, Brown and Maskell-Graham, 2012; Ward, Brown and Westlake, 2012). The next chapter explores how such parents become motivated to change and engage with supportive services.

Key points from Chapter Four

- False compliance, failure to cooperate and resistance to social work involvement are common features of parents involved in child protection interventions.
- Mental health problems including impaired personality functioning and/or learning disabilities can reduce the ability of parents to understand the impact of their behaviour on children's wellbeing or to acknowledge the need for change.
- Apparent unwillingness to change can reflect internal factors such as shame, ambivalence about the need to change, and lack of confidence about capacity to change.
- External factors such as the imbalance of power, if not handled carefully, can compound and exacerbate resistance to social work involvement.
- The professional relationship between the social worker and the service user is central to effective practice. The Family Partnership Model was developed to provide practitioners working with families with an explicit and detailed understanding of the dynamic processes of helping, and may facilitate the development of supportive and more effective partnership working with parents.
- A sensitive social work approach, based on principles employed in Motivational Interviewing, can reduce resistance and help parents contemplate change.
- Family Group Decision-Making involves relatives and others in sharing responsibilities for addressing children's needs and gives families a real opportunity to make their own decisions about how to solve family problems; however the research evidence does not indicate that it prevents further maltreatment.
- Where there is no acknowledgement of a problem, in families where there are:
 perpetrators of sexual abuse; extreme domestic abuse where the perpetrator
 shows a pervasive pattern of disregard for and violation of the rights of others;
 there is both substance misuse and violence in the home; and/or where parents
 consciously and systematically cover up *deliberate* abuse, parents are unlikely to
 make sufficient changes to protect children from harm within an appropriate
 timeframe.

Chapter Five: Motivation and engagement

Introduction

The previous chapter focused on ambivalence, denial and resistance to change; it considered the research literature concerning those parents who, for various reasons, are unable to acknowledge the existence of harmful behaviour patterns or their damaging impact on children's wellbeing, and some of the approaches that may reduce barriers to change. This chapter looks at the more positive side of the picture, and explores how parents whose children have previously been likely to suffer significant harm can become motivated to change and engage with services designed to support them through the process. Practitioners who are responsible for making structured professional decisions concerning how far parents are motivated to change, and how they can be supported in translating motivation into action will require the same skills in relationships supported by knowledge of the issues, information from a wide range of other sources including data from standardised instruments, and the same supportive working environment as those who are assessing capacity to change in situations where parents deny that there is a problem.

Much of the extensive literature on capacity to change is drawn from studies that explore how adults overcome problems such as alcohol and substance misuse and domestic abuse, and how long-standing mental health problems can be controlled and alleviated. However, this evidence largely comes from studies of adults who are attempting to overcome a relatively discrete, single issue, while those whose children are on the edge of care are more likely to be facing complex combinations of such problems, together with a whole range of other exacerbating factors which further reduce their ability to parent. The combination of factors which place their children at risk of future harm may well prove overwhelming. As Chapter Three has indicated, theories of change that have been developed from different, less complex, populations may not be sufficiently nuanced to reflect the multi-layered changes that may be needed to overcome the multiple problems faced by such parents.

Motivation: the decisional balance

Research on human behaviour suggests that becoming motivated to change self-destructive behaviours is not a straightforward, rational process. There are factors that increase motivation, and counteracting factors encouraging individuals not to leave the familiar *status quo*. It appears that change occurs when the decisional balance reaches a tipping point, and the potential gains begin to outweigh the anticipated losses (see Prochaska *et al.*,1994). In a seminal paper, Janis and Mann argued that the anticipated gains and losses could be categorised into four different types of consequences: (a) utilitarian gains or losses for self, (b) utilitarian gains or losses for significant others, (c) approval or disapproval from significant others, and (d) self-approval or self-disapproval

(Janis and Mann, 1977, quoted in Prochaska *et al.*, 1994, p.40). Although commentators have since argued that these categories could be simplified into pros and cons (see Prochaska *et al.*, 1994), they form a useful construct for helping practitioners and parents to understand why behavioural change is so difficult, and identifying where the tipping points might be. There is some evidence to suggest that motivation to change involves an increased understanding of the benefits, while action is triggered by a decreased perception of the costs (Prochaska *et al.*,1994).

Tipping the decisional balance: becoming motivated to change

Ekendahl (2007) found that current circumstances, such as an individual's time perspectives and social context, had an impact on their willingness to make the changes necessary to overcome substance misuse. Experiencing the negative consequences of destructive behaviour patterns, such as loss of employment, health problems or separation from a partner and/or children, can all be factors that motivate people to seek change. There is evidence that people who have the most severe problems are most likely to seek help (see Ekendahl, 2007; Gossop *et al.*, 2006).

However, Freyer and colleagues (2005) have shown that there is a difference between readiness to seek help or treatment and readiness to change (see also Littell and Girvin, 2005). Engagement with services can be superficial and sometimes instrumental. These authors cite a number of studies that focus on motivation to overcome substance misuse to support their argument that treatment for addictive behaviours may be perceived as offering a brief respite from the immediate adverse consequences rather than a means of achieving long-term abstinence (see O'Toole *et al.*, 2006, 2008).

Parents facing multiple problems may also become motivated to change in one area, but may not necessarily appreciate the need for change in others. For instance, nearly 50% of adults with mental health problems have met the criteria for a substance use disorder at some stage in their lifetime (see DiClemente *et al.*, 2008, p.25). The coexistence of these two problems is associated with: 'non-adherence with medication, symptom exacerbation, rehospitalisation, poor social adjustment and worse prognosis'. These patients may accept hospitalisation in order to alleviate a particular symptom, and while there, take medication and accept treatment to overcome substance misuse; however, once discharged they may avoid all treatment (see DiClemente *et al.*, 2008, p.25).

A recent study of motivation for change through perpetrator programmes for fathers who are violent towards their partners (Stanley *et al.*, 2012a) distinguished between intrinsic (internal) and extrinsic (external) motivation. The study found that whilst both forms of motivation were valuable, intrinsic motivation was generally more closely associated with greater long-term behaviour change. Men with greater intrinsic motivation wanted to control their behaviour and change their lives; they were motivated by a desire to secure access to their children, avoid losing their children to the care system, and to free their family from the scrutiny of children's services. Strong intrinsic motivation appears to be associated with a shift in self-image such that, for instance, men who come to regard themselves primarily as fathers who want the best for their children may be better able to

appreciate the impact of domestic abuse and improve their parenting capability than those whose self-image remains focused on developing a macho persona that controls and demands respect from their partners and other family members (Stanley *et al.*, 2012a).

The motivation to change behaviour may also be related to an individual's *locus of control*, i.e. whether they feel that events in their lives are within their own control (*internal locus of control*) or whether they are in the hands of others (*external locus of control*). Fisher and colleagues (1998) found that an internal locus of control among sex offenders prior to treatment was an important predictor of treatment success. However, Bowen and Gilchrist (2004) found that when self-referred perpetrators of domestic abuse are compared with those referred by the courts, the role of locus of control is not straightforward. This study found that although self-referred men had higher internal locus of control than those mandated by the courts to attend treatment, self-referred men also saw others as having power over their actions. The authors suggest that this could be interpreted as 'illustrating the importance of their partner's involvement in their treatment' (p.289).

Turning points

Some parents will experience events or circumstances which create a turning point in their lives and motivate them to make the changes needed to overcome adverse behaviour patterns and improve their parenting. Social workers need to be able to recognise these potential turning points so that they can help parents make the most of this window of opportunity. Although the circumstances of every parent will be unique, the literature highlights certain commonalities in how parents experience turning points and the factors which trigger them.

Sudden realisation and gradual change

Parents can experience turning points following a sudden realisation or as the result of a gradual process of change. Where sudden realisation occurs, individuals have described experiencing 'key instances and pivotal moments' (Higginson and Mansell, 2008, p.319) or an 'Aha!' moment (Jung-Beeman *et al.*, 2004, p.500). Participants in a study by Carey and colleagues (2007) described this moment as 'the lights going on, putting a shilling in the meter, a load being lifted' (p.182). On the other hand, reaching a turning point through a process of gradual change has been described as 'realisation creeping up' (Higginson and Mansell, 2008). An example might be a parent referred to children's services who does not appreciate the risk of losing their child until well into the referral process (Ward, Brown and Westlake, 2012). It is possible for parents to experience turning points as both a sudden and gradual realisation (Carey *et al.*, 2007). This has been described as:

A period in which there was a slow, cumulative build-up of information and the beginnings of new tentative perspectives which was followed by a significant instance which stuck in the participant's memories as being of high importance (Higginson and Mansell, 2008, p.319).

Hitting rock bottom as a turning point

A common hypothesis is that some people need to hit rock bottom before they reach a turning point and make positive changes (Field *et al.*, 2007). The findings of Carey and colleagues (2007) support this hypothesis, with participants describing a moment of realisation occurring as they reached a particularly low point in their lives. However, other research has cast some doubt on this theory. Field and colleagues' (2007) study of motivation to change among substance users produced conflicting findings. Although increasingly negative life events made seeking treatment more likely among their sample, experiencing increased emotional distress did not.

The role of specific circumstances as a turning point

Certain specific life events or circumstances have been identified as turning points that could lead to improved parenting. The threatened loss of a partner can provide a turning point for domestic abuse perpetrators, for instance, leading them to enrol in treatment programmes (Hester *et al.*, 2006; Stanley *et al.*, 2012a). Stanley and colleagues (2012a) also found that the involvement of the courts or children's services led some violent fathers to enrol in voluntary treatment.

King and colleagues' (2009) longitudinal study of young people who were homeless or at risk of homelessness found that young women saw pregnancy as a turning point in their lives. These young women differentiated between their past and present selves, with pregnancy providing an opportunity to distance themselves from their street identity and assume the role of parent. Resultant changes included finding more secure housing, going out less, and staying away from drugs and street-involved friends. In some cases, pregnancy became the trigger to reconnect with family for support and it created a renewed interest in returning to education. Pregnancy has also been found to provide a turning point for drug or alcohol dependent women (Bessant, 2003; Etherington, 2007; Kreager et al., 2010). The birth of a child can also act as a turning point for fathers, and has been found to lead to a reduction in their levels of criminal behaviour and use of alcohol and tobacco (Kerr et al., 2011). However, the changes triggered by pregnancy and the birth of a child are potentially short term in nature. Women participating in King and colleagues' (2009) study described anxiety about the future and their ability to maintain change. They emphasised the importance of access to interventions supporting them in the longer term. Moreover, Bessant (2003) found that the ability of substance misusing mothers to maintain change in the long term was dependent upon their being able to access drug treatment, which in turn hinged on the availability of suitable housing. Turning points should therefore be viewed by social workers as opportunities to engage with parents and introduce appropriate ongoing support.

Collins and colleagues (2012) found that individuals with alcohol dependence sought treatment because it offered a brief respite from the negative consequences of their

addiction. Likewise, studies have found that concern over physical health and acute hospitalisations among homeless and non-homeless substance users provides a 'treatable moment' to engage with them (O'Toole *et al.*, 2006, p.144; O'Toole *et al.*, 2008, p.1329). However, as with research involving pregnant women discussed above, these studies found that the window of opportunity for engagement was limited. Motivation to change behaviour reduced as an individual's physical condition improved following medical treatment.

Cues for action: listening to children

Hahn and colleagues' (1996) study of parental engagement in health prevention activities explored the impact on parents of different cues to action and found hearing their children's views of their behaviour to be 'the most pervasive and effective cue' (p.168). Humphreys and colleagues (2011) and Stanley and colleagues' (2012a) research on domestic abuse reached a similar conclusion. Studies of domestic abuse have also found that a potent cue for action for male perpetrators is the realisation that, in the eyes of their children, they may become like their own abusive fathers (Sheehan *et al.*, 2012; Stanley *et al.*, 2012a).

However, Randolph and colleagues' (2009) review of parental engagement identified a dearth of literature on cues for action, generally suggesting that this is linked to the difficulties in measuring the influence of the varying cues. The relationship between cues for action and parental engagement could usefully be explored by further research.

Coercion

Prochaska and Prochaska (2002) assert that if parents become conscious of how they defend themselves when they feel threatened, for instance, by care or adoption proceedings, they will be more likely to progress to the next stage in the change process. However, given the internal dynamics, coercive action can have a kill or cure effect, pushing some parents who have previously been ambivalent about the *need* for change into taking action to overcome their problems, while those who have been ambivalent about their *capacity* to change become further entrenched in adverse behaviours that shield them from reality. The point is vividly illustrated by the extent to which parents who are permanently separated from their children tend to become deeply enmeshed in the behaviours that led to the initial intervention immediately after an adoption order has been made (see Forrester and Harwin, 2008; Neil, 2010; Ward, Brown and Westlake, 2012).

There is therefore some debate as to whether coercion strengthens motivation to change, or whether it is counterproductive. Bowen and Gilchrist (2004) examined how engagement varied between perpetrators of domestic abuse attending programmes in the UK on a voluntary or a mandatory basis. Using the Trans-Theoretical Model of Change (Prochaska and DiClemente, 1983), the authors found that self-referring domestic abuse offenders were more likely to be in the 'contemplation' stage and court-referred offenders in the 'pre-contemplation' or denial stage. The results of this study

suggest that self-referred offenders have already acknowledged their problem and believe that they are taking steps to change their behaviour, whereas court-referred offenders have yet to accept that a problem exists. The two groups of offenders may fundamentally differ in their levels of motivation to change; by ignoring these differences when referring them to offender programmes, we may be reducing the chances of successfully helping them to change their behaviour (Bowen and Gilchrist, 2004).

Voluntary participation in domestic abuse treatment programmes can therefore be a strong predictor of the likelihood that an individual will remain in treatment (Bowen and Gilchrist, 2004; Cadsky *et al.*,1996). However, other factors may also be important. For instance, men who have invested the most effort in the programme, such as those who have had the greatest distance to travel to participate, have been found to be more likely to continue in treatment (DeHart *et al.*, 1999). In addition, individuals with greater insight into their own problematic behaviour will be more motivated to change because they are less likely to blame external factors for their domestic abuse (Fisher, Beech and Browne, 1998). Zalmanowitz and colleagues (2013) make a distinction between internal and external motivational factors. A court requirement to participate in an intervention can be an external motivator; however, internal motivation will also need enhancing as part of the intervention to reduce the likelihood of recidivism. In the UK the courts do not have the power to impose such participation on parents in care proceedings, but participation in these programmes is sometimes a term of a written agreement made at the preproceedings stage.

Some studies have, however, produced conflicting evidence about whether court-referred or self-referred domestic abuse offenders have higher levels of attrition from treatment programmes. For instance, in an evaluation of a domestic abuse intervention programme in London, Burton and colleagues (1998) found that some self-referred offenders attend treatment in the short term as an attempt to prevent their partners leaving them. The authors suggest that if this initial display of intent to change behaviour is effective, or their partner decides to leave them anyway, there is insufficient reason to continue attending the programme, indicating that their motivation was largely superficial and instrumental, rather than rooted in a genuine desire to change. Bowen and Gilchrist (2004) also suggest that court-referred men have the added incentive of further legal action to maintain their attendance in the programme.

Gregoire and Burke (2004) explored the relationship between coercion and motivation to change alcohol consumption. The study found that involvement in the criminal justice system may be linked to an increase in motivation to change, even when controlling for problem severity. The study suggests that legal coercion may help to tip the decisional balance and move individuals from denying that a problem exists towards taking action to overcome it. The authors conclude:

Legal coercive pressure over the course of the treatment experience may have the most potential for influencing client behaviour toward outcomes since it is perhaps the most consistent type of coercion, less subject to a change than coercion by a family member or employer, and less subject to influence by client manipulation (Gregoire and Burke, 2004 p.35).

There may therefore be some merit in legal involvement in encouraging individuals to engage with interventions.

As we have already seen, in families where adult behaviour patterns impose a risk of harm to the child, the issues become more complex. A key principle set out by the Children Act 1989 is that local authorities should work in partnership with parents, whenever such an approach is consistent with the child's welfare (Department of Health, 1989). Programmes such as the Family Partnership Model (Harnett and Day, 2008; see also Chapter Four) are designed to facilitate this. However, the imbalance of power between parents and social work practitioners means that both parties are well aware that such partnerships are unequal and that, ultimately, the social worker has the authority to take coercive action by obtaining a court order for the child's removal. Where children are on the edge of care, the potential use of coercion becomes a particularly pertinent issue. Arguably there will always be a level of involuntary engagement in social work practice where children are suffering, or likely to suffer, significant harm. It is therefore important to understand which factors can inhibit or facilitate the likely success of interventions where participation is mandatory.

In evidence submitted for the Munro Review, the Family Rights Group asserted that parents feel scared to trust and work openly with social workers, and to reach agreement about how their children can be kept safe when they fear they may be removed by the local authority: 'the system does not support families to take responsibility; instead parents often feel decisions and actions are done *to* rather than *with* them, thus encouraging a sense of dependency and resentment' (Family Right Group, cited Department for Education, 2011, p.36). In some cases therefore, once court proceedings have begun, the opportunity to engage parents can be lost (Harwin and Ryan, 2008). On the other hand, the threat of removal can act as a turning point for some parents, tipping the decisional balance and providing a wake-up call that jolts them into action. The following quotation comes from one of a number of interviews with mothers who succeeded in overcoming substance misuse following a social work intervention:

And I think the scare what social services gave me was a kick up the arse and the scare that I needed... They were going to put [baby] into foster care... And I thought to myself, I just cannot, you know, you know what, I felt like a movie, I felt like, oh my god, my baby, not my baby. And he was so tiny, I felt like, oh no my baby, I felt like, and I thought you're having a laugh, I couldn't believe, you know... A big shock, a big shock, it was a big wake-up call and it was just a terrible feeling, I couldn't believe it (see Ward, Brown and Westlake, 2012, p.137).

Therapeutic jurisprudence

In both the USA and now England, child welfare interventions have been developed that use the element of coercion provided by a court to address problematic parental behaviour patterns. These interventions are based on the theory of therapeutic jurisprudence, described by one of its founders as 'the law as therapeutic agent' (Wexler, 1993, p.17). This approach uses the power of the courts as a means of solving problems rather than offering punishment, and may provide an effective means of engaging parents in changing their behaviours. It also requires judges and lawyers to be aware of the therapeutic and anti-therapeutic consequences their actions and decisions can have for offenders (Casev and Rottman, 2000). Harwin and Rvan (2008) describe its focus as being 'the health, welfare and rehabilitation of the offender, as well as their punishment' (p.283). The approach does not place the appendic considerations above all others, but incorporates them within the judgment process (Wexler, 1993). The use of therapeutic jurisprudence with parents to reduce risk to their children would therefore involve judges and magistrates assuming a more proactive and involved role (Harwin and Ryan, 2008). Wexler suggests that therapeutic jurisprudence provides an opportunity for the courts to help offenders deal with their 'cognitive distortions', recognise the problems with their current behaviours and start a change process (Wexler, 1993).

Family Drug and Alcohol Courts

In the USA principles of therapeutic jurisprudence underpin the development of specialist problem-solving courts which acknowledge that problematic parental behaviours require a range of interventions from other disciplines supported by the element of authority provided by the courts. Problem-solving courts have been developed widely across jurisdictions, but they have a number of key elements. These include a focus on case outcomes, non-traditional roles in the court room, multi-disciplinary collaboration and specially trained judges or magistrates who regularly monitor the progress of the case (Harwin and Ryan, 2008).

A national evaluation of this approach in family drug treatment courts in the USA produced positive findings (Worcel *et al.*, 2008). It has now been introduced in England through the Family Drug and Alcohol Court (FDAC), an approach to care proceedings in cases where parental substance misuse is a key element in the local authority decision to take legal action (Harwin *et al.*, 2011). FDAC was developed in response to concerns in relation to care cases involving parental substance misuse, including poor child and parent outcomes; insufficient co-ordination between adult and children's services; late intervention to protect children; delay in reaching decisions; and the increasing costs of proceedings, linked to the cost of expert evidence (Harwin *et al.*, 2011; see also Masson *et al.*, 2008, Ward, Brown and Westlake, 2012).

The distinguishing features of FDAC are that: the judge adjudicates the care proceedings and also holds responsibility for running a specialist treatment court; he (or she) plays a non-traditional role in order to motivate parents as well as to remind them of their responsibilities; a specialist multi-disciplinary team is attached to the court and

coordinates an intervention plan for parents which includes ongoing support and monitoring as well as assessment; parental progress is monitored and supported through regular fortnightly problem-solving therapeutic reviews during which the judge talks to parents and social workers directly; and parents are advised and supported by parent mentors who have themselves come through similar experiences (including, ideally FDAC) (Harwin *et al.*, 2011, 2013, 2014).

FDAC has been piloted by three local authorities in London. An evaluation of the pilot initially compared a sample of 55 families (with 77 children) who entered FDAC with 31 families (49 children) who were the subjects of usual care proceedings during the same period in three other London authorities (Harwin et al., 2011, 2013, 2014).9 Parental substance misuse was a key issue in all cases. From this original sample, it was possible to follow 41 FDAC and 19 comparison cases until the final court order. The results show that more FDAC (48%:19/41) than comparison parents (36%:7/19) had controlled their substance misuse by the end of proceedings and had been reunited with their children. FDAC parents were also engaged in more substance misuse programmes over a longer duration than the comparison parents. FDAC makes use of motivational interviewing techniques, and interviews with parents suggest that this may be one reason for positive findings (see Harwin et al., 2013). Data on the costs were analysed for a sub-sample of 22 FDAC cases and 19 comparison families. Although the samples were small, the findings were encouraging; for instance, the data suggest that local authorities saved £682 per FDAC family on court hearings and £1,215 per family on the provision of expert evidence to the court. Financial savings in FDAC cases were made because assessments performed 'in house' by the FDAC team were less expensive than those undertaken by independent experts in ordinary proceedings; children spent less time in out-of-home-placements; and there were fewer contested hearings (Harwin et al., 2011).

A second stage of the evaluation (Harwin *et al.*, 2014) extended the sample to 106 FDAC families (149 children) whose cases were listed to be heard by FDAC between January 2008 and December 2010, and 101 comparison families (151 children), with the aim of increasing the robustness of the findings and testing outcomes after completion of court proceedings. Once again, more FDAC parents were offered substance misuse services (95% (52/55) FDAC mothers and 58% (28/48) FDAC fathers) than comparison parents (55% (45/82) mothers and 27% (17/64) fathers, and more stopped misusing substances. Of those cases which could be followed until final court order, 40% (35/88) of FDAC mothers and 25% (13/52) of FDAC fathers ceased misusing compared with 25% (24/95) and 5% (2/38) in the comparison group. These differences were all statistically significant. These data also indicate that FDAC may be more successful in engaging and supporting fathers than ordinary proceedings, a point also made by some of the

⁹ This was a non-randomised comparative trial in six local authorities with a relatively high level of evidence (Level B, see Chapter Six).

children's guardians and social workers who participated in the study (Harwin *et al.*, 2011, p116).

A one year follow-up of 42 families where the courts decided that children should live with their parents found that there were fewer further episodes of abuse or neglect in FDAC than in comparison families (25%: 6/24 *versus* 56%:10/18). Further evaluation with a larger sample would provide more robust data, particularly as there appear to have been quite high attrition rates in some of the earlier samples studied. Nevertheless the English pilot of FDAC shows that, when combined with a package of timely, co-ordinated interventions from a range of professionals, the authority provided by the courts can play a constructive role in motivating parents to change.

Stress – Strain – Coping –Support Model (SSCS)

While therapeutic jurisprudence utilises the formal power of the courts to motivate parents to change, other models use the informal power of family and social networks. Historically, service providers have paid only limited attention to the impact of addictive behaviours on the family, ignoring the family's potential influence on the course of addiction problems and the severe stress and strain addiction can cause to family members (Copello *et al.*, 2009; Orford *et al.*, 2005). The Stress-Strain-Coping-Support model (SSCS) conceptualises addiction in a family 'as creating chronic stress for family members, giving the latter the task of finding ways of coping with that stress, and of seeking support, both professional and informal, in meeting that task' (Orford *et al.*, 2005, p.1613). The SSCS model rejects the idea that families or their members are dysfunctional or pathological and instead views them as ordinary people trying to cope with issues not of their own making (Copello *et al.*, 2009; Orford *et al.*, 2010).

Social Behaviour and Network Therapy

Social Behaviour and Network Therapy (SBNT) was originally developed to help treat alcohol addiction and is based upon the SSCS concept that involving family and friends of the person undergoing treatment to create a network of social support is essential in helping them overcome addictive behaviours (Copello *et al.*, 2006). The approach has three stages: identifying and approaching those family and friends who could form part of a support network; building, engaging and mobilising the social network over a number of training sessions by covering a range of topics including communication, coping and dealing with lapse and relapse; and preparing for the future, including planning how to deal with changing circumstances to provide the best chance of long-term success (Copello *et al.*, 2002).

A large randomised controlled trial compared the impact of SBNT in treating alcohol addiction with that of Motivational Enhancement Therapy (an approach based on Motivational Interviewing, see Chapter Four) with 742 participants across seven sites in the UK. It found that both groups reported substantial reductions in alcohol dependence and problems, and better mental health-related quality of life over the subsequent twelve months (UKATT, 2005a). Both therapies also saved about five times more in expenditure

on health, social, and criminal justice services than they cost (UKATT, 2005a, 2005b). However SBNT did not differ significantly in effectiveness and cost effectiveness from Motivational Enhancement Therapy (UKATT, 2005a, 2005b)¹⁰.

SBNT has also been adapted for use by therapists in the treatment of drug addiction. A small before and after study¹¹ involving 12 specially trained therapists delivered SBNT over an eight month period to 24 drug users in the West Midlands. Standardised measures used at baseline and at a three month follow-up showed a significant reduction in drug use and improvements to the participant's family environment (Copello *et al.*, 2006). Although the findings are promising, the small sample and lack of a comparison group means we do not know whether the intervention is better than service as usual. However, a randomised controlled trial is currently underway in the UK, looking at the use of a reduced programme (Brief-SBNT) with heroin users in opiate substitution treatment (Day *et al.*, 2013). This approach mirrors some (but not all) of the elements of Family Group Decision-Making (see Chapter Four).

Engagement

Whether or not coercion or pressure from courts and family members play a part, there is no guarantee that engagement with services will lead to change. A number of studies have found little correlation between engagement and outcome (see Gossop *et al.*, 2006; Littell and Girvin, 2005, 2006), and Platt (2012) points out the danger 'of assuming that good engagement is predictive of change' (p.139). Some adults are able to make significant changes to self-destructive lifestyles without support from services (see Sobell, Cunningham and Sobell, 1996; Ward, Brown and Westlake, 2012). Others may be strongly motivated to change, but lack the self-esteem, confidence and skills required to translate motivation into capacity. External factors will also act as barriers to change, and almost all parents will, in varying degrees, experience some ambivalence about both the necessity and the desirability of change.

The absence of a single definition has resulted in a lack of clarity about how parental engagement can be assessed (Altman, 2008). Broadly, engagement can be considered as compliance, involvement, attendance rates and participation (Gladstone *et al.*, 2012); a process rather than a one-off event (McCurdy and Daro, 2001), and continuing over prolonged periods (Randolph *et al.*, 2009). There is also a distinction between engagement as an outcome (i.e. service usage behaviours) and engagement as a causal influence (a state of mind, affect, or attitude) (Yatchmenoff, 2005). Social workers and parents both play a role in the engagement process, with both parties contributing to the success of an intervention (Altman, 2008; McCurdy and Daro, 2001). Taylor and

¹⁰ This was a large randomised controlled trial in seven sites with a high level of evidence (Level A, see Chapter Six).

¹¹ Evidence Level C, see Chapter Six

colleagues' (2008) study of parents who dropped out of an alcohol treatment programme found that the positive and negative factors relating to engagement meant that it frequently became a staged process, with parents going through a testing out period, and then dropping out, or missing appointments before re-engaging.

Platt (2012) proposes the following definition:

The mutual, purposeful, behavioural and interactional participation of parent(s) and/or carers in services and interventions provided by social work and other relevant agencies with the aim of achieving positive outcomes (Platt, 2012, p.142).

Models of engagement

A number of theoretical models of parental engagement have been developed for use in differing contexts. However models based on factors associated with parents' voluntary engagement in programmes (for example, McCurdy and Daro, 2001 and Randolph *et al.*, 2009) may not be entirely relevant for use in a child welfare context where a level of coercion often has a role.

An Integrated Model of Parental Engagement

Platt (2012) proposes an integrated model of parental engagement, specifically developed for use in a child welfare context in the UK. This comprehensive model incorporates conceptual advances from a number of fields including clinical psychology (Drieschner *et al.*, 2004); models of parental engagement in voluntary family support services (Littell and Tajima, 2000; McCurdy and Daro, 2001); and the Multifactor Offender Readiness Model (MORM) (Ward *et al.*, 2004).

Platt (2012) maps out the specific factors that are known to be associated with engagement. These include the family's particular circumstances; the parents' perceptions of their situation and the intervention; and factors related to provider and programme. His model demonstrates how these factors interact to produce varying degrees of engagement. The model has not yet been formally evaluated, and the relationship between the different factors and the outcomes of engagement are not yet fully understood. Nevertheless, it is designed to offer practitioners 'a framework for thinking and analysis, within which to understand and assess parental engagement, to attempt to anticipate problems of engagement, and to identify key issues to address where engagement is problematic' (Platt, 2012, p.146). Motivation to change and engagement with services are complex issues. A model such as this cannot predict change; for instance, parents will exercise their own free will and may perceive different factors related to engagement in unexpected ways. Nevertheless such a model should provide social workers with a conceptual map that facilitates a greater understanding of the issues to be addressed, as well as a guide for adopting an analytical approach to the evidence they are required to present in court proceedings (see Ministry of Justice, 2013). This is discussed further in Chapter Eight.

Figure 5.1 provides an illustration of Platt's model of engagement. Background factors include social issues such as poverty, unemployment and housing; individual psychology or psychopathology; the nature of the problem; and the overall efficacy of the chosen treatment. These may play a role in parental engagement, although research evidence in relation to their significance has been varied (Platt, 2012).

The two key types of driver in Platt's model are *internal*, parent-related determinants, and *external*, service provision determinants. *Internal* determinants are personal, psychological or behavioural factors. These include whether a parent has negative feelings towards services; how their past experiences may influence their attitude; how they perceive their own ability to address their problems; and their fears and expectations. These factors may also be compounded by depression, anxiety, mistrust, confusion and self-blame, as well as an inability to seek help, and poor communication skills, inadequate literacy skills and a lack of confidence (Platt, 2012).

Internal determinants of engagement clearly overlap with internal factors that contribute to parents' understanding and acceptance of the need for change, discussed in Chapter Four. They also encompass the parent's perception of those factors that contribute to the decisional balance between potential gains and anticipated losses that underlies motivation to change. They operate in combination with external factors to produce different levels of engagement.

In Platt's model *external* determinants of engagement include circumstantial factors, such as the nature of the behaviour towards the child and society's response to it; whether the intervention is voluntary or mandatory; the availability of resources; the availability of social and professional support; community cohesion and cultural factors; the efficacy of the programme, such as matching parent and programme goals, perceived appropriateness, and tailoring of services to meet parents' needs; and the knowledge, skills and values of the practitioner including their reliability, honesty, listening, role clarification, empathy, even-handedness and collaborative problem-solving abilities (Platt, 2012).

One of the strengths of Platt's model is the emphasis it places on the organisational and professional factors that promote or inhibit engagement. Inadequate resources (Farmer and Lutman, 2012), frequent staff turnover (Ward, Brown and Westlake, 2012), and levels of staff experience (Gladstone *et al.*, 2012) have all been shown to contribute to parents' willingness (or unwillingness) to engage with services.

Figure 5.1: An Integrated Model of Parental Engagement with Child Welfare Services

(Developed from Ward *et al.*, 2004 and reproduced from Platt, 2012, p.141, with permission)

Background conditions

solving, etc.

Social (poverty, unemployment, housing, etc.)

Individual psychology and psychopathology

The nature of the problem Internal / individual determinants Cognitive Affective Behavioural Identity **Engagement with services** Volition / motivation Behavioural / attitudinal: Outcomes Attendance / keeping Parent: appointments, etc. Openness concerning Changes in attitude problems Changes in behaviour Using worker's contribution Child: Task completion and commitment **External determinants** Safety and health Making sacrifices Emotional / behavioural Taking the initiative Circumstances / developmental Resources Education Interactional (working alliance): Support Programme Goals Worker / therapist Tasks skills, e.g. honesty Bonds and role clarification, listening, empathy, even-handedness, collaborative problem

(Solid arrows indicate primary direction of effects. Broken arrows indicate feedback loops).

Platt (2012) identifies two observable features of engagement with services affected by the determinants discussed above. These are *behavioural indicators* (i.e. keeping appointments, parent's openness concerning their problems and whether they complete tasks) and *interactional indicators* or 'the working alliance' (mutual agreement on goals and tasks and the bond between social worker and parent). An assessment of overall engagement needs to explore systematically the various determinants identified as being relevant to a particular case and consider how they impact on the observable behavioural and interactional indicators. It should help practitioners reach a judgment concerning the extent to which parents are ready and willing to take the actions necessary to ensure that children who are on the edge of care could be better safeguarded in the future.

Conclusion

Previous chapters have explored the interlocking factors that undermine parenting capability and increase the likelihood of abuse and neglect. These include factors which may relate to the parent themselves, their environment, or their children. They have also considered how these factors promote or inhibit change. This and the preceding chapter have considered how constellations of these factors facilitate or discourage the engagement with services that may be necessary for change to occur. Compliance is not the same as engagement; assessing parents' motivation to change and the strength of their engagement with services are by no means simple tasks, as the complexity of the models described in this and the preceding chapter show. Potent factors include the nature and delivery of services themselves and the parent's own perceptions and actions. The next chapter carries the argument further by presenting some of the research on what is known about which types of services are effective.

Key points from Chapter Five

- Change is a complex process, particularly for parents facing multi-layered combinations of problems and whose children are on the edge of care.
- There is a difference between readiness to seek help or treatment and readiness to change, although one may lead to the other.
- Some parents will engage superficially with services in order to meet short-term objectives without the intention of making lasting changes.
- Becoming motivated to change self-destructive behaviours is not a straightforward matter. Change occurs when the decisional balance reaches a tipping point and the potential gains are perceived as outweighing the anticipated losses. Clarifying the perceived advantages and disadvantages of change may help practitioners and parents understand why change is so difficult and how it might be facilitated.
- Parents facing multiple problems may become motivated to change in one area, but may not necessarily appreciate the need for change in others.
- Some parents will experience events or circumstances which create a turning point in their lives and motivate them to make the changes needed to overcome adverse behaviour patterns and improve their parenting. Turning points provide opportunities to promote and support change.
- There will always be a level of coercion in statutory interventions where children
 are suffering, or likely to suffer, significant harm. Coercion can be the cue to action
 that is required to help parents realise that change needs to happen, but it can
 also be counterproductive and push parents who are uncertain about their
 capacity to change towards becoming further entrenched in adverse behaviours
 that shield them from reality.
- When combined with a package of timely, co-ordinated interventions from a range of professionals, the authority provided by the courts and informal pressure from

friends and family members can play a constructive role in motivating parents to change.

- There is no guarantee that engagement with services will lead to change.
- Parental engagement is influenced by a number of interlocking factors, including internal and external determinants and background factors.
- A theoretical model of the different factors that promote or inhibit parental engagement with child welfare services can facilitate a greater understanding of issues that need to be addressed and serve as a framework for assessing how far parents are ready and willing to change.

Chapter Six: Building Parenting Capability through Evidence-based Interventions

Introduction

The previous chapter considered how parents whose children may be suffering or likely to suffer significant harm can become motivated to address those difficulties which are undermining their parenting capability and engage with services. This chapter explores how different types of interventions can support parents in overcoming these difficulties and how their impact can be interpreted and understood. Although some parents are able to change without formal support, most will need help in overcoming abusive or neglectful behaviour patterns, improving their parenting capability and in addressing the underlying risk factors that reduced their capacity to provide a nurturing home. Children may also need support in overcoming the impact of maltreatment. Both parents and children may need support in repairing the dysfunctional family relationships which both engender, and are exacerbated by, abuse and neglect.

Local areas are required to have a range of effective, evidence-based services available to support children and families in need, including those where there are significant child protection issues (HM Government, 2013). These may range from universal (primary) and targeted (secondary) services to prevent maltreatment before it has taken place. through to specialist (tertiary) services designed to prevent further impairment, or a recurrence of maltreatment in families where abuse and neglect have already been identified and children are on the edge of care. There is a very wide spectrum of services available, delivered by a range of practitioners who have safeguarding responsibilities across numerous agencies, although the same services are not available from every local authority or NHS Trust. This chapter identifies some of the issues that need to be taken into account in considering whether a service is likely to be or has been effective, and sets out some examples of interventions that are known to have a positive impact and are increasingly available in this country. However, it is outside the scope of this report to offer a comprehensive overview of the multiplicity of interventions in this area. Fuller information about a wider range of specific interventions can be found in the numerous systematic reviews that have been undertaken (see for example Barlow and Schrader McMillan, 2010; Barlow, Simpkiss and Stewart-Brown, 2006; MacMillan et al., 2009; Montgomery et al., 2009).

Current guidance states that services that focus on parents 'should always be evaluated to demonstrate the impact they are having on the outcomes for the child' (HM Government, 2013, p.13). Not only do social workers need to be aware of the types of support available, they also need to know whether they are appropriate to the needs of the family and what each might be expected to achieve. This is another reason why case conceptualization and goal setting are important (see Chapter Three). Without a theoretical understanding of the interplay of factors within the family that lie behind

maltreatment, it is not possible to set realistic targets or to identify the types of support that might help parents achieve and sustain change. The most effective services may not be available, or may not be acceptable to the family, and this chapter also considers some of the organisational barriers to successful service delivery. As Chapter Seven points out, change can be a lengthy process; many parents of children on the edge of care will need long-term support of varying intensity. This support will need to be offered well before the question of legal proceedings arises.

Understanding effective services and interventions

There are both ethical and economic reasons why it is important to understand the likely impact of services that are designed to ensure that children are adequately safeguarded. Ineffective services may not only fail to ensure that children are safe, they may also fail to offer parents adequate support in overcoming destructive behaviour patterns and demonstrating capacity to change. Moreover some interventions are worse than ineffective in that they have been shown to be harmful (see Jarrett, 2008; Lilienfeld, 2007; Rhule, 2005). A number of services have been developed in other countries, but translating them into a UK context can be a complex exercise and so only interventions that have been shown to have a positive impact in this country should be regarded as likely to be effective. An evaluation should show whether one of the services studied has significantly greater impact than another (or than no service at all) and how long the impact is likely to be sustained. It is unlikely that any service will benefit all participants, so it is important to know for what proportion of people a desired outcome has been achieved in the past, and whether people with some types of need are more likely to benefit than others.

General principles concerning evidence of effectiveness

Practitioners need to be able to assess whether evaluations of effectiveness have been sufficiently rigorous to produce reliable findings about impact and how these can be interpreted. Early evaluations of a new service are often undertaken by the developers, and so there may be a conflict of interests and a possibility of bias: rigorous reviews of evidence such as those undertaken by NICE take this into account. Services which have been evaluated with a very small sample, or which have had a high drop-out rate, so that a large proportion of participants failed to complete the programme, have not shown sufficient evidence of impact. Moreover, not all of the evidence is of equal quality: the weight given to the findings of an evaluation will be determined both by the rigour with which the research exercise was conducted and by the methodology employed. There is an established hierarchy of methodologies, and each level is considered to produce a different standard of evidence, as shown in Table 6.1.

The extreme complexity of most child welfare cases can make it difficult to be certain that like is being compared with like, and this needs to be considered in assessing the

evidence from comparative evaluations. Nevertheless, rigorously conducted randomized controlled trials (Evidence Level A), where service users with similar needs are randomly allocated to the intervention being evaluated or to a control group that receives no service or service as usual, are considered to produce the strongest evidence of effectiveness. However, they may be inappropriate in many areas of children's social care, where to place children in a control group that receives no service is not an option. There are also, quite rightly, ethical constraints on randomly allocating children to highly intrusive interventions such as adoption which will have long-term consequences for both them and their families. Unless we genuinely do not know whether an intervention is likely to be beneficial, not only for one child but for all children participating in a trial, random allocation cannot be justified. Comparing the impact of two types of existing service (Evidence Level B) is often more ethically appropriate, but the results will only tell us whether one service is likely to be more effective than the other, and not whether either is more effective than doing nothing. Before and after studies (Evidence Level C) may measure changes over time, but they cannot tell us how far (if at all) such changes are attributable to the service received, or to other factors within the service user's life that have contributed to change. Retrospective quantitative studies (Evidence Level D) and small case studies (Evidence Level E) may identify correlations between services and outcomes if appropriate data have been collected, but they are not evaluations in themselves; their findings indicate the questions that need to be asked in prospective studies of impact (for a fuller discussion see Davies and Ward, 2012, pp.96-99).

Findings from quantitative studies are generally considered to be statistically significant if they could have occurred by chance less than five times in a hundred (p=<0.05). Results are also often (but not always) given in terms of effect sizes: these represent the known impact of an intervention, using a simple statistical formula. In the statistical tests used in the studies in this report, an effect size of 0.2 is considered to be small, one of 0.5 is medium and one of 0.8 or greater is large (see Cohen, 1969, 1992). No service is 100% effective – the question to ask is whether it is more effective than an alternative.

Many intensive interventions, such as programmes designed to address parents' problems or multi-faceted programmes that address the needs of both parents and children, are standardised, and those who deliver them are expected to follow a particular procedure, often specified in a manual. The different elements of these interventions have been tested out with different groups of people and the manual reflects what has been shown to be effective. Altering the procedure may mean that the intervention is no longer so effective and the revised version will need to be re-evaluated. Implementation of a new intervention is a complex business and takes on average two to four years to complete (Fixsen *et al.*, 2005; see also Wiggins, Austerberry and Ward, 2012).

Table 6.1: Hierarchy of levels of evidence

Hierarchy of levels of evidence		
Design	Advantages	Disadvantages
Level A		
Randomised Controlled Trials	No systematic differences between conditions therefore any changes are due to treatment effects	Can be impractical or unethical to implement
Level B		
Two-Group Non- Randomised Comparative Trials	Groups can be matched to minimise known differences Practical for pre-existing groups	Groups may differ on factors for which the groups were not matched, potentially confounding the results
Level C		
Single-Group Pre-Post Studies	Measures change over time Often the only practical option	Impossible to know whether changes are due to the intervention or other factors
Level D		
Retrospective Quantitative Studies	Data may already be available and may provide some useful indications for more rigorous evaluation at a later date	Data may not have been collected specifically to evaluate this intervention and may therefore be incomplete or inadequate
Level E		
Case Studies	Data may provide useful indications for more rigorous evaluation at a later stage. Qualitative data may indicate potential areas for further explorations and analysis – suggesting why rather than what is happening	Data from a small number of examples may not be generalisable

^{*}Compiled from Montgomery, P., Gardner, F., Bjornstad, G. and Ramchandani, P. (2009) Systematic Reviews of Interventions following Physical Abuse: Helping Practitioners and Expert Witnesses Improve the Outcomes of Child Abuse. Report to Department for Children, Schools and Families: University of Oxford (Department for Education Research Brief DCSF-RBX-09-08A) pp.12-15, and reprinted from Davies, C. and Ward, H. (2012) p.96.

Understanding effective social work

The general case management and family support provided by social workers in England has not been formally evaluated (see MacMillan *et al.*, 2009). Where evidence of effectiveness exists it is often relatively weak (Levels C, D or E), and much of it has to be pieced together from studies where this issue was peripheral to the main focus of the research.

Nevertheless, while experimental trials of social work practice in England are in their infancy, research in this area increasingly incorporates a comparative element, making it possible to explore change over time or whether one group of children has had more successful outcomes than another (see Davies and Ward, 2012, p.27). Studies which make use of survey and case file data and interviews with professionals indicate that social work interventions for maltreated children are effective if they are characterised by thorough assessments leading to the identification of clearly specified goals and targets concerning what needs to be changed; provision of social work and specialist services to support such changes; careful planning that includes children and families; and strong proactive case management (Farmer and Lutman, 2012; Wade *et al.*, 2011).

Studies of users' views about social work case management rarely measure impact by exploring changes over time. Single, unrepeated interviews provide a weak source of evidence (Level E); they are sometimes undertaken with a convenience sample rather than with one that has been carefully selected and the findings may not be generalizable (David and Sutton, 2011). They are also often skewed by the untoward effect of rigorous gatekeeping and consent procedures that can result in a poor and uneven response rate (see Heptinstall, 2000; Ward, Brown and Westlake, 2012). Repetition of data collection over time, response rates and sampling techniques should all be taken into account when exploring what users say about the effectiveness of a service. Nevertheless users' views provide an important source of information that cannot be accessed in any other way, and the requirement to take them into account is enshrined in both legislation and policy (Department of Health et al., 2000; HM Government, 2013). Qualitative data from interviews with parents indicate that they value child and family social workers who are 'not afraid to break bad news' and are straightforward and honest about what needs to change and the likely consequences of failure to do so (Spratt and Callan, 2004; Ward, Brown and Westlake, 2012); who show sensitivity and are prepared to listen to their point of view and understand their circumstances (Forrester et al., 2008b; Morgan, 2008; Ward, Brown and Westlake, 2012); who use their power to support rather than to penalise them (Dumbrill, 2006); and who offer practical support and advocacy (Dumbrill, 2006). Thus both sources of information corroborate one another and point to the important part the relationship between the parent and the social worker can play not only in assessing capacity to change, but also in supporting the change process.

Evidence about the characteristics of *ineffective* relationships between social workers and parents, taken from studies of serious case reviews, can shed further light on how

social workers can support parental change. These show that the relationship can become dysfunctional if the social worker's decisions are influenced by factors such as unrealistic expectations of capacity to change, parents' overt hostility, or by theories of cultural relativism which are sometimes used to condone abuse or neglect (see Brandon et al., 2009, 2010, 2011; Ward, Brown and Westlake, 2012). All of these turn the focus away from the central purpose of safeguarding the child. Laming (2003) argued that social workers should adopt a position of 'respectful uncertainty', keeping an open mind about information presented to them by parents for whom much is at stake. Such a position needs to be supported by supervision that encourages the development of reflective practice (Dalzell and Sawyer, 2007), and should help social workers maintain sufficient distance to remain clear sighted about their role in supporting parents to change and the extent of progress that has been made.

While there is evidence concerning the components of effective social work practice, a number of studies have identified areas where practice is weak or ineffective. For instance, Farmer and Lutman (2012) found that care planning had been inadequate for over a third (36%) of a sample of 138 neglected children; Ward, Brown and Westlake (2012) found that about one in three infants identified before their first birthdays as likely to suffer significant harm, were still experiencing abuse and neglect by the time they were three. Both Farmer and Lutman (2012) and Wade and colleagues (2011) found that plans for maltreated children who return home from care are often unrealistic, with children frequently returning to parents who have been unable to overcome the behaviour patterns that precipitated the original removal. Farmer and Lutman (2012) found that in half (51%) of the families in their study 'a clear focus on the key issues in the case had not been consistently maintained by children's services' (p.68).

Barriers to effective social work

There are a number of reasons why social work practice in supporting parents to change is not always as effective as it might be. First, some studies have identified deficiencies in training, which may give too little weight to the acquisition of up-to-date knowledge in areas such as the impact of abuse and neglect on childhood development and risk and protective factors in families where children are likely to suffer significant harm (see Brandon *et al.*, 2009, 2010, 2011; Daniel *et al.*, 2011; Ward, Brown and Westlake, 2012). Child development is still not covered thoroughly in all social work qualifying courses, many of which fit it within a broader curriculum of human growth and behaviour or lifespan development (see Brandon *et al.*, 2011, p.20; Department for Educaton, 2011, p.75). Training may also fail to explore sufficiently the inherent conflicts in the social worker's role, such as how statutory responsibilities to define and act on child protection concerns can be reconciled with values that emphasise the importance of empowering parents (see Healy and Darlington, 2009).

Secondly, there are wide variations between authorities (see Farmer and Lutman. 2012; Sinclair et al., 2007; Wade et al., 2011). These are likely to relate to the prevalence of a number of organisational factors that have been identified as barriers to effective social work practice. Differences in organisational cultures can be reflected in the nature of supervision, which in some authorities still focuses more on performance management than on the development of reflective practice (Carpenter et al., 2012). Changes of social worker have also often been noted as detrimental to successful case management because they obstruct the development of constructive, supportive relationships with parents and children, and the implementation of plans (see Skuse and Ward, 2003; Ward, Holmes and Soper, 2008). Such changes reflect the widespread use of agency staff, the high turnover of more permanent staff, and the organisation of services which often require cases to be transferred from one team to another as families move through the system. Furthermore, pressure to close cases, frequently due to restricted resources, means that social work interventions are often relatively short term. The national statistics indicate that 29% of children remain the subject of protection plans for three months or less, only 19% for more than a year, and as few as 3% for two years or more (Department for Education, 2014b). Moreover, both social work and more specialist interventions tend to end abruptly with often inadequate arrangements for long-term, less intensive support or monitoring of children's circumstances (Ward, Brown and Westlake, 2012). Given the entrenched and complex difficulties facing parents whose children are likely to suffer significant harm and what is known of timescales for recovery (see Chapter Seven), there appears to be a mismatch between needs and services.

Intensive Family Preservation Services (IFPS)

While our search did not find any formal evaluations of general social work case management and family support, it did identify studies that explored the impact of Intensive Family Preservation Services (alternatively known as family preservation services or intensive family support services), often delivered by social workers. IFPS are intensive programmes, introduced to reduce the need for placing children in care by addressing crises, improving family functioning and promoting the use of social support systems (Al et al., 2012). Many programmes are based on the American 'Homebuilders' model (see: http://www.institutefamily.org/programs IFPS.asp). IFPS incorporate a number of common characteristics; for instance, families are usually considered to be 'in crisis' with the crisis generally linked to the possibility of a child entering care; services take place in the family's home; services are flexible and available to families 24 hours a day, seven days a week; they are concentrated in a period targeted at four weeks; and workers have very small caseloads (see Forrester et al., 2008b, p.412). Two extensive meta-analyses of IFPS programmes (Dagenais and colleagues' (2004) study of 27 programmes involving 10,296 children or families and Al and colleagues' (2012) metaanalysis of 20 programmes involving 31,369 participants) have found that IFPS have little impact on preventing children from being placed away from home, although there is

some evidence of improvement in family functioning (Al and colleagues (2012) found a medium positive effect of 0.486 on this variable).

IFPS have also been implemented and evaluated in the UK. For example, Option 2 is a Welsh intensive family preservation service aimed at reducing the need for children to enter care from families experiencing parental substance misuse (Forrester et al., 2008b). The Option 2 Service is based on the American Homebuilders model and uses a combination of motivational interviewing, solution focused counselling styles and other therapeutic and practical interventions. The service is delivered over four to six weeks and staff members are available to families 24 hours a day. The evaluation of Option 2 (Forrester et al., 2008b) compared the outcomes for 278 children receiving the service with those for 89 children who were referred, but could not be offered it immediately because there was no capacity¹². The average follow-up period was 3.5 years. As with the US studies, the evaluation found that participation in the Option 2 service did not reduce the likelihood of entering care (40% of the children in Option 2 entered care versus 41% in the comparison group). However, children in the Option 2 group entered care on average 117 days later than those in the comparison group; they spent less time in care (766 days vs. 958 days) and were more likely to return home (68% of Option 2 children were at home at follow up compared with 56% of the comparison group). The reduction of time spent in care meant that there were considerable cost savings (over £1000 per child in one of the participating local authorities). All families interviewed valued the service, the key components they regarded as helpful being: 'a nonjudgmental and understanding approach, good open communication between the worker and family, availability, reliability and high frequency of contact, suggesting helpful strategies and offering practical support if needed, support with substance misuse when required and help with family relationships when required' (Forrester et al., 2008b, pp 417-8). However, although Option 2 appeared to have a lasting impact with those families whose problems were less severe and entrenched, change was not maintained in the longer term with the families which were facing more complex and severe problems. Moreover, the finding that the impact of Option 2 was to delay entry rather than reduce the likelihood of admission to care raised questions concerning whether children benefited from the service if it prolonged their experience of abuse or neglect (see Forrester et al., 2008b, p.423). In both the American (Al et al., 2012) and the Welsh (Forrester et al., 2008b) evaluations there is some indication that the brief intensive crisis intervention that is characteristic of IFPS may not be of a long enough duration to help families in which there are concerns about child abuse and neglect to address complex and entrenched problems sufficiently to prevent their children from coming into care.

¹² This was a non-randomised comparative trial with a relatively high level of evidence (Level B).

Outcomes of care

Even with long-standing support from social workers and other agencies, some parents are unable to protect their children from harm. In cases where there is clear evidence of significant harm or its likelihood, multiple risk factors that are known to be associated with future harm, no mitigating protective factors and no active engagement or evidence of parental change, there is a strong possibility that children's life chances will be seriously compromised unless they are placed away from home (see Brown, Hyde-Dryden, Thomas and Ward, forthcoming; Ward, Brown and Maskell-Graham, 2012; Ward, Brown and Westlake, 2012; Wade *et al.*, 2011).

There are long-standing and well-documented concerns about the poor outcomes of separating children from their birth families and placing them in the care of local authorities (see for instance House of Commons Children, Schools and Families Committee, 2009; Sergeant, 2006) and there are some difficult issues that need to be confronted. These include instability of placements, low aspirations and insufficient support for young people making the transition to independence. Such evidence will be of concern to social workers who are faced with parents who are unable or unwilling to change. It is easy to argue that care can be damaging for children, and this has been the assumption behind much public debate over the last few years. It is difficult to ascertain the validity of such assertions because this is an area where randomised controlled trials are ethically problematic and few comparative studies have been undertaken. However, a number of increasingly sophisticated research studies indicate that the majority of children who become looked after benefit from care. International studies need to be approached with caution in this area because differences in definitions, perceived objectives, thresholds for entry, and the quality of provision may all contribute to misleading comparisons (see Ward, 2009). Nevertheless, research from France (Dumaret and Coppel-Batsch, 1998), Australia (Barber and Delfrabbro, 2005). Norway (Moe and Slinning, 2001) and some, though not all, studies from the USA show that placing children in care can have a positive impact on their welfare (Horwitz et al., 2001; Taussig et al., 2001; but see Lawrence, Carlson and Egeland, 2006 and Lloyd and Barth, 2011). Forrester and colleagues (2009) undertook a review of British research undertaken since 1991 that included data on changes in welfare over time for children in care. Twelve studies met their rigorous criteria. They concluded that:

...there was little evidence of the care system having a negative impact on children's welfare. Indeed, in almost all of the studies children's welfare improved, while there was none in which it deteriorated (p.450).

Maltreated children tend to do better in care than those who remain with abusive or neglectful families or return to them (see Wade *et al.*, 2011).

Outcomes of kinship care

Where children cannot remain with their birth families, both legislation and policy indicate that local authorities should make arrangements for them to live with 'a relative, friend or other person connected with [them] who is also a local authority foster parent, unless that would not be reasonably practicable or consistent with [their] welfare' (Children Act 1989, s.22.6 (a)). However, although there are obvious ethical as well as legal reasons why children should be placed within their extended families, research on kinship care has produced mixed findings.

Many of these findings are positive. Children in kinship care have been found to be less likely to be maltreated and to have greater placement stability than those placed with strangers (Newton, Litrownik and Landsverk, 2000). Kin carers are also more likely to go the extra mile in trying to meet the child's emotional needs, and to promote stronger ties with the extended family (Hunt, Waterhouse and Lutman, 2008). A systematic review by Winokur and colleagues (2009) concludes that children in kinship care 'experience better outcomes in regard to behaviour problems, adaptive behaviours, psychiatric disorders, wellbeing, placement stability, and guardianship than do children in foster care' (p.15). However MacMillan and colleagues' (2009) review of the international literature also found that some studies showed negative outcomes in terms of more delinquent behaviour and slower cognitive development among children in kinship care.

There is some evidence of lower standards of approval for kinship carers, so that children are sometimes placed with carers who are unable to meet their needs, and/or who do not know them and have extremely tenuous links with their birth family (see Sinclair *et al.*, 2007; Ward, Brown and Westlake, 2012; Ward, Munro and Dearden, 2006). Hunt and colleagues (2008) found that kinship placements tended to be more positive for younger children than for those who were older; they also identified concerns about the extent to which kinship carers met the basic needs of one in four children placed with them. Similar concerns have been raised by other studies (see Peters, 2005; Sinclair *et al.*, 2007). More rigorous assessment and higher thresholds for approval should address this issue.

There is also evidence that kinship carers may be particularly vulnerable. MacMillan and colleagues (2009) found that, on average, kinship carers were 'older, less well educated, less likely to be married, report more problematic parenting attitudes, and receive fewer non-child welfare services than unrelated carers' (p.260). If birth parents or other relatives do not accept the placement decision, kinship carers may also be dealing with difficult family situations (see Farmer, 2010). The research evidence indicates the need for extensive post-placement support; yet this is not always forthcoming. Kinship carers receive less caseworker support than unrelated carers (MacMillan *et al.*, 2009). Despite the legal requirement for parity of financial support with unrelated carers (*R v. Manchester City Council*, 2001), inadequate remuneration remains an issue. There has, for instance, been an increase in numbers of informal kinship arrangements, where financial support is discretionary and social work support is not always readily available (Selwyn *et al.*, 2013).

Outcomes of adoption

Studies of the long-term outcomes of infant adoptions are necessarily based on children who were placed several decades ago at a time when there were still powerful economic and social pressures on unmarried women to relinquish their children. Adoptions of these children, who are now in late middle age, show favourable psychosocial outcomes and low disruption rates (see Selwyn *et al.*, 2006 for further details). It is unlikely that many of these children will have experienced abuse before placement.

However, the majority of children placed for adoption currently in the UK have experienced maltreatment prior to entry to care or accommodation, and many will have experienced lengthy delays, insecurity and instability before permanence decisions are made and adoptive placements found. On average, children are looked after for two years and seven months before the adoption is finalised (Department for Education, 2013b), though they reach their permanent placement within a shorter period (Ward *et al.*, 2006).

Given their experiences it is not surprising that both looked after children and those who are adopted, experience emotional and behavioural difficulties, depression and confusion over identity (Neil, 2000; Smith and Brodzinsky, 2002). Biehal and colleagues (2010) found no significant difference in average scores on a standardised measure to identify clinically significant emotional and behavioural difficulties (the Strengths and Difficulties Questionnaire) between adopted children and those in long-term foster care, but both groups had higher scores than the general population. A study conducted in the USA (Keyes *et al.*, 2008) also found that the prevalence of behaviour problems was (marginally) higher amongst adopted children than in the general population. These comparisons do not include children who have been subject to child protection proceedings but returned home.

About 3% of adoptions from care disrupt within twelve years; substantially fewer than placements with foster carers, special guardians or carers with residence orders (Biehal *et al.*, 2010; Selwyn, Wijedasa and Meakings, 2014). However, research using the self-reported feelings of adopted children suggests that statistics on placement breakdown hide an underlying unhappiness for some children in placements that do persist (Thoburn, 2002), and about one in four adoptive parents describe major challenges and inadequate support in caring for a child with multiple and overlapping difficulties (Selwyn, Wijedasa and Meakings, 2014). Disruptions are closely associated with emotional and behavioural difficulties, and especially 'aggressive, acting out behaviours including cruelty to others, getting into fights, threatening others, overactivity, restlessness, hanging out with bad friends and overt sexualised behaviour' (Selwyn *et al.*, 2006). There are greater risks in adoption for sibling groups and children with additional needs (Rushton, 2003). Nevertheless, the majority of adoptions last until adulthood. Moreover, Howe's (1998) review of outcome studies found that, on

a measure that combined disruption rates, developmental rates and adopter satisfaction rates, 50-60% of adoptions of older children were successful.

There are strong indications that, the younger the child is when placed in local authority care or accommodation, or with an adoptive family, the better the chances of both a stable placement and successful psychosocial outcomes (see van den Dries *et al.*, 2009; Ward, Holmes and Soper, 2008). The older children are at placement, the more likely they are to display behavioural problems, including problems with peer relationships, attachment, conduct disorder and poor concentration (Biehal *et al.*, 2010; Haugaard, Wojslawowicz and Palmer, 1999), and therefore the greater the risk of disruption. Both adoption and local authority care can provide a nurturing environment from which most children will benefit, but they cannot always overcome the consequences of extensive experience of abuse and neglect.

Returning home from care

Only a very small proportion of children who become looked after are placed for adoption. Just over one in three children (35%) who leave care or accommodation each year are reunited with birth parents or other relatives (Department for Education, 2013a). Reunification is by far the most common reason for leaving care, and the proportion of maltreated children who return home is almost exactly the same as that for those who become looked after for other reasons (Wade *et al.*, 2011). However these children frequently return home to parents who have not been able to overcome the problems that led to the separation. About a third of maltreated children who return home re-enter care or accommodation within six months and two thirds within four years, including 81% of those who return home to substance misusing parents (Wade *et al.*, 2011). Children who experience repeated attempts at reunification to parents who have shown insufficient capacity to change experience the least satisfactory outcomes (Farmer and Lutman, 2012; Wade *et al.*, 2011).

Specific interventions to support parents and children

It is evident that social work support needs to be complemented by other, more specific interventions, delivered by a range of professionals if parents are to be given the best opportunities to overcome adversities and meet their children's needs within an appropriate timeframe. An increasingly wide range of effective specific interventions are now available; there is, however, some evidence that the majority of parents whose children are on the edge of care 'do not currently receive any formal intervention to improve their parenting skills' (Barlow *et al.*, 2008, p.3), and that many have inadequate access to the type of services designed to address specific problems such as alcohol and drugs misuse, domestic abuse, and mental health problems (see Farmer and Lutman, 2012). Social workers need to be aware of the range of specific interventions available within their area, and what is known about their effectiveness. Some of these are

designed to prevent problems from occurring or recurring in families where children are at high risk of being abused and neglected and others to support parents in overcoming problems that are already compromising their children's life chances.

Targeted preventive programmes

Parenting programmes

Safeguarding children is increasingly seen as a public health issue, and the growing prevalence of parenting programmes is part of an international initiative, supported by the World Health Organisation to increase at population levels the prevalence of parenting that is sensitive to children's needs (Eshel *et al.*, 2006). Parent education programmes are almost all designed to help parents to develop appropriate expectations of their children, to respond sensitively to their needs and to learn how to discipline children by using positive techniques rather than corporal punishment (Barth, 2009). Some programmes, such as Triple P-Positive Parenting and the Incredible Years Parent Programme have been implemented on a population-wide basis in a number of countries and have been shown to be successful (Health Scotland, 2007; Prinz *et al.*, 2009; Sanders *et al.*, 2008; Webster Stratton *et al.*, 2011). However, a number of studies have found that community based parenting programmes, designed as universally available services for low-risk populations, or targeted preventive services for families with moderate levels of need, are less effective with parents in families where the likelihood of maltreatment is high (see Casanueva *et al.*, 2008; Chaffin, Bonner and Hill, 2001).

Parent training as targeted prevention for parents with learning disabilities

Parents with learning disabilities are defined as having significant limitations in both intellectual functioning and adaptive behaviour. Both these limitations are likely to have an impact on parenting capability, and these parents benefit from support being provided at the earliest opportunity (MacIntyre and Stewart, 2011). There is evidence that parents with learning disabilities are able to acquire adequate parenting skills to provide sufficient and safe care for a child through parent training programmes (Glazemakers and Deboutte, 2013), home based safety interventions (Llewellyn *et al.*, 2003) and developing supportive peer relationships (McGaw *et al.*, 2002; see also McConnell and colleagues, 2011). However, such parents are likely to need ongoing support to adapt to new challenges as children develop, and to help them address issues such as poverty, social isolation and poor psychological wellbeing (Darbyshire and Kroese, 2012), to which they are vulnerable. They and their children may also need more intensive support if their inherent vulnerability is compounded by other factors such as substance misuse, mental health problems and/or domestic abuse.

Targeted prevention for vulnerable teenage parents: Family Nurse Partnerships

The Family Nurse Partnership programme is specifically designed for vulnerable young pregnant mothers expecting their first child, and their partners, and is now available in 90 English local authorities. It offers intensive, structured home visiting, delivered by a specially trained nurse on a weekly or fortnightly basis until the child's second birthday.

Nurses support the mother's personal health, including use of drugs and alcohol; their environmental health, including attempting to ensure adequate housing and helping her to access optimal support from the community; and life course development. They also work with the mother 'to help her develop the knowledge and skills to confidently support the health and development of her child' and to manage her relationships with others including the baby's father. The nurse also supports the mother throughout the pregnancy, helping her to attend ante-natal appointments and make adequate preparations for the birth (see: https://www.education.gov.uk/commissioning-toolkit). The programme has a strong theoretical basis, drawing on human ecology theory to explore how parents can be helped to consider the impact of their social context on their growing child and develop strong relationships with those who can play a supportive role; on attachment theory to guide parents on sensitive and responsive care-giving and on self-efficacy theory to 'enable parents to understand why particular actions are important, to develop the confidence necessary to achieve these' and to support positive change (http://fnp.nhs.uk/about/history/theories).

The Family Nurse Partnership programme has been extensively evaluated in the USA in randomised controlled trials (Evidence Level A) where it has been shown to be effective in reducing child abuse and neglect and parental welfare dependency. Positive impacts on outcomes for children and young people, such as reduced anti-social behaviour, engagement with the criminal justice system and substance misuse have been shown to persist for at least fifteen years, except in homes where there are moderate to high levels of domestic abuse (Eckenrode *et al.*, 2000; Olds *et al.*, 1986, 1998). When the Family Nurse Partnership programme is delivered by paraprofessionals (home visitors without professional training) rather than specially trained nurses, outcomes have been found to be only half as successful; lack of training may also account for the ineffectiveness of some other home visiting programmes (Olds, 2006). The Family Nurse Partnership programme is currently undergoing an extensive trial (Evidence Level A) in the UK that is exploring the extent to which it improves outcomes for young mothers and their children in comparison with standard services. The potential for pregnancy to act as a catalyst for change was also discussed in Chapter Five.

Specific programmes to address abusive parenting and prevent impairment

Barlow and colleagues (2008) undertook a systematic review of studies evaluating the effectiveness of brief individual or group based parenting programmes designed to treat physical abuse or neglect in high risk families. Only seven studies, however, were sufficiently rigorous to be included in this review. The findings suggest that 'parenting programmes that incorporate additional components aimed specifically at addressing problems associated with abusive parenting (e.g. excessive parental anger, misattributions, poor parent-child interaction) may be more effective than parenting programmes that do not' (p.9).

Some parenting programmes include modules that are specifically tailored to meet the needs of abusive or neglectful parents. These include the Triple P-Positive Parenting Programme (Triple P); The Incredible Years, and Parent-Child Interaction Therapy (PCIT).

The Triple P-Positive Parenting Programme (Triple P)

The Triple P-Positive Parenting Programme (Triple P), is made up of four standard modules, plus a fifth module (enhanced Triple P) tailored specifically to 'parents of children with concurrent child behaviour problems and family adjustment difficulties such as parental depression or stress and partner conflict, (see: http://www.triplep.net/files/4413/6057/1876/The_Triple_P_System.pdf). This module addresses parental attitudes, beliefs and practices that are known to be related to child

addresses parental attitudes, beliefs and practices that are known to be related to child maltreatment such as blame and misattribution, unrealistic expectations, harsh parenting styles, poor parental satisfaction and low efficacy. Triple P has been extensively evaluated in several countries (see http://www.pfsc.ug.edu.au/research/evidence/ for full details). A systematic review and meta-analysis of 101studies (Sanders et al., 2014) shows that Triple P has an enduring positive impact across a wide range of parenting problems and children's behavioural difficulties. A randomised controlled trial (Evidence Level A) with 98 participants found that the enhanced version of Triple P has a particular impact on negative parental attributions, potential for child abuse and unrealistic parental expectations in families where parents' anger management is an issue (Sanders et al., 2004). Tellegen and Sanders (2013) undertook a systematic review and meta-analysis of twelve studies involving 659 families participating in Stepping Stones, a version of Triple P designed specifically for families of children with disabilities. They found significant medium effect sizes in respect of reducing child problems (0.537), satisfaction and efficacy (0.523), and improvements to observed child behaviour (0.523), and a medium to large effect size on improving parenting style (0.725).

There have, however, been some suggestions that the impact of Triple P has been overstated (Wilson *et al.*, 2012), with an attempt to implement it in Birmingham showing little impact (Little *et al.*, 2012¹³). Triple P may prove to be less effective with disadvantaged parents whose children are on the edge of care: a meta-analysis of eleven evaluations (Thomas and Zimmer-Gembeck, 2007) concluded that a high proportion of evaluations of Triple P have focussed on its use with parents in middle or higher socio-economic groups and that 'it is not certain that findings can be generalised to low income or high risk groups' (p.491).

¹³ A randomised controlled trial with a high level of evidence (Level A)

Webster Stratton Incredible Years Parenting Programme

The Incredible Years Parenting Programme is aimed at parents, teachers and children and has been implemented in a number of countries. The programme consists of a basic universal element designed for any parents whose children have behavioural problems and an additional advanced programme designed for use with families with more complex needs, such as those involved with children's social care. The programme is delivered over a number of weekly sessions (see: http://incredibleyears.com for further information). High quality (Level A) evidence from evaluations in the UK of the basic programme shows an improvement in parenting skills (Hutchings et al., 2007; Jones et al., 2007; Scott et al., 2001). The study by Hutchings and colleagues (2007)¹⁴ involved 153 parents living in socially disadvantaged areas with children aged 3-5 years at risk of conduct disorders. It found that the Incredible Years had a significant positive impact on parenting behaviours, with a medium effect size (0.57). A combination of the basic and advanced Incredible Years Parenting Programme has been evaluated in the UK in a single group pre-post study (Evidence Level C) that involved 280 parents of children aged 8-13 at risk of adolescent antisocial behaviour in six local authorities (Hutchings et al., 2011). Data from parents showed high levels of socio-economic disadvantage, substance misuse, depression and a member of the family with a history of crime. The evaluation found statistically significant evidence of improvements to parenting skills and a reduction in parental depression, both with large effect sizes (1 and 0.8 respectively), although the authors concluded that 'the parents of children living in homes where there is a history of criminality may require an additional targeted intervention' (p.141).

Parent Child Interaction Therapy (PCIT)

Parent-Child Interaction Therapy (PCIT) is an individualised intervention developed for parents and children aged 4-7 years old with behavioural problems. Its aim is to alter parents' behaviour through direct coaching strategies. The change in parental behaviour is expected to reduce the child's problem behaviours and improve parent-child interaction. The use of direct coaching and practice of skills in didactic parent-child sessions in which parents are treated alongside children differentiates PCIT from most other parent-training interventions (Thomas and Zimmer-Gembeck, 2007; for further details see Davies and Ward, 2012). A meta-analysis of thirteen evaluations of PCIT (Thomas and Zimmer-Gembeck, 2007) found better evidence for the effectiveness of PCIT when compared with Triple P, for instance, where studies involved a control group of participants on a waiting list for an intervention, were based upon parents' reports of child negative behaviours, or observations of

¹⁴ A randomised controlled trial with a high level of evidence (Level A)

parents' negative behaviours. However, it was thought that these findings could have been the result of weaknesses in the evidence base rather than a genuine disparity in effectiveness.

High quality evidence (Level A) from a randomised controlled trial showed that after about two years, considerably fewer parents receiving PCIT had continued to physically abuse their children (19%) than had those attending a standard community based parenting group (49%). A further trial (Evidence Level A) (Thomas and Zimmer-Gembeck, 2011) tested the most commonly used PCIT standard protocol with 150 mothers at risk of, or with a history of, maltreatment who were randomly assigned at a ratio of 2:1 between PCIT and a waiting list. After 12 weeks, the PCIT group showed significant improvements in parent-child interactions such as giving praise compared with the waiting list group. They also showed a significant reduction in child abuse potential (with a small to medium effect size of -0.4). An association was also found between PCIT and a reduced chance of being referred to children's social care because of suspected child abuse.

Specific interventions to address parental problems that increase the risk of maltreatment

Barth (2009) argues that those elements of parent training programmes that emphasise the development of self-efficacy through learning the skills of sensitive, responsive parenting can also have a positive impact on the types of parental problem that increase the risks of maltreatment: 'first helping parents to be more effective with their children can help address mental health needs and improve the chances of substance abuse recovery' (p.109). Certainly there is evidence that learning how to improve parenting reduces mental health problems (DeGarmo, Patterson and Forgatch, 2004), and that improving mothers' positive interactions with their children can reduce dependency on drugs (Pajulo *et al.*, 2006). Barth (2009) suggests that a staged parenting programme such as Triple P might act as a filter, providing support for those parents whose problems respond to increased self-efficacy, and identifying those who require additional, specialist support alongside parenting interventions.

A relatively high proportion of parents whose children are on the edge of care are likely to require such support. Parents who are trying to overcome issues such as alcohol or substance misuse, mental health problems and/or domestic abuse may need support from a range of specialist adult services, many of which are provided by health professionals. The National Institute for Health and Care Excellence (NICE) produces evidence-based guidance and advice to improve outcomes for people using the NHS and other public health services. The guidance is based on wide-ranging and rigorous reviews

of evidence concerning effective (and cost-effective) interventions¹⁵. The reviews provide detailed evidence of the impact (effect) of interventions, and the duration of effects, as well as indicating the quality of the supporting research (e.g. sample numbers, attrition, and comprehensiveness of data). NICE has produced guidelines on issues such as the management of drug misuse (CG51 NICE, 2007a; CG52 NICE, 2007b); alcohol-use disorders (CG115 NICE, 2011a); common mental health disorders (CG123 NICE, 2011b) and the identification and prevention of domestic abuse (NICE, 2014). There is also guidance on co-existing problems, such as psychosis with co-existing substance misuse (CG120 NICE, 2011c). NICE guidelines indicate which interventions are most likely to be effective for people with different configurations of problems (and which are ineffective and therefore not recommended) and the expected length of treatment programmes. For instance, alongside a number of other effective interventions, detoxification programmes are recommended for substance misusers who 'have expressed an informed choice to become abstinent' (CG52 NICE, 2007b, p.7); these are generally thought to be effective when offered for up to twelve weeks in a community setting, although up to four weeks detoxification in a residential setting is more effective for people who have significant comorbid physical or mental health problems or who require concurrent or sequential detoxification from more than one substance. Following detoxification, six months continued treatment, support and monitoring should be offered, to avoid relapse (CG52 NICE, 2007b).

People experiencing domestic abuse

While there is a wide range of effective interventions for adults who have problems with alcohol and drug abuse and/or mental health problems, there is much less evidence concerning effective interventions for people experiencing domestic abuse, and fewer specialist services available. Following the identification of 'impairment suffered from seeing or hearing the ill-treatment of another' as a form of significant harm in the Adoption and Children Act 2002 (s.120), police now notify children's social care when they are called to a domestic abuse incident in a family where children are involved. Stanley and colleagues (2011) studied the social work response to 184 families following police notification in two local authorities, using a retrospective quantitative design (Evidence Level D), but including the collection of some qualitative data. Sixty per cent of referrals resulted in no further action (although over half of these were re-referred in the 21 months follow-up period), and only 15% received further assessment or family support/safeguarding services. In 23% of cases the response was to send a letter to parents, advising them that the authority was aware of the incident and would take action if it was repeated. These letters were perceived as threatening by parents (and by the

¹⁵ In April 2013 NICE acquired new responsibilities under the Health and Social Care Act 2012 to develop guidance and quality standards for social care in England, to be undertaken through similarly rigorous reviews of evidence. This is expected to provide an opportunity to apply an evidence-based system to decision-making in the social care sector, similar to that provided for the NHS. NICE Guidance on identifying and responding to child abuse and neglect is due to be published in 2016 (see www.nice.org.uk/socialcare).

researchers), and were ineffective in preventing further referrals. The families received some social work support, but it was sporadic and:

the 'stop-start' approach appeared inadequate to effectively address the complex web of long term problems that often involved mental health problems and/or substance misuse and for some families, behavioural or mental health problems in children as well as long-established patterns of domestic abuse (p.309).

The authors concluded that the pattern of repeated referrals, assessments and case closures that this study showed 'will not contribute to building the consistent and trusting relationship between social workers and families that promote disclosures of need and engagement with relevant services' (p.311).

Although men and women in heterosexual relationships may both be perpetrators of domestic abuse, men are the perpetrators in a greater number of incidents and the violence they use is much more severe and more likely to involve controlling their partners through fear (Hester, 2013). A frequent response to domestic abuse is to exclude the male perpetrator from the household. However, as Stanley and colleagues (2011) point out, separation is not an assurance of safety because many men do not 'conveniently disappear' (p.311). Moreover, separation in itself often results in an escalation of violence. Richards's (2004) analysis of multi-agency domestic abuse murder reviews in London found that women were particularly at risk of femicide in the first two months following separation. Domestic abuse provides the most common context for femicide in the UK, with most deaths occurring in the period after a woman leaves her partner (Povey, 2004).

Perpetrators of domestic abuse

The recent NICE (2014) review concluded that there was 'a lack of consistent evidence on the effectiveness of programmes for people who perpetrate domestic violence and abuse' (p.30). However, it did find moderate evidence¹⁶ that individual interventions for abusers may 'improve aggressive feelings towards the partner, attitudinal change, understandings of violence and accountability, and short-term help-seeking' (p.57). Gondolf (2004) undertook a large scale evaluation of male perpetrator programmes in four sites in the USA using 'a naturalistic comparative design' (Evidence Level B). Although the interventions varied, they all used a gender-based cognitive behavioural approach. Those who completed two months or more of the programme were less likely

¹⁶ NICE guidance summarises the overall strength of research evidence (meaning its quality, quantity and consistency) using the following categories: no evidence; weak evidence; moderate evidence; strong evidence; and inconsistent evidence. It does not refer to size of effect. For further information, see: <a href="http://publications.nice.org.uk/methods-for-the-development-of-nice-public-health-guidance-third-edition-pmg4/reviewing-the-scientific-evidence#assessing-the-quality-of-the-evidence#assessing-the-quality-of-the-evidence#assessing-the-guidance-third-edition-pmg4/reviewing-the-scientific-evidence#assessing-the-quality-of-the-evidence#assessing-the-quality-of-the-evidence#assessing-the-guidance-third-edition-pmg4/reviewing-the-scientific-evidence#assessing-the-quality-of-the-evidence#assessing-the-quality-of-

to re-assault their partners than those who did not (36% vs. 55%) (p.618). Moreover, although just under half (49%) of the perpetrators did at some stage re-assault within four years of entering the programme, the rate of re-assault was highest within the first six months (when the men were still in the programme) and subsequently diminished. Furthermore, over two-thirds of the women said their quality of life had improved and 85% felt very safe at follow up. The authors conclude 'There is a clear de-escalation of re-assault and other abuse, the vast majority of men do reach sustained nonviolence' (p.605). However they also found that 'about 20% continuously re-assault'.

Gondolf (2011) argues that there is strong generic evidence for gender-based cognitive behavioural group-work programmes for perpetrators of domestic abuse, and that poor outcomes from some programmes may be more accurately attributed to inadequate implementation and insufficient attention to those who drop out, rather than the approach itself. In the UK, such programmes have been implemented through the probation and the prison services, and include 'a package of inter-agency risk assessment, proactive offender management and structured victim contact from the women's safety service' (Bullock *et al.*, 2010, p iii). However, although the engagement of partners is widely accepted as central to the management of risk, securing their participation has proved to be problematic (*ibid.*, p.13).

People who have been victimised by domestic abuse

There is also evidence from a number of sources that the low self-esteem and isolation of women who have experienced domestic abuse frequently renders them vulnerable to establishing subsequent relationships with violent men – or to returning to previous abusive relationships. Interventions aimed at supporting mothers to improve their self-esteem or repair relationships with their children may prove to be helpful, and some may benefit from arrangements for delivery that suit their characteristics, such as being placed in single gender groups (Hannett, 2007).

The NICE (2014) review found moderate¹⁷ evidence that advocacy services and skill-building interventions may improve self-esteem, coping and decision-making among people who have been victimised by domestic abuse (pp.51-53). Advocacy services may also improve women's access to community resources and therefore reduce the isolation which so many experience (p.51). Sullivan and Bybee (1999) conducted a randomised controlled trial (Evidence Level A) in which 284 women who had spent at least one night in a shelter were assigned after leaving it either to a control group or to receive advocacy

¹⁷ NICE guidance summarises the overall strength of research evidence (meaning its quality, quantity and consistency) using the following categories: no evidence; weak evidence; moderate evidence; strong evidence; and inconsistent evidence. It does not refer to size of effect. For further information, see: http://publications.nice.org.uk/methods-for-the-development-of-nice-public-health-guidance-third-edition-pmg4/reviewing-the-scientific-evidence#assessing-the-quality-of-the-evidence

counselling for ten weeks. Re-abuse rates were high, but at the end of the intervention fewer of those in the treatment group than in the control group reported re-abuse at the two year follow-up (76% vs. 89%). There is also moderate evidence that counselling and psychological therapy interventions are effective in improving some of the psychosocial consequences of victimisation (NICE, 2014, pp.83-4). However, O'Reilly and colleagues' (2010) systematic review of screening and interventions for domestic abuse during pregnancy found little evidence of effectiveness.

Perpetrators of sexual abuse

As with domestic abuse, with which it can overlap, the routine response to sexual abuse is also to exclude the perpetrator from the household. However, many perpetrators of sexual abuse also go on to establish a new, abusive relationship with another vulnerable family (see Cleaver and Freeman, 1995). Most perpetrators are male, and interventions that focus on reducing the vulnerability of the mother or the children may be effective (see MacMillan et al., 2009), but they do not address the abuser. Lösel and Schmuker's (2005) comprehensive meta-analysis of the effectiveness of interventions for sexual offenders included data from 69 rigorous (Evidence Level A) studies published before June 2003. They found that the mean rate of sexual recidivism was 17.5% for those in control groups and 11.1% for those who received treatment. The definition of recidivism 'included outcomes ranging from incarceration to lapse behaviour' (p.120), although the length of follow up is not clearly stated. Overall, programmes designed specifically for sex offenders were found to reduce recidivism, whereas non-specific programmes were found to have no effect. When effect size was calculated, physical treatments showed larger effects than psychosocial treatments. Dropping out from treatment was found to double the likelihood of recidivism, a finding which was considered to emphasise the importance of motivating offenders and engaging them in appropriate programmes.

Impaired personality functioning

There is good evidence that some elements of impaired personality functioning can be treated with long-term intensive therapy. These include emotional dysregulation and an impaired capacity for self-reflection or recognising mental states in oneself and others. Programmes such as mentalisation-based treatment (Bateman and Fonagy, 2010, dialectical behavioural therapy (Linehan, 2000; Linehan *et al.*, 2002), and schemafocussed therapy (Giesen-Bloo *et al.*, 2006) are increasingly available in this country and have been shown to be effective in improving functioning. For instance, Giesen-Bloo and colleagues (2006) compared the impact of schema focused therapy (SFT) and psychodynamically based transference-focused psychotherapy (TFP) in a randomised controlled trial (Evidence Level A) in which 86 patients with borderline personality disorder participated. They found statistically and clinically significant improvements in patients across both groups after one, two and three years of treatment, although significantly more patients in the SFT group recovered or showed clinical improvements in respect of their borderline personality disorder. These interventions are, however, relatively long-term: for instance mentalisation based treatment takes 18 months to

complete, although participants show some improvements after six months (Bateman and Fonagy, 2010), while patients receive schema-focussed therapy for three years (Giesen-Bloo *et al.*, 2006). Where parents have these problems and children are on the edge of care, it is important to ascertain the extent to which parental change in response to these programmes fits in with the timeframe for the child.

A recent NICE review (NICE, 2009) on the treatment, management and prevention of anti-social personality disorder found that group-based cognitive and behavioural interventions are effective in addressing problems such as impulsivity, interpersonal difficulties, anti-social behaviour and offending behaviour (p.20). For instance, the review included five studies (Armstrong, 2003; Liau, 2004; Porporino, 1995; Ross, 1988; and Van Voorhis, 2004) involving group-based cognitive and behavioural interventions for the treatment of offending behaviour in adults. It found the intervention had a modest effect on re-offending with a relative risk of 0.78¹⁸ (National Collaborating Centre for Mental Health, 2010). The NICE review found no high quality evidence for the treatment of people who meet the criteria for psychopathy and dangerous and severe personality disorder; however these people represent only a small proportion of those with antisocial personality traits (NICE, 2009, p.22).

Intensive, multi-faceted interventions for families with complex needs

Many of the studies cited in this report draw attention to 'the complex web of long-term problems' that face parents in families where there are significant child protection issues and children are on the edge of care. Parents' problems may be intricately interwoven; they may also be reflected in, and exacerbated by, children's emotional and behavioural difficulties. Interventions that aim to help adults overcome issues such as alcohol and drug misuse or mental health problems do not necessarily focus on improving parenting capability in families struggling with complex problems.

A multi-faceted approach that integrates a range of services for the whole family may prove to be a more effective means of increasing parental capability than a series of discrete interventions, each aimed at a specific parental problem. Such an approach not only addresses parents' needs, but also those of children whose development may be compromised by abuse and neglect. Some interventions that use this approach also integrate parent-training techniques with a programme tailored to address specific problems in the parent, the child and their relationship, and also explore how protective factors in the wider family and environment can be strengthened in order to maintain progress.

¹⁸ Relative risk (RR) is described as, 'the ratio of the treatment event rate to the control event rate. An RR of 1 indicates no difference between treatment and control'. For example, an overall RR of 0.73 is described as indicating that the event rate associated with the intervention is about three quarters of that with the control intervention meaning that the relative risk reduction is 27%. (NICE, 2009, p48).

Parents under Pressure (PUP)

Parents under Pressure (PUP) is an intensive home-based programme developed in Australia specifically to address the needs of multi-problem families (Harnett and Dawe. 2008). The programme 'combines methods for improving parental mood and parenting skills within a multi-systemic framework that takes into account the contextual influences on family functioning such as social support, housing and child care' (Dawe and Harnett 2007, p.383). Cognitive mindfulness techniques designed to help participants learn emotional regulation have been found to be effective in the treatment of addictions and the prevention of relapse following major depression, and these are incorporated into the programme (ibid.). PUP begins with a comprehensive assessment and case conceptualisation conducted collaboratively with the family: as part of the process. specific targets for change are identified, and these form the focus of the intervention, which is delivered over a ten to twelve week period. A small randomised controlled trial (Evidence Level A) involving 64 parents receiving methadone maintenance at two clinics in Brisbane, Australia (Dawe and Harnett, 2007) assigned participants to a home-based PUP programme (n=22), a brief, two session parenting skills intervention (n=23), and standard care from the methadone clinic (n=19). The study showed PUP to be effective in reducing parental stress and methadone dose, and there were significant improvements in children's behavioural problems. At the six month follow up, risk of abuse measured using the Child Abuse Potential Inventory had decreased from high to low for 36% (8/22) of the sample receiving PUP and 17% (4/23) of those receiving the brief intervention. There was no decrease in risk for families receiving standard care and in fact the level increased for 42% (8/19) of these parents. The risk of abuse did not increase for any of the parents receiving PUP; however it should be noted that 36% (8 /22) of them remained at high risk throughout the study. The PUP programme is currently being evaluated in a UK context across eleven NSPCC service centres. including a randomised controlled trial at six centres (see: http://www2.warwick.ac.uk/fac/med/about/centres/wifwu/research/pup/ for further information).

Multi-Systemic Therapy for Child Abuse and Neglect (MST-CAN)

Multi-Systemic Therapy for Child Abuse and Neglect (MST-CAN) (Swenson *et al.*, 2010a), is an example of a more complex multi-faceted intervention, designed specifically for families where physical abuse is a cause for concern. It adopts well-recognised techniques to engage families by interviewing each member to ascertain their views of desired outcomes, and these are then agreed as over-arching goals for treatment. A comprehensive assessment of the family's strengths and needs is then undertaken, and the drivers (or fit factors) of target behaviours are identified; for instance harsh discipline may be associated with parental alcohol misuse, depression and poor parenting skills, and children's non-compliance. The most powerful drivers are prioritised and evidence-based interventions are implemented with the support of the wider family and community. Progress is routinely assessed and the programme adjusted and tailored to meet needs until desired outcomes have been achieved. The programme is usually delivered through

three sessions a week, over six to nine months. Where there are serious safeguarding concerns, all family members agree a child safety plan and the practice team work closely with social workers in statutory services to ensure that social work decisions can be informed by clinical need or progress (for further details, see Davies and Ward, 2012, pp.160-161). MST-CAN has been rigorously evaluated (Evidence Level A) in a trial in which 86 physically abused young people and their parents were randomly assigned to receive MST-CAN or Enhanced Outpatient Treatment (EOS) (the standard service plus enhanced engagement and parent training interventions). Sixteen months after entering the programme, the percentage of parents in the MST-CAN group exceeding clinical thresholds for psychiatric distress had decreased (from 20.5% to 5.3%), and the percentage of young people scoring in the clinical range on self-reported post-traumatic stress disorder (PTSD) symptoms had dropped by half (from 17.8% to 8.9 %). In contrast, in the EOT group, parents showed virtually no change on the first measure (16.7% to 15.8%), and the young people showed slightly greater evidence of PTSD (19% to 21.4%). During the study period the MST-CAN young people reported about half as many incidents of severe assaults by their parent as did those in the EOS group (4.7 vs. 9.8 incidents) indicating that the programme does appear to reduce the incidence of physical abuse, although it does not eradicate it (Swenson et al., 2010b). At the time of writing, MST-CAN is being piloted or implemented in Leeds, Cambridge and Greenwich in the UK, and also in the Netherlands and Switzerland as well as the USA, where it originated (http://www.mstcan.com/programs/).

Systematic reviews undertaken as part of the Safeguarding Children Research Initiative identified a range of multi-faceted intensive interventions, developed using this approach, which appear to be effective in addressing physical abuse, and emotional abuse and neglect (see Barlow and Schrader McMillan, 2010; Montgomery *et al.*, 2009; for further details see Davies and Ward, 2012, Chapter Five). Two of these interventions (Functional Family Therapy and Multi-Systemic Therapy) are being implemented across the UK with support from the Department for Education, with the specific purpose of preventing children from coming into care or custody (Department for Education, 2013a). UK trials are still ongoing for these programmes; not all authorities have access to them. Not all children in families with a complex web of long-term problems will benefit from them, but trials have shown that more will benefit and to a greater degree than those who receive routine services.

Multi-faceted, integrated programmes such as those described above offer a template for addressing the complex web of entrenched problems that beset families struggling to overcome abuse and neglect, and their consequences. As services develop it is possible that a 'common elements' approach that identifies those interventions most commonly required in flexible configurations to meet these complex needs can be developed to support training and service delivery (see Bentovim and Elliott, 2014).

Multi-faceted programmes for families with complex needs do not replace social work services, but are part of a planned, multi-agency intervention. They offer intensive

support for a limited period. They need to be dovetailed with other services and to be part of a child protection plan that reflects the need to step-up and to step-down the intensity of support as required.

Conclusion

This chapter has highlighted the various elements to be considered in identifying effective interventions designed to support parents in overcoming problems that compromise children's welfare, and help them meet their needs. Many of the interventions now available will change as research informs new developments in effective practice; this chapter provides an overview of the types of service that have currently been shown to be most effective in promoting change.

Key points from Chapter Six

- Some interventions are ineffective and others are harmful. It is important to
 understand how to interpret evidence of effectiveness, not all of which is of
 equivalent quality. No intervention is 100% effective: it is important to note what
 percentage of participants are likely to change and how much greater this is than
 the impact of standard services.
- There is very little systematic evidence concerning the elements of effective social work. Social workers who are open, straightforward and clear-sighted, and who adopt a position of respectful uncertainty appear to be most effective in supporting parents through the process of change.
- Organisational barriers such as an unstable workforce, changes that are built into the system and restricted resources can limit the effectiveness of social work interventions.
- Intensive Family Preservation Services (IFPS) have been shown to improve family functioning. However changes have not been maintained in the long term with families with entrenched and complex problems. IFPS delay children's entry to care and reduce the time they spend away from home, but they do not reduce the likelihood of them becoming looked after.
- Abused or neglected children tend to do better in care than those who remain with or return to parents who are unable to change. About one in three maltreated children who are reunified with birth parents become looked after again within six months.
- Parent training programmes have been shown to be effective in the general population, but only those which include specific modules tailored to address attitudes, beliefs and practices related to maltreatment are likely to be effective with parents whose children are on the edge of care.

- Parent training can help learning disabled parents to acquire adequate parenting skills to provide sufficient and safe care, but such parents are likely to need longterm support to adapt to new challenges.
- The Family Nurse Partnership programme is effective in reducing child abuse and neglect in vulnerable young pregnant mothers expecting their first child.
- Professionals offering substance misuse, alcohol abuse and mental health services under the NHS are required to take NICE guidelines fully into account in selecting services that have been shown to be effective.
- Gender-based cognitive behavioural group work can be an effective intervention for perpetrators of domestic abuse. Such interventions are often court-mandated and require close liaison with the probation service. Advocacy services and skillbuilding interventions may improve self-esteem, coping and decision-making among people who have been victimised by domestic abuse.
- Specific interventions for sex offenders can be effective but, even with treatment, just over one in ten will re-offend.
- Intensive multi-faceted integrated interventions for families with complex needs are more effective than routine services, but should be dovetailed with other services to form a child protection plan that can offer different levels of support as required.

Chapter Seven: Relapse and the maintenance of change

Introduction

The three previous chapters have first examined how parents whose children are on the edge of care may initially find it difficult to acknowledge that adverse behaviour patterns risk causing harm to their children, and then considered how they can become motivated and engage with a wide range of services and interventions designed to support them through the process of change. However, successful engagement in an evidence-based intervention is not the only concern for social workers assessing parental capacity to change. Within the assessment they also need to consider those factors which are known to affect the extent to which change is likely to be maintained in the long term.

Many of the models for assessing or understanding the change process discussed in previous chapters involve considering the likelihood of change being sustained in the long term. For example, relapse is a discrete stage of Prochaska and DiClemente's Trans-Theoretical Model of Change (TTM) (Prochaska and DiClemente, 1982), and Harnett's (2007) extended procedure for assessing parents' capacity to change involves considering what further support and intervention parents will require to maintain progress once it has been made. Models such as these can assist social workers in conceptualising the change process, and provide evidence concerning whether, and for how long, change is likely to be sustained.

Despite the development of various models and validated assessment tools, the evidence base is not sufficiently robust to predict parents' future behaviour accurately; nor is it ever likely to be so because unforeseen events and circumstances will always have the potential to disrupt the change process. The courts rely upon social workers using their professional judgment to decide whether parents are likely to sustain change in the light of all they know about the circumstances of a case and informed by evidence from a range of souces, including that obtained using validated assessment tools and the evaluations of effective interventions, which show their likely impact. Assessments also need to consider what would be the likely impact of a relapse on the child's safety and wellbeing, and this will depend on their age, vulnerability and experience of abuse and neglect. This chapter looks at what the research evidence tells us about maintenance of change following completion of an intervention; the role of relapse within the change process; and the factors that potentially support or undermine parents attempting to sustain change in the long term.

The impact of different types of problem on sustained change

As previous chapters have discussed, parenting capability can be compromised for a variety of reasons including substance misuse; domestic abuse; mental health problems; or because parents never had a satisfactory parenting role model themselves. The ability of parents to sustain change in the long term will depend to an extent upon the type and number of difficulties they are trying to overcome. Some problems can be completely resolved, whereas others can only be controlled or alleviated in such a way that the risk of serious harm to the child is reduced. This is an important distinction to recognise as it may have implications for a parents' ability to maintain long-term change. Where a parent has never experienced appropriate parenting themselves and lacks a role model, gaps in knowledge may be addressed through engagement in a parent training programme. However, where the underlying problem can only be controlled or alleviated, parents face an ongoing challenge as they attempt to meet their children's needs whilst continuing to deal with continuing adversity.

Chronic mental health problems and learning disabilities are two conditions which may always be present and require continuing support or treatment to enable successful parenting. Research has produced mixed messages about the ability of parents with learning disabilities to retain and use parenting skills in the long term (Feldman *et al.*, 1989; Peterson *et al.*, 1983). There is also mixed evidence on how far they are able to transfer skills learnt into new contexts and situations (Booth and Booth, 1993). McGaw and Sturmey (1994) suggest that to be most effective, skills training should be adapted to the needs of the individual and may need to continue over a period of years. Parents' ability to cope will depend upon the extent of their learning disability and contextual issues such as the support of a partner or family member without learning disabilities as well as long-term support from services (McIntyre and Stewart, 2011, p.6).

How long is needed to establish whether parents will successfully maintain change?

A challenge for social workers in assessing parental capability is that it cannot be assumed that after sufficient progress has been made, a parent will successfully continue to maintain change. In those situations where parents are struggling with a complex web of problems, capacity to maintain change will only become evident over time, and the process of sustaining change may require continuing support. For example, effective interventions are available to help parents overcome substance misuse, but long-term abstinence is required before it is possible to say with confidence that sustained change has been achieved. Laudet and White (2008) describe recovery from substance misuse as 'a process that unfolds over time rather than a time-limited 'event' (p.28) and compare it with having a chronic condition such as diabetes or hypertension.

A study by Hser and colleagues (2001), reporting on a 33 year follow up of 581 male heroin users, found that one in four still relapsed after fifteen years abstinence (p.507). They also found that for those individuals who stopped using heroin within ten years of initial use, achieving stable recovery still took a period of eight to ten years (Hser, Longshore and Anglin, 2007). Similarly, Moos and Moos (2006) found that 42% of people who received treatment for alcohol misuse had relapsed within sixteen years (p.216). Hibbert and Best (2011) also found recovery from alcohol addiction to be a gradual process of change. Their cross-sectional study of 53 people recovering from alcohol misuse identified those who had been sober for between one and five years as being in early recovery and those who had achieved sobriety for longer than five years as in stable recovery. However, they found that five years did not represent a plateau, because considerable further improvement occurred beyond this point. These studies of drug and alcohol addiction show that recovery from some problems may realistically extend throughout a child's formative years. The findings from these studies all support the argument for providing long-term, light touch support to help parents maintain progress once an intensive period of intervention has been completed. However there is considerable evidence to suggest that pressure to close cases means that social work services are often withdrawn prematurely, with inadequate arrangements for stepping down support (Farmer et al., 2011; Farmer and Lutman, 2012; Ward, Brown and Westlake, 2012).

Chapter Six discussed what research tells us about the effectiveness of interventions intended to support parents through the process of change. The remainder of this chapter will explore what is known about the process of recovery and relapse, and identify some of the factors known to support or undermine sustained change.

The process of recovery and relapse

In view of the continuing challenge of maintaining change, it is unsurprising that parents experience instances of relapse within the recovery process. Although relapse is a term commonly associated with recovery from drug and alcohol addiction, its definition, 'to deteriorate after a period of improvement' can equally be applied to a parent who stops adhering to medical treatment or who loses confidence in their new parenting skills. In the field of substance abuse recovery, relapse is viewed as an integral part of the change or recovery process. Hser and colleagues (2001) referred to heroin users experiencing 'repeated cycles of remission and resumption' (p.503). Scott and colleagues (2005) monitored the recovery of drug users over a three year period and identified significant movement in both directions between abstinence and relapse. They described a 'chronic and cyclical nature of addiction and treatment'.

Although the literature describes relapse as part of a process of gradual recovery, risk of relapse is highest in the early stages (Laudet and White, 2008). As we have already seen (Chapter Six), although up to half of male perpetrators of domestic violence assault their

partners again, there is a marked de-escalation of assault after the first six months of treatment (Gondolf, 2004). A follow up study of the Parent-Child Assistance Program (PCAP) aimed at women who used drugs and alcohol heavily during pregnancy found that as the recovery cycle progressed, mothers accrued longer periods of abstinence between instances of relapse (Grant *et al.*, 2003). Relapse is more likely to occur where there is temptation (i.e. reminders of positive aspects of the behaviour); where there is perceived need (i.e. stress relief); or in the event of boredom, although there may be no obvious trigger (West, 2011). Social workers and parents should not therefore view relapse as a failure, or an indicator that change will never be maintained, but should regard it as a natural stage in the recovery and change process.

Virtuous circles

It is important to note that a key feature of the recovery and change process is the way in which achieving one small step has been found to enhance parents' sense of selfefficacy or feelings of motivation, thereby propelling them on to achieve further change. In essence, a virtuous circle is formed where success in one area leads to success in others. Hibbert and Best (2011) found that as people recover from alcohol abuse, over time, their feelings of depression reduce and their perceptions of self-efficacy increase. It also seems likely that improved perceptions of self-efficacy may help vulnerable young women break a cycle of engaging in abusive intimate relationships (Wathen and MacMillan, 2003). Jones and Prinz (2005) describe how higher levels of parental selfefficacy are likely to lead to more successful parenting practices, and how these in turn reinforce those feelings of self-efficacy. Higginson and Mansell's (2008) small qualitative study of the process of psychological change found that individuals viewed themselves as changed people after they successfully overcame a problem in their lives, and that this gave them a new found sense of confidence in their capabilities. It is therefore important to support parents through the change process so that virtuous circles can be reinforced, and more positive self-images established and maintained. Recording progress and celebrating achievements will allow both parents and practitioners to measure the distance travelled and support the process of change (Dawe and Harnett, 2007).

Factors supporting or undermining the maintenance of change

As well as identifying the risk of relapse, the research evidence also points to a number of factors that have the potential to support or undermine a parent in maintaining long-term change. In many respects these factors mirror the *internal* (client related) and *external* (service provision) factors identified in Platt's (2012) Integrated Model of Parental Engagement (discussed in Chapter Five) and we therefore use the same categories in the following discussion. The following paragraphs do not offer an

exhaustive list of factors supporting or undermining sustained change, but indicate the broad range of those involved at this stage of the change process.

Examples of factors supporting change

Internal factors supporting change are those which relate to the individual, including personal, psychological or behavioural factors. Relapse has been found to be less likely among individuals belonging to certain demographic groups, including those who are female or older, married and better educated (Moos and Moos, 2006). People with higher levels of self-efficacy have been found to have an increased likelihood of maintaining change (Cummings et al., 2010; Moos and Moos, 2006). Having a place in society, for example, finding a role or identity, making a positive contribution and 'giving something back' have all been identified as supporting continued change. For example, Duffy and Baldwin (2013) found that former substance misusers wanted to become involved in educating young people about the dangers of drugs. Studies of overcoming substance misuse have also found that the opportunity to assume a 'normal' role in society such as a student, employee, volunteer or parent were viewed as accomplishments and contributed positively to recovery (Skinner et al., 2010; VanDeMark, 2007). Being a parent in itself is a powerful factor that supports the achievement and maintenance of change, but the stresses of resuming the role if children have been cared for elsewhere can undermine progress. These issues are further discussed below.

External factors supporting change are those related to personal circumstances and to the provision of services or interventions. The existence of positive support networks has been identified as a factor supporting change (Hibbert and Best, 2011; James, 2004). Duffy and Baldwin (2013) found that 'family contact was the key motivator stated by participants for maintaining their recovery' (p.6). This study also found that peer support played an important role in recovery. In relation to parents with learning disabilities, having the support of someone without a learning disability has been described as 'one of the single most important factors influencing the ability of parents to manage' (McIntyre and Stewart, 2011, p.6).

Support from professionals is also important in maintaining change. Proactive social work support throughout the process is associated with successful reunifications from care (see Farmer and Lutman, 2012). Incorporating parenting training into interventions addressing other issues may also increase the likelihood of successfully maintaining behaviour change (Porowskia, Burgdorf and Herrell, 2004), and this approach may be as relevant to fathers as to mothers (Collins *et al.*, 2003).

Tailoring services in response to parents' individual needs and circumstances has an impact on their ability to sustain change. This includes making support available before they reach crisis point and continuing support for as long as is necessary. This is a particularly important issue when parents have learning disabilities. Like all parents they will face different challenges as their children grow older; however, because they need extra support to adapt and to learn, their attempts to continue parenting successfully may

be undermined by the pace of their child's development if they do not continue to receive adequate training and support to help meet these changing needs (Accardo and Whitman, 1990; McGaw, 2000).

It is also worth noting that people's priorities have been found to alter as they progress through the process of change or recovery, with the result that certain factors may have a different impact depending on whether a parent is in the early or later stages of the process (Laudet and White, 2008, 2010). Although substance misusers may initially consider abstinence to be their greatest priority, over time issues such as finding employment or looking after their health may become increasingly important (*ibid.*). Laudet and White (2010) also produced further findings of particular relevance to professionals engaged in supporting changing behaviour: this study showed that achieving abstinence did not automatically lead to improvements in other areas of individuals' lives. This suggests that a wraparound approach is required, which looks at supporting people not only to overcome the particular difficulties that compromise their parenting, but that also enables them to reduce continuing stressors, such as poor housing or living conditions, and unemployment.

Examples of factors undermining change

Negative emotions are among the *internal factors* that undermine change. Feelings of stress can trigger a return to active addiction (Laudet and White, 2008). Likewise, low self-efficacy in relation to parenting and negative feelings about the impact of past drug use on children have been found to undermine parents' attempts to sustain change (Baker *et al.*, 2003; Greenfield *et al.*, 2000). VanDeMark (2007) found that having the parenting role removed can undermine long term change. However, this must be considered in the light of any risk to the child of continued co-residence or contact.

The impact of co-morbidity on the process of change has been discussed throughout this report and also needs to be considered in relation to parents' ability to sustain change. Charney and colleagues' (2010) follow-up study of individuals receiving treatment for alcohol addiction found that those who had a secondary substance misuse problem were more likely to slip up or relapse in their drinking within the first four weeks of abstinence; such a relapse was then found to be a predictor of alcohol consumption at twelve weeks (p.46). Parents who are struggling with combinations of problems such as poor mental health, substance misuse or domestic abuse are less likely to be able to maintain change in the long term (Duffy and Baldwin 2013; Skinner *et al.*, 2010).

External factors which can undermine change include having an inadequate support network. James (2004) highlights the importance of family members in supporting parents with learning disabilities; Edmonson and Schneider (2001) found that the risk of family breakdown in this situation could be minimised by involving as many people as possible in supporting the family to create a circle of support. However McConnell and colleagues (2008) found that parents with learning disabilities often received limited

social support from family, friends or their community, leaving them reliant upon formal support services.

Falkin and Strauss (2003) found that women who misused drugs were likely to become cut off from their social networks. Consequently, attempts at recovery could be compromised by their finding themselves reliant for support upon the very people who were facilitating their continuing drug use. However they were also reluctant to break ties with those who prevented their recovery, as these 'may have also been the main people who were there for them in many constructive ways' (p.142). Other studies have also found that drug or alcohol users who become cut off from their family and friends can find themselves having to live or associate with other substance users, and may then form intimate relationships with them (Carlson *et al.*, 2006; Porowskia *et al.*, 2004; Skinner *et al.*, 2010).

Women who leave their homes to escape abusive relationships may find themselves isolated. This may be a particular problem for BMER women and disabled mothers, who may previously have been reliant on their communities for social support (see Stanley *et al.*, 2012b). Loneliness and a loss of social support can increase their vulnerability and become a major factor in their re-establishing an abusive relationship or entering into a new one (see Ward, Brown and Maskell-Graham, 2012).

There is also evidence that continuing exposure to factors such as poverty, financial instability, debt, and poor housing add to parents' stress and increase the likelihood of relapse (VanDeMark, 2007). MacIntyre and Stewart (2011) identify a tendency for some professionals to assume too readily that parents with learning disabilities are prevented from parenting effectively by their disability, when in reality they too are struggling with the same lack of resources that compound the difficulties faced by many other parents whose children are on the edge of care.

The role of parenthood in supporting and undermining change

Parenthood is a key motivator in sustaining change; however there is also evidence that, if inadequately supported, the stress of resuming greater parental responsibilities can undermine progress. Firstly, there is a wealth of research evidence to show that being a parent motivates people to maintain progress and avoid relapse. Duffy and Baldwin (2013) found that being a parent prevented people from relapsing back into drug use because they wanted to avoid the feeling of 'not being there' for their children (p.7). Kim and colleagues (2006) found that fathers' relationships with their children had a positive impact on their substance use during residential treatment and on recovery. The authors described the parent-child relationship as potentially playing 'a key function in supporting adult individuals throughout the recovery process and encouraging maintained abstinence' (p.92).

The opportunity to retain or regain custody of children is also a strong motivator for achieving and sustaining change (Carlson *et al.*, 2006; Sword *et al.*, 2009). However

resuming the parental role may bring with it additional stresses, for reunification can be a lengthy and sometimes problematic process (Bullock *et al.*, 1993). It is important to ensure that progress is sufficiently advanced for parents to cope with the emotions and stress that reunification may bring (Carlson *et al.*, 2006). Stressful family relationships can undermine the changes parents have made (Duffy and Baldwin, 2013). For example, family dynamics may become strained when parents are able to resume their role and take over the care of children from grandparents who have previously stepped in to help (Knis-Matthews, 2007).

Stressful family relationships are not the only issues which can undermine parents' ability to maintain long-term change during this period. Carlson and colleagues' (2006) study of mothers who were reunited with their children after overcoming substance misuse identified difficult relationships with social services as a stressful factor which could threaten their progress. Poorly managed reunification can also have an undermining effect on sustained change; family reunification involving multiple children or reunification following a significant period of separation can be particularly overwhelming for parents (Carlson *et al.*, 2006; Cordero 2004; Hess and Folaron 1991). Reunification is also particularly difficult for children who return home separately or are reunited with siblings who have been able to remain with both parents (Wade *et al.*, 2011), and this can exacerbate the family tensions that add to parental stress.

The availability of resources is also a major factor in parents' ability to sustain change. Inadequate support networks mean that parents may lack the necessary help with child care from family or friends. Harnett (2007) identified this as a major factor that prevented parents from attending ongoing treatment. Finally, being unable to access the resources needed specifically for resumption of the parental role, including a safe living environment, food or recreation facilities can undermine a parent's efforts to maintain change (Carlson *et al.*, 2006).

Relapse and harm to the child

When parental recovery suffers a setback, the relapse can impact on both the child's current and long-term safety and wellbeing. The immediate impact may be similar to that of initial exposure to parental problems in that parenting capability may diminish and, unless there are supportive adults available who can compensate and protect them from harm, children may experience the lack of basic care, stimulation, emotional warmth, and so on that contribute to abuse and neglect (see Chapter Two for further details). There may be immediate safety issues if, for instance, children are exposed to physical abuse from a parent whose self-control is diminished by alcohol, or if they are left with access to drug paraphernalia. Relapse will have a different impact depending on the age and vulnerability of the child. Nevertheless, evidence from a number of sources also indicates that the longer children are exposed to abuse and neglect, the greater will be the impact, and the more difficult it will be to overcome (see for instance Hildyard and Wolfe, 2002; Rossman and Ho, 2008). We know, for instance, that behavioural problems

and symptoms of post-traumatic stress disorder (PTSD) increase the longer that children are exposed to domestic violence (Rossman and Ho, 2008), and that the chronicity, rather than the type or severity of maltreatment is the strongest predictor of negative outcomes (Bolger and Patterson, 2001). The impact of relapse needs to be assessed within this context.

Finally, relapse has its own impact in that it can undermine children's sense of stability and security. Relapse may mean that children experience repeated attempts at reunification followed by readmissions to care; those children who move constantly between home and care in an unplanned manner tend to have the least satisfactory outcomes (Wade *et al.*, 2011).

Conclusion

Social work support is often provided for a relatively short period of time. Where parents are able to progress successfully through the stages of change there is a strong case for light-touch monitoring to ensure that their progress is maintained in the long term (see Davies and Ward, 2012). Parents who have maltreated their children are likely to be struggling with deep-seated and entrenched problems that are extremely difficult to overcome. Intensive long-term support for those who are struggling to maintain change is therefore necessary (see Laudet and White, 2010). Even where change appears to be established, swift withdrawal of services and premature case closures are likely to be unreliable and ineffective in safeguarding children in the longer-term (Ward, Brown and Westlake, 2012).

This chapter has highlighted how successfully engaging in and completing an intervention is only one step in the change process. For many parents, the real challenge will be in maintaining that change in the long term. The challenge for social workers will be in assessing whether, in light of what they know about the parent, the nature and number of issues faced, factors which may support or undermine long-term change and the support available for the future, they consider that a parent is likely to achieve and sustain change within the child's timeframe.

Key points from Chapter Seven

- The ability of parents to sustain change in the long term will be affected by the type and number of difficulties they are trying to overcome, and whether these can be fully addressed or only controlled and alleviated.
- Recovery from problems such as substance misuse is a gradual process extending over a period of years rather than a time limited event.
- Relapse forms a natural part of the recovery cycle.
- A key factor in the recovery and change process is the establishment of virtuous circles in which success in one area leads to success in others.

- Internal factors supporting the maintenance of change include certain demographic factors; self-efficacy; having a 'normal' role in society. External factors include positive support networks and appropriate support from professionals.
- Internal factors that undermine the maintenance of change include stress, negative emotions, and co-existence of problems. External factors include isolation, inadequate support networks and poverty.
- Parenthood is a key motivator in sustaining change; however, if parents are inadequately supported, the stress of resuming greater parental responsibilities can undermine progress.
- Parents' priorities are likely to change as they progress through recovery, so different factors will be relevant at different stages of that process.
- Change or recovery in one area of parents' lives does not automatically result in improvements to other areas.
- Relapse may have a negative impact on children's wellbeing because they may be at risk of immediate harm; continuing exposure to abuse and neglect may compromise their development; and they may experience greater instability.
- Assessments will need to consider whether, in view of what is known about the child's needs, the issues parents face, the factors which increase or decrease the likelihood of relapse, and the support available for the future, sufficient change can be achieved and sustained within the child's timeframe.

Chapter Eight: Conclusion: Making use of the research findings to inform professional practice and assessments required by the courts

Introduction

The preceding chapters of this report present the findings from a body of research that adds to our knowledge of how, and in what circumstances, parents are able to overcome abusive or neglectful behaviour patterns that place their children at risk of significant harm. The research is drawn from a wide range of disciplines and focusses on parents in families where there are significant child protection concerns and children are on the edge of care. The report is intended as a reference resource for social workers in their work to support families where children's safety and developmental functioning are at risk. However the research findings should also help social workers and children's guardians to produce more focussed and robust assessments of parenting capability and parental capacity to change, and assist judges and other legal professionals in evaluating the quality of assessment work presented to the courts by the local authority and Cafcass. This last chapter examines how these research messages might support social work practice with families and more specifically, how they might be used to inform assessments required by the courts in cases where the threshold for significant harm is thought to have been reached and there has been a decision to instigate proceedings.

The President of the Family Division has set out the courts' expectations of the social worker's statement and the children's quardians' case analyses within the revised Public Law Outline (Munby, 2013a, 2013b; Re B-S 2013). There is an emphasis on assessment and analysis, and on the production of succinct, evidence-based reports, supported by other materials if required. Materials placed before the court by the local authority and the children's guardian will include, among other requirements: a threshold statement showing significant harm or likelihood of significant harm (usually drafted by a local authority lawyer); assessments of the child's needs and the parent's capacity (or capability) to meet those needs; an analysis of why there is a gap between parental capacity (or capability) and the child's needs; assessment of other significant adults who are potential carers; and a plan for the child that is consonant with the evidence accepted by the court and produced by these assessments. The expectation is that social workers and children's guardians are already experts in these matters, and that they will play a leading role in undertaking assessments, although they will consult with colleagues from a wide range of other agencies; and that additional assessments from specialists will only be sought in those cases where parents or children have needs that are outside the area of social work expertise. In the following paragraphs we have indicated where the findings from this overview of the literature might inform the assessments for the courts to be made by social workers and children's guardians in care or related proceedings. Findings have particular relevance for assessments of children's needs, parental capacity (or capability) to meet needs, and an analysis of why there might be a gap between parental capability and the child's needs.

Assessment of parents' capacity to meet needs (parenting capability)

Chapter Two presents up to date findings from empirical research that seeks to clarify the circumstances that lead to parental problems such as domestic abuse, mental health problems and substance misuse, and the ways in which these interact with one another and with learning disability, as well as with parents' previous histories and experiences, to compromise their ability to meet their children's needs. This type of evidence, and its implications for children's development, is fundamental to an assessment of parents' capacity to meet children's needs (parenting capability). Parenting does not take place in a vacuum, and such an assessment also needs to take account of the research data which show how environmental factors such as poverty, debt, poor or overcrowded housing, unemployment and the debilitating experience of living in a dangerous or violent neighbourhood can interact with other risk factors and reduce parenting capability (see for instance Fernandez, 2007; Jack and Gill, 2013). Assessments of parenting capability should identify strengths as well as difficulties and, where appropriate, the information should be checked and discussed with the child and their parents or carers (HM Government, 2013, p.19)

Chapter Two also provides examples of the ways in which parents' problems can impact on their ability to meet children's needs across the six dimensions of parenting capability set out in the Assessment Framework: basic care, ensuring safety, emotional warmth, stimulation, guidance and boundaries and stability. Diminished parenting capability in these dimensions increases the likelihood that abuse and neglect will occur. For instance, substance misuse, problem drinking and some mental health problems can all lead parents to misinterpret experiences and observations and/or reduce their ability to control feelings of anger and frustration, so that they cannot ensure their children's safety. These are the types of indicators that can inform social workers' assessments. Concrete examples of how parenting capability has been reduced, taken from social workers' own observations, from talking and listening to children, from parents' own reports and those of other interested parties, including family members and professionals from other agencies, should be included in statements for the courts. Such information should be supported by the results from validated instruments and scales and, where there are specific, additional concerns that are outside the area of social work expertise. specialist assessments from professionals such as psychologists, psychiatrists, probation officers and drug counsellors.

Assessments of children's needs

Chapter Two also presents up to date findings from empirical research concerning abuse and neglect and its long-term impact on children's development. The research evidence provides a basis for understanding the harmful impact of maltreatment across the whole spectrum of children's development and sets out how, unless adequate action is taken, there can be long-term adverse consequences throughout adolescence and into adulthood. The research evidence should help professionals with safeguarding responsibilities understand how abuse and neglect impact on parent-child interactions, attachment styles, and children's physical, cognitive and psychosocial development. Emotional abuse and neglect are particularly difficult to recognise because they rarely produce a crisis, but the chronic, long-term exposure to these forms of maltreatment has a corrosive impact on children's development (see for instance Norman *et al.*, 2012); the research evidence can identify how this happens and help practitioners understand why these issues are important.

The research evidence also shows how some children, such as those who are very young, those with health problems or disabilities, or those with emotional or behavioural difficulties, can be particularly challenging for parents, and are disproportionately likely to experience abuse and neglect (Hindley, Ramchandani and Jones, 2006; Sullivan and Knutson, 2000). These findings should alert practitioners to the potential significance of concerns about abuse and neglect when children show these characteristics.

Understanding this type of research evidence is fundamental to assessments of children's needs, which should explore how well children are functioning across all dimensions of development and examine whether, and to what extent, this is being compromised by abuse or neglect. Chapter Two presents research evidence showing ways in which abuse and neglect have been found to impact on development from before birth to adulthood. The Assessment Framework sets out seven dimensions of development which assessments should cover: health (including physical development). education (including cognitive development), identity, family and social relationships, emotional and behavioural development, self-presentation and self-care skills. Assessments of children's needs should identify strengths as well as difficulties and, wherever possible, be discussed with the child and their parents (HM Government, 2013, p.19). Neglectful or abusive parenting may have a different impact depending on the age and vulnerability of the child. Concrete examples of the ways in which children's development appears to be compromised by abuse and/or neglect, taken from social workers' own observations, or from those of other professionals such as health visitors and teachers, should be included in statements to the courts. Such statements should be supported by the results from validated instruments and scales, reports from other family members, reports from professionals and, where there are specific, additional concerns that are outside the area of social work expertise, specialist assessments from professionals such as paediatricians, child psychologists and psychiatrists.

Assessing the likelihood of significant harm

Assessments of children's needs that show impairment of health or development should help to determine whether the threshold for significant harm is likely to be reached. A key question that also needs to be resolved is whether children will be exposed to a continuing risk of significant harm if they remain with their birth parents. In a small number of cases the risks are so serious that it is obvious that children cannot safely remain. However in the majority of cases it is important to weigh up the various factors that indicate the extent to which children are likely to be protected in the future. Practitioners who have established a relationship with parents and are aware of their history are in a unique position to make such assessments, but professional judgments alone have been found to be insufficiently reliable as a means of assessing the likelihood of future harm (see for instance Arad-Davidson and Benbenishty, 2008; DePanfilis and Girvin, 2005; Dawes, Faust and Meehl, 1989). There is a strong case for practitioners to use standardised, evidence-based tools as a means of supporting, though not replacing structured professional decision-making. However Barlow, Fisher and Jones' (2012) systematic review of tools available to assist practitioners in assessing the likelihood of current or future significant harm found that most need further validation or piloting in a UK context – an initiative that is currently under way.

Analysis of why there might be a gap between parental capability and the child's needs

Case conceptualisation

There is considerable evidence to suggest that practitioners are usually competent at collecting information, but that they find it more difficult to analyse it and draw out the implications when developing care plans or writing court reports (see for instance Turney et al., 2011). Analysis requires some form of conceptual map or model that serves as a guide for the practitioner, helping them to make sense of information, weigh up its significance and draw out conclusions. The Assessment Framework is recommended by Working Together (HM Government, 2013) as a simple ecological model, which can form the basis for core assessments. Models such as this can facilitate the development of a case conceptualisation – a hypothesis concerning the various internal and external factors that affect a parents' ability to meet children's needs, the strengths and weaknesses within the child, the family and the environment, and the issues that need to be addressed. This type of conceptual map allows practitioners to move from observation to analysis; it helps them to make sense of the interrelationship between factors in each domain and dimension, and set goals for the process of change.

As previous chapters have shown, there is no shortage of more complex conceptual models, designed to guide practitioners through the processes of assessing risks of future harm, parents' capacity to change, and engagement with services. Models such as

the research-based approach for assessing the risk of further maltreatment (Baynes, Jones and Bowyer, 2013), based on Jones and colleagues' systematic reviews of risk and protective factors where children are likely to suffer future harm (see Chapter Two), and Platt's (2012) model for assessing parents' engagement with services (Chapter Five) have been designed to facilitate such analysis, although they have not yet been fully piloted and evaluated in this country.

Gaps between parenting capability and children's needs

An analysis using a case conceptualisation based on the Assessment Framework will help practitioners to identify why there are gaps between parenting capability and children's needs. For instance, this model might guide practitioners into asking whether insufficient stimulation is having a negative impact on children's cognitive development, and whether this is related to factors within the family history or functioning (for example parents' learning disability or substance misuse). However, a more dynamic model would not only explore whether there are gaps in parents' capability, why they have occurred and whether the child's needs are being met, but would also ask whether parents have the capacity to change, and might take such assessments further forward. Harnett's (2007) model (see Chapter Three) can be understood as a further development of some of the theories that support the Assessment Framework; it builds on an initial crosssectional assessment by guiding practitioners through the process of working with parents to identify and set achievable goals, selecting and implementing interventions appropriate to achieving those goals and measuring progress. A dynamic conceptual model such as this should quide practitioners towards collecting information that will enable them to show what goals have been achieved so far and within what time scale, what services have been offered, how far parents have been motivated and engaged in taking up the services on offer and whether they have made a difference, and what is the likely timescale for parents' capability improving and progress being reliably sustained to the extent that children are adequately safeguarded.

Implications for day to day practice

Court requirements and implications for social work practice

It should be evident from the above that assessments required by the courts should not only be undertaken to inform proceedings, but should also be an integral part of day to day social work with parents whose children are on the edge of care. Such assessments are necessary to monitor parental progress towards agreed goals as well as to inform decisions about the safety of children living in families where there are serious child protection concerns, and also to inform child protection and looked after children plans, effective interventions for parents and children, plans for reunification from care, continuing social work support, and so on.

The literature review identified a number of messages concerning the knowledge and skills that social workers need to acquire in order to undertake such assessments and take appropriate action. Above all, they need to be able to observe and communicate with children and listen to their views. However, they also need to be able to recognise the indicators of substance misuse, alcohol abuse and domestic abuse. They need to be able to recognise parental learning disabilities and identify the symptoms of mental health problems and poor personality functioning. Where social workers recognise indicators, they need to assess matters further using evidence-based assessment tools or, where appropriate, refer the parent for specialist assessment and support. Social workers need to know when and how to refer to mental health professionals, and what interventions are available and might make a difference. They also need to know that harm to children arises from parents' functioning and behaviour, not their diagnosis; that psychiatrists often cannot provide a definitive prognosis; and that mental health assessments are not a substitute for parenting assessments. When parents have mental health problems or poor personality functioning, the social worker's role is to assess the extent to which observable symptoms are impacting on parents' ability to meet their children's needs.

The research evidence also indicates that social workers need to be aware that, in families where there are serious child protection issues and children are on the edge of care, parents are likely to experience overlapping problems that interact with one another. For instance, parents with mental health problems are more likely than others to be problem drinkers and/or misuse substances (Beckwith *et al.*, 1999; Woodcock and Sheppard, 2002); they may also be in a violent relationship (Finney, 2004) and have financial difficulties and housing problems (Jack and Gill, 2013).

Resources such as the NSPCC training pack on substance misuse for child care professionals (see

http://www.nspcc.org.uk/Inform/trainingandconsultancy/learningresources/seeingandhear ing_wda56195.html), guidance for professionals working with parents who misuse alcohol (Alcohol Concern, 2006), training materials on identifying neglect (Department for Education, 2012) and the Social Work, Alcohol and Drugs (SWALC) website (see http://www.swalcdrugs.com/) provide guidance to practitioners on recognising indicators of different problems and their possible impact on children's wellbeing.

Understanding the process of change

The literature review also provides useful material concerning the process of change. Research on overcoming alcohol or substance misuse has identified a complex process whereby individuals often deny the need to change their behaviour patterns before gradually becoming motivated to change and engage with services. Progress is also unlikely to follow a linear pathway, but is commonly marked by relapse before further engagement (Prochaska *et al.*,1992). This concept of progression through different stages of change in which decisions to engage and take action are finely balanced between perceptions of the advantages and disadvantages of altering the status quo.

may be helpful to social work practitioners in understanding the behaviour patterns of parents whose children are on the edge of care, and who may be struggling with more complex, inter-related problems than adults who are trying to overcome addiction *per se*.

Many parents, when advised of serious child protection concerns will, at least initially, deny that there is a problem or that their behaviour patterns have an impact on their children; others will outwardly comply with social workers' requirements, while inwardly remaining disengaged. Some parents with learning disabilities or specific mental health problems may not understand that certain behaviour patterns and adverse parenting practices can place their children at risk of significant harm. However resistance to change can also reflect internal factors such as fear of stigma, shame, ambivalence about the benefits of change and parents' lack of confidence about their capacity to overcome factors that place their children at risk. Much of the social worker's task involves supporting parents through the process of change — encouraging them to become motivated and engage with services and to persist in the face of discouragement and relapse. Supporting children through this process is, of course, a key element of this task.

Parents become motivated to change for a number of different reasons. Many of these are unclear, although it does appear that motivation to change begins when the perceived benefits start to outweigh the perceived costs. There is also evidence that some parents will experience events or circumstances that create a turning point in their lives and motivate them to overcome adverse behaviour patterns and improve their parenting. Turning points can include: a sudden shock, such as experiencing the death of a close relative or friend from alcohol misuse; hitting rock bottom; life events such as pregnancy and general maturation. Such turning points create windows of opportunity in which practitioners can engage with parents and introduce appropriate support.

Practitioners need to be aware that engagement is determined both by internal factors such as parents' self-confidence and expectations and external factors such as the effectiveness of the service and the availability of resources. Cancelled appointments or changes of staff can reduce parents' motivation to engage with a service. It may prove helpful to map out the different internal and external factors that promote or inhibit engagement in order to understand what can be improved to ensure that parents are more effectively engaged with services.

The process of change is, however, a lengthy business, and there is much evidence to indicate that services are often withdrawn too abruptly, with insufficient step down support following intensive interventions. Continuing support should be provided at least during the six months following an intensive intervention, when the likelihood of relapse is greatest. Thought should also be given to the type and duration of support services required by parents when children are returning home from care.

Relationships between social workers and parents

The research findings emphasise the significance of relationships between professionals and parents in facilitating and sustaining change. Many parents will not develop sufficient capacity to meet their children's needs without support. However a substantial body of research (e.g. Dumbrill, 2006; Forrester et al., 2012; Forrester and Harwin, 2011) suggests that the role of the social worker is, in itself, a major factor in the change process and that many parents may not be so much resistant to change as resistant to the involvement of social workers in their families. Parents' approach to the relationship with social workers will be coloured by their past experience and their current attachment styles, as well as by their awareness that the social worker's role is to assess and make judgments concerning their parenting capability. Social workers may find it difficult to resolve their twin roles of supporting and empowering parents and making judgments and taking authoritative action when children are likely to suffer significant harm. The relationship can therefore be fraught with unresolved tensions that practitioners need to appreciate and understand. The way in which social workers use, and are perceived as using, their power is a major factor in the success or failure of the relationship. The use of coercion, through the instigation of proceedings or threats to remove children can act as a powerful motivator for some parents, but it can also backfire and cause parents to become overwhelmed by their difficulties.

If social workers adopt a confrontational approach in communicating with parents and are perceived as using their power to penalise rather than to support them, they are likely to exacerbate parents' resistance to their involvement and the relationship may become focussed on the struggle for power rather than on the problems that need addressing (Dumbrill, 2006). On the other hand, if social workers can establish a strong working relationship with parents, that is characterised by honesty about what needs to change and why, sensitivity and a willingness to listen to parents' points of view, respectful uncertainty in the face of dissimulation and supportive use of power, they may be better able to help parents become motivated and engage with services.

A number of approaches may be useful in helping social workers develop such relationships. Firstly, social workers may be perceived as more sensitive and appreciative of parents' points of view if they develop motivational interviewing skills. This approach has been found to be an effective way of helping problem alcohol users overcome ambivalence about change and become engaged in treatment (Bowen and Gilchrist, 2004). Its emphasis on collaboration rather than confrontation and on eliciting and building on the service user's views about how change can be facilitated may reduce some of the elements in relationships with social workers that are known to contribute to resistance, and refocus interactions on the welfare of the child. However a formal evaluation of its effectiveness in child protection work has not yet been completed (see Forrester *et al.*, 2012).

Research on parents' experiences of social work involvement suggests that some are unaware of the extent of social work concerns about their children's welfare until care

proceedings are initiated (see Ward, Brown and Westlake, 2012); however this should happen less frequently as it becomes more common to send a letter before proceedings (see Department for Education, 2014a, pp.16-17). Openness and transparency, which are also fundamental to motivational interviewing techniques, are valued by parents, who appreciate social workers who are honest about their concerns and 'not afraid of breaking bad news' (Ward, Brown and Westlake, 2012, p.193). This is one reason why agreeing concrete goals and monitoring progress are considered to be an effective means of supporting parents through the process of change.

Introducing family group decision-making (FGDM) can also redress the imbalance of power in social worker/parent relationships by engaging the wider family in child protection processes. FGDM tends to be viewed positively by parents, can be effective for BMER families as it is respectful of their culture and language, and has been shown to increase the commitment and engagement of family members, including fathers and paternal relatives, in making plans for children's welfare (Barnsdale and Walker, 2007; Morris and Connolly, 2012). Concerns have, however, been raised that FGDM conferences can be controlled by dominant family members (or indeed professionals) who do not give a voice to the views of children and vulnerable adults such as victims of domestic abuse, and that both professionals and family members can show a lack of commitment to implementing the plans that have been agreed (Barnsdale and Walker, 2007; Cashmore and Kiely, 2000). There is also contradictory evidence concerning the effectiveness of FGDM in improving outcomes for children compared with other child protection services (Berzin, 2006; Edwards et al., 2007; Gunderson et al., 2003; Sundell and Vinnerljung, 2004; Titcomb and LeCroy, 2005). Some of the mixed research findings may relate to the complex needs of families referred to FGDM (see Sundell and Vinnerliung, 2004) or derive from differences in the ways in which FGDM is implemented. Concerns about variations in standards of FGDM in England are being addressed by the development of an accreditation framework (Haresnape and Brown, 2013).

The literature review has also found considerable evidence of the extent to which external factors such as poverty, unemployment, poor and overcrowded housing and lack of community cohesion can exacerbate parental problems and magnify the challenges faced by parents whose children are on the edge of care. The research also shows that parents appreciate it when social workers use their position of power to act as advocates on their behalf and help them to negotiate with other authority figures such as landlords. Dawe and Harnett (2007) found that too little attention is given to the practical assistance that social workers can provide. Focussing on practical as well as emotional support might be a further means of improving the relationship, as well as supporting very vulnerable families.

Relationships with other professionals

It is obvious that good working relationships with other professionals are also of key importance. Many parents who are concerned about the perceived stigma of social

services involvement will initially turn to other agencies for support. Only just over half of children who are subjects of serious case reviews are known to children's social care (Brandon *et al.*, 2009). Social workers need to have close working relationships with professionals from a range of agencies in order to establish an effective team around the child.

Where parents are trying to overcome complex and entrenched problems, social work support needs to be complemented by other, more specific interventions, delivered by professionals from a wide range of agencies – another reason for the development of good inter-agency relationships. Tensions between adult and children's services are common, particularly if one professional focusses exclusively on the needs of the parent and another on the needs of the child (Tompsett *et al.*, 2009; Ward, Brown and Westlake, 2012). Such tensions will not be resolved without open discussion.

In order to support parents in achieving the goals that have been agreed, social workers need to be aware of the services available in their area. Not all services are effective and some can be harmful. It is therefore important for social workers to know what has been shown to work (and what has not) and to be able to interpret the findings from evaluations. They also need to understand what services are appropriate for parents and children with different configurations of need. For instance, while parents whose children are at serious risk of abuse and neglect will need to learn better parenting skills, they are more likely to profit from programmes that have been specifically tailored to meet their needs than from those that have been developed for a population whose problems are less severe. Tailored parenting skills programmes may help parents develop greater selfefficacy and improved self-esteem, and this may have a knock on effect on their ability to address other problems such as alcohol misuse. However parents with more entrenched needs may also require additional, specialist services to help them address specific problems. An intensive, multifaceted approach that integrates a range of services tailored to meet the needs of the whole family may be most effective when parents are facing a complex web of entrenched problems, parent-child relationships are poor, and children's development is compromised by abuse and neglect.

Assessing parents' capacity to change within a child's timeframe

When there are significant child protection issues and children are on the edge of care, the fundamental question for both social workers and the courts is whether parents can make sufficient changes to ensure that children are adequately protected from harm within an appropriate timeframe. There is substantial evidence to indicate that abuse and neglect can have an adverse and enduring impact across the whole spectrum of children's development. There are factors within the child, the family and the environment which can mitigate and help children overcome the consequences of maltreatment; nevertheless, the longer children are exposed to abuse and neglect and the more severe the maltreatment, the harder it is to overcome the consequences. Therefore

professionals need to base their decisions on what is known about the timeframes for parental change compared with their knowledge of timeframes for childhood development.

No one can predict how long it will take for abusive or neglectful parents to develop sufficient capability to protect their children from harm; it is, however, evident that this is unlikely to happen overnight, that the process of change may be lengthy and that setbacks are common. The review of the literature found no evidence concerning how long it takes for parents whose children are on the edge of care to accept the need for change and begin to engage with services. Some do not appear to do so, or at least not within a child's timeframe. The research suggests that unequivocal cases are rare, but that where there are serious child protection concerns and parents do not acknowledge that a problem exists, they are unlikely to make sufficient changes to protect children from harm within an appropriate timeframe in families where the following factors are present: extreme domestic abuse where the perpetrator shows a pervasive pattern of disregard for and violation of the rights of others (Gondolf, 2002; Scott, 2004); there is both substance misuse and domestic abuse and violence in the home (Forrester and Harwin, 2008); children are not protected from perpetrators of sexual abuse; and/or where parents consciously and systematically cover up deliberate maltreatment (Brandon et al., 2008).

However where parents do accept that there is a problem and begin to engage with services, there may still be several months before significant and enduring progress is achieved. Intensive interventions themselves take several weeks to deliver. For instance, alcohol withdrawal programmes are typically of two to three weeks duration, and rehabilitation programmes usually last from three to six months (NICE, 2011a, p.33). Some evaluations have shown that adults with complex problems require longer interventions before changes are evident. Intensive, multifaceted interventions for parents and families with complex needs may take up to nine months to complete – MST-CAN is offered over a period of three sessions a week for six to nine months (Swenson *et al.*, 2010a).

Once a programme has been completed successfully, there may also be a vulnerable period before change is fully established – for instance, the highest rate of relapse for alcohol dependency occurs during the first three months post treatment (Raistrick *et al.*, 2006); perpetrators of domestic abuse are most likely to assault their partners again within the first six months of entering a programme (Gondolf, 2004). Chapter Seven has also shown that relapse is an integral part of the process of change, and it may take many years for the risk to diminish – five years to achieve stable recovery from alcohol misuse (Hibbert and Best, 2011); eight to ten years if the problem is misuse of heroin (Hser, Longshore and Anglin, 2007). Continued monitoring and reinforcement of progress may therefore also be necessary to support maintenance of change.

These timescales all point to the need for effective, proactive case management once abuse has been identified. This is particularly important where children's development is being compromised by parents' inability to meet their needs. Proactive case management needs to include early agreement about goals to be met and provision of effective services that will support parents in meeting them. What we know about likely timescales for overcoming destructive behaviour patterns and improving parenting capability makes it clear that much of this work will need to be undertaken *before* decisions are made to instigate proceedings. The requirement to complete court proceedings within 26 weeks does not allow time for delaying decisions about what may or may not be effective in helping parents to change, but that is not the fundamental issue. Given what is known about the impact of abuse and neglect on early childhood development, and the time it takes for parental change to become established, delayed assessment and service provision will not only compromise children's life chances, but also deny parents the support they need to overcome adverse behaviour patterns within an appropriate timeframe.

Of course parents will show progress before a programme is completed, but practitioners (and, where they are involved, the courts) will need to know how much progress has been made, and how far this is likely to be sustained before making decisions about when and whether children are safe from harm. A parent's slow progress or frequent setbacks will have a different impact depending on the age and vulnerability of the child, and these factors will also need to be taken into account, as well as the child's wishes. Where there is a question of reuniting children with birth parents from whom they have previously been separated, practitioners need to be aware that poorly managed reunification can undermine parents' progress (see for instance Carlson *et al.*, 2006); change needs to be firmly established before children return home, and parents will need continuing support following reunification.

Implications for continuing professional development

Research in this area is constantly producing new evidence concerning both the factors within the child, the family and the environment that place children at greater risk of significant harm, and those which protect them; the long-term adverse consequences of continuing exposure to abuse and neglect, and also the ways in which maltreatment can be prevented or its consequences mitigated. The empirical evidence is important because it not only shows what happens to children growing up in adverse circumstances, but it also helps us to understand why. Putting together the evidence from a wide range of disciplines, as we have done in this report, provides a stronger basis for understanding. Learning about the most up to date evidence on these issues should not only be a core element of pre-qualifying training, it should also be an integral element of continuing professional development for all practitioners in this area. The plethora of ongoing research is a major reason why even the most experienced practitioners should be encouraged to participate in post-qualifying training (Carpenter *et al.*, 2010).

Conclusion

The findings from this review of literature have numerous implications for general everyday practice as well as for assessments of children's needs, parenting capability and the analyses of the parenting gap required by the courts. Many of the issues it raises will already be familiar to social workers and other practitioners. The overview of this wide body of research should strengthen practitioners' understanding of parental capacity to change. It has brought together from a range of sources research messages concerning theoretical models that can underpin the process of assessment and analysis and empirical findings concerning the aetiology of abuse and neglect and the process by which adults overcome adverse behaviour patterns, with the aim of improving professional decision-making in situations where there are serious child protection concerns and children are on the edge of care.

Key Points from Chapter Eight

- The findings from this review should help social workers and children's guardians
 produce more robust and focussed assessments of parenting capability and
 parental capacity to change, and assist judges and other family justice
 professionals in evaluating the quality of assessment work presented to the courts.
- The types of assessment required by the courts are necessary to monitor parents'
 progress towards agreed goals as well as to inform decisions about the safety of
 children living in homes where there are serious child protection concerns. They
 should also be an integral part of social work practice when children are on the
 edge of care.
- Further resources and training materials are available to help practitioners improve their understanding of issues such as parental substance misuse or neglect and its impact on children.
- Concepts of progression through different stages of change and relapse, and the
 decisional balance that underlies motivation, engagement and action, may all be
 useful to practitioners in understanding the behaviour of parents whose children
 are on the edge of care.
- The research findings underline the significance of relationships between practitioners and parents in facilitating and sustaining change, and the key importance of relationships with other professionals in supporting parents and providing a team around the child.
- The process of change may be lengthy. Even after parents have become motivated to change and engage with services, intensive support programmes to address specific problems may last several weeks or months, and relapses are common within the first three to six months of completion. Much of this work needs to be undertaken before the decision is made to instigate proceedings.

- Delayed assessments and service provision will not only compromise children's development, but also deny parents the support they need to overcome adverse behaviour patterns within an appropriate timeframe.
- Research in the areas covered by this report is continually producing new findings
 of key importance for decision-making, and should therefore be an integral part of
 continuing professional development for all those who have professional
 responsibilities for safeguarding children on the edge of care.

Appendices

Appendix One: search terms

	Academic Press	AIP Journals and Conference Proceedings	BMJ Journals
--	----------------	---	--------------

	British Psychological Society	British Standards Online	CrossRef
--	-------------------------------	--------------------------	----------

Directory	y of Open Access Journal	s (DOAJ) Elsevier ScienceDirect	Emerald journals & ebooks
-----------	--------------------------	---------	--------------------------	---------------------------

Gale collections InfoSci journals	IngentaConnect
-----------------------------------	----------------

JSTOR	LexisNexis Academic Law Reviews	MEDLINE/PubMed
33101	LEXISINEXIS ACQUEITIC LAW INCHES	

MUUEIII HUIIIAIIIILES NESEAICII ASSOCIALIOII — OXIOIU OIIIVEISILV FIESS — FSVCAN I	rn Humanities Research Association Oxford University Press	PsycARTICLES
--	--	--------------

D DOOKO	D ODITIONEO	D 1 M 1 O 1
PsvcBOOKS	PsvcCRITIQUES	PubMed Central
Lavidoono	1 300011113000	i univieu Gentiai

Sage	SpringerLink	Taylor & Francis journals
Ouge	Opiniquient	rayior a riariois journais

Web of Science Wiley Online Library

Table 1: search terms

Search Items	Search Items	Search Items
Parent* AND change	Mother AND skill	Violence AND mother AND child protection
Mother AND change	Father AND skill	Violence AND father AND child welfare
Father AND change	Parent* AND ability	Violence AND father AND child maltreatment
Parent* AND assess*	Mother AND ability	Violence AND father AND child protection
Mother AND assess*	Father AND ability	Batter* AND parent* AND child welfare
Father AND assess*	Parent* AND train*	Batter* AND parent* AND child maltreatment
Parent* AND screen*	Mother AND train*	Batter* AND parent* AND child protection
Mother AND screen*	Father AND train*	Batter* AND mother AND child welfare
Father AND screen*	Parent* AND maltreat*	Batter* AND mother AND child maltreatment
Parent* AND tool	Mother AND maltreat*	Batter* AND mother AND child protection
Mother AND tool	Father AND maltreat*	Batter* AND father AND child welfare
Father AND tool	Change AND motivation	Batter* AND father AND child maltreatment
Parent* AND instrument	Change AND millingness	Batter* AND father AND child protection
Mother AND instrument	Change AND opportunity	Alcohol AND parent* AND child welfare
Father AND instrument	Change AND opportunity Change AND readiness	Alcohol AND parent* AND child maltreatment
Parent* AND program*	Change AND mechanism	Alcohol AND parent* AND child protection
		Alcohol AND mother AND child welfare
Mother AND program* Father AND program*	Change AND time	Alcohol AND mother AND child welfare Alcohol AND mother AND child maltreatment
1 0	Change AND relapse	
Parent* AND model	Change AND attrition	Alcohol AND mother AND child protection
Mother AND model	Change AND factors	Alcohol AND father AND child welfare
Father AND model	Behaviour AND change AND parent*	Alcohol AND father AND child maltreatment
Parent* AND therapy	Behaviour AND change AND mother	Alcohol AND father AND child protection
Mother AND therapy	Behaviour AND change AND father	Drugs AND parent* AND child welfare
Father AND therapy	Parent* AND resistance	Drugs AND parent* AND child maltreatment
Parent* AND intervention	Mother AND resistance	Drugs AND parent* AND child protection
Mother AND intervention	Father AND resistance	Drugs AND mother AND child welfare
Father AND intervention	Parent* AND engag*	Drugs AND mother AND child maltreatment
Parent* AND capacity	Mother AND engag*	Drugs AND mother AND child protection
Mother AND capacity	Father AND engag*	Drugs AND father AND child welfare
Father AND capacity	Substance AND parent* AND child welfare	Drugs AND father AND child maltreatment
Parent-child AND interaction	Substance AND parent* AND child maltreatment	Drugs AND father AND child protection
Mother-child AND interaction	Substance AND parent* AND child protection	Personality disorder AND parent* AND child welfare
Father-child AND interaction	Substance AND mother AND child welfare	Personality disorder AND parent* AND child maltreatment
Parent-child AND attachment	Substance AND mother AND child maltreatment	Personality disorder AND parent* AND child protection
Mother-child AND attachment	Substance AND mother AND child protection	Personality disorder AND mother AND child welfare
Father-child AND attachment	Substance AND father AND child welfare	Personality disorder AND mother AND child maltreatment
Parent-child AND relationship	Substance AND father AND child maltreatment	Personality disorder AND mother AND child protection
Mother-child AND relationship	Substance AND father AND child protection	Personality disorder AND father AND child welfare
Father-child AND relationship	Violence AND parent* AND child welfare	Personality disorder AND father AND child maltreatment
Parent* AND competence	Violence AND parent* AND child maltreatment	Personality disorder AND father AND child protection
Mother AND competence	Violence AND parent* AND child protection	Psychotic AND parent* AND child welfare
Father AND competence	Violence AND mother AND child welfare	Psychotic AND parent* AND child maltreatment
Parent* AND skill	Violence AND mother AND child maltreatment	Psychotic AND parent* AND child protection
Psychotic AND mother AND child welfare	Bipolar AND mother AND child maltreatment	Dissociative AND mother AND child protection
Psychotic AND mother AND child maltreatment	Bipolar AND mother AND child protection	Dissociative AND father AND child welfare

Search Items	Search Items	Search Items
Psychotic AND mother AND child protection	Bipolar AND father AND child welfare	Dissociative AND father AND child maltreatment
Psychotic AND father AND child welfare	Bipolar AND father AND child maltreatment	Dissociative AND father AND child protection
Psychotic AND father AND child maltreatment	Bipolar AND father AND child protection	Neurocognitive AND parent* AND child welfare
Psychotic AND father AND child protection	Emotional AND parent* AND child welfare	Neurocognitive AND parent* AND child maltreatment
Delusion AND parent* AND child welfare	Emotional AND parent* AND child maltreatment	Neurocognitive AND parent* AND child protection
Delusion AND parent* AND child maltreatment	Emotional AND parent* AND child protection	Neurocognitive AND mother AND child welfare
Delusion AND parent* AND child protection	Emotional AND mother AND child welfare	Neurocognitive AND mother AND child maltreatment
Delusion AND mother AND child welfare	Emotional AND mother AND child maltreatment	Neurocognitive AND mother AND child protection
Delusion AND mother AND child maltreatment	Emotional AND mother AND child protection	Neurocognitive AND father AND child welfare
Delusion AND mother AND child protection	Emotional AND father AND child welfare	Neurocognitive AND father AND child maltreatment
Delusion AND father AND child welfare	Emotional AND father AND child maltreatment	Neurocognitive AND father AND child protection
Delusion AND father AND child maltreatment	Emotional AND father AND child protection	Mental retardation AND parent* AND child welfare
Delusion AND father AND child protection	Mani* AND parent* AND child welfare	Mental retardation AND parent* AND child maltreatment
Neurodevelopmental AND parent* AND child welfare	Mani* AND parent* AND child maltreatment	Mental retardation AND parent* AND child protection
Neurodevelopmental AND parent* AND child maltreatment	Mani* AND parent* AND child protection	Mental retardation AND mother AND child welfare
Neurodevelopmental AND parent* AND child protection	Mani* AND mother AND child welfare	Mental retardation AND mother AND child maltreatment
Neurodevelopmental AND mother AND child welfare	Mani* AND mother AND child maltreatment	Mental retardation AND mother AND child protection
Neurodevelopmental AND mother AND child maltreatment	Mani* AND mother AND child protection	Mental retardation AND father AND child welfare
Neurodevelopmental AND mother AND child protection	Mani* AND father AND child welfare	Mental retardation AND father AND child maltreatment
Neurodevelopmental AND father AND child welfare	Mani* AND father AND child maltreatment	Mental retardation AND father AND child protection
Neurodevelopmental AND father AND child maltreatment	Mani* AND father AND child protection	Development* AND parent* AND child welfare
Neurodevelopmental AND father AND child protection	Mood AND parent* AND child welfare	Development*AND parent* AND child maltreatment
Mental health AND parent* AND child welfare	Mood AND parent* AND child maltreatment	Development*AND parent* AND child protection
Mental health AND parent* AND child maltreatment	Mood AND parent* AND child protection	Development*AND mother AND child welfare
Mental health AND parent* AND child protection	Mood AND mother AND child welfare	Development*AND mother AND child maltreatment
Mental health AND mother AND child welfare	Mood AND mother AND child maltreatment	Development*AND mother AND child protection
Mental health AND mother AND child maltreatment	Mood AND mother AND child protection	Development*AND father AND child welfare
Mental health AND mother AND child protection	Mood AND father AND child welfare	Development*AND father AND child maltreatment
Mental health AND father AND child welfare	Mood AND father AND child maltreatment	Development*AND father AND child protection
Mental health AND father AND child maltreatment	Mood AND father AND child protection	Schizo* AND parent* AND child welfare
Mental health AND father AND child protection	Anxiety AND parent* AND child welfare	Schizo* AND parent* AND child maltreatment
Depress* AND parent* AND child welfare	Anxiety AND parent* AND child maltreatment	Schizo* AND parent* AND child protection
Depress* AND parent* AND child maltreatment	Anxiety AND parent* AND child protection	Schizo* AND mother AND child welfare
Depress* AND parent* AND child protection	Anxiety AND mother AND child welfare	Schizo* AND mother AND child maltreatment
Depress* AND mother AND child welfare	Anxiety AND mother AND child maltreatment	Schizo* AND mother AND child protection
Depress* AND mother AND child maltreatment	Anxiety AND mother AND child protection	Schizo* AND father AND child welfare
Depress* AND mother AND child protection	Anxiety AND father AND child welfare	Schizo* AND father AND child maltreatment
Depress* AND father AND child welfare	Anxiety AND father AND child maltreatment	Schizo* AND father AND child protection
Depress* AND father AND child maltreatment	Anxiety AND father AND child protection	Somat* AND parent* AND child welfare
Depress* AND father AND child protection	Dissociative AND parent* AND child welfare	Somat* AND parent* AND child maltreatment
Bipolar AND parent* AND child welfare	Dissociative AND parent* AND child maltreatment	Somat* AND parent* AND child protection
Bipolar AND parent* AND child maltreatment	Dissociative AND parent* AND child protection	Somat* AND mother AND child welfare
Bipolar AND parent* AND child protection	Dissociative AND mother AND child welfare	Somat* AND mother AND child maltreatment
Bipolar AND mother AND child welfare	Dissociative AND mother AND child maltreatment	Somat* AND mother AND child protection
Somat* AND father AND child welfare	Mother AND sustain*	·
Somat* AND father AND child maltreatment	Father AND sustain*	
Somat* AND father AND child protection	Parent* AND maintain	

Search Items	Search Items	Search Items
Addictive AND parent* AND child welfare	Mother AND maintain	Social Relie
Addictive AND parent* AND child maltreatment	Father AND maintain	
Addictive AND parent* AND child protection	Parent* AND effect* AND treatment	
Addictive AND mother AND child welfare	Mother AND effect* AND treatment	
Addictive AND mother AND child maltreatment	Father AND effect* AND treatment	
Addictive AND mother AND child protection	Support AND child welfare	
Addictive AND father AND child welfare	Support AND child protection	
Addictive AND father AND child maltreatment	Change AND child welfare	
Addictive AND father AND child protection	Change AND child protection	
Behaviour AND parent* AND child welfare	Parent* AND impact	
Behaviour AND parent* AND child maltreatment	Parenting AND capacity	
Behaviour AND parent* AND child protection	Parent* AND treatment AND maintenance	
Behaviour AND mother AND child welfare	Parent* AND drug AND maintenance	
Behaviour AND mother AND child maltreatment	Parent* AND drug AND recovery	
Behaviour AND mother AND child protection	Parent* AND drug AND relapse	
Behaviour AND father AND child welfare	Parent* AND alcohol AND maintenance	
Behaviour AND father AND child maltreatment	Parent* AND alcohol AND recovery	
Behaviour AND father AND child protection	Parent* AND alcohol AND relapse	
Sexual abuse AND parent* AND child welfare	Parent* AND violence AND maintenance	
Sexual abuse AND parent* AND child maltreatment	Parent* AND violence AND recovery	
Sexual abuse AND parent* AND child protection	Parent* AND violence AND relapse	
Sexual abuse AND mother AND child welfare	Parent* AND domestic abuse AND treatment	
Sexual abuse AND mother AND child maltreatment	Parent* AND domestic abuse AND reoffend	
Sexual abuse AND mother AND child protection	Parent* AND substance AND recovery	
Sexual abuse AND father AND child welfare	Parent* AND substance AND relapse	
Sexual abuse AND father AND child maltreatment	Parent* AND substance AND maintenance	
Sexual abuse AND father AND child protection	Parent* AND long-term impact AND intervention	
Parent* AND recovery	Systematic review AND parent*	
Mother AND recovery	Meta analysis AND parent*	
Father AND recovery		
Parent* AND maintenance		
Mother AND maintenance		
Father AND maintenance		
Parent* AND relapse		
Mother AND relapse		
Father AND relapse		
Parent* AND abstinence		
Mother AND abstinence		
Father AND abstinence		
Parent* AND sustain*		

Appendix Two: scientific advisory group and steering group members

Scientific advisory group

Jane Barlow: Professor of Public Health in the Early Years, University of Warwick

Paul Harnett: Senior Lecturer in Clinical Psychology, University of Queensland

Judith Harwin: Director of Social Work Division, Director of Child Focused Research

Centre, Brunel University

Judith Masson: Professor of Socio-Legal Studies, University of Bristol

Duncan McLean: Consultant Adult Psychiatrist, Anna Freud Centre

Dendy Platt: Senior Lecturer in Social Work (Child Care), University of Bristol

Nicky Stanley: Professor of Social Work, University of Central Lancashire

Steering group members

Chris Cuthbert: Head of Strategy and Development for Under-Ones, NSPCC

Jonathan Dickens: Senior Lecturer, University of East Anglia

Judith Freedman: Consultant Psychiatrist, Head of the Consortium of Expert Witnesses to the

Family Courts

Liz Gillett: Consultant Clinical Psychologist, Director of the Phoenix Psychology Group

Bridget Lindley: Deputy Chief Executive and Legal Adviser, Family Rights Group

Brendah Malahleka: Service Manager, Business Strategy, Fostering Placement and

Procurement, Lewisham Borough Council

Her Honour Judge Lesley Newton: Circuit Judge Manchester

Helen Pustam: Head of Senior Legal and Policy Adviser Team, Magistrates' Training

Division, Judicial College

Steve Walker: Deputy Director of Children's Services, Safeguarding, Specialist and

Targeted, Leeds City Council

References

Accardo, P.J. and Whitman, B.Y. (1990) 'Children of mentally retarded parents.' *American Journal of Disabled Children 144*, 1, 69-70.

Aggleton, P., O'Reilly, K., Slutkin, G. and Davies, P. (1994) 'Risking everything? Risk behavior, behavior change, and AIDS.' *Science 265*, 5170, 341-345.

Al, C.M., Stams, G.J., Bek, M.S., Damen, E.M., Asscher, J.J. and Van der Laan, P.H. (2012) 'A meta-analysis of intensive family preservation programs: placement prevention and improvement of family functioning.' *Children and Youth Services Review 34*, 1472 – 1479.

Alcohol Concern (2006) *Understanding Alcohol Issues for Professionals Working with Parents*. London: Alcohol Concern.

Aldridge, J. and Becker, S. (2003) *Children Caring for Parents with Mental Illness:Perspectives of young carers, parents and professionals.* Bristol: The Policy Press.

Alhusen, J.L., Hayat, M.J. and Gross, D. (2013) 'A longitudinal study of maternal attachment and infant developmental outcomes.' *Archives of Women's Mental Health* 1-10.

Altman, J.C. (2008) 'A study of engagement in neighbourhood-based child welfare services.' *Research on Social Work Practice 18*, 6, 555-564.

Anders, S.L. and Tucker, J.S. (2000) 'Adult attachment style, interpersonal communication competence, and social support.' *Personal Relationships 7*, 379-389.

Anderson, C. J. (2003) 'The psychology of doing nothing: Forms of decision avoidance result from reason and emotion.' *Psychological Bulletin 129, 1,* 139-166.

Anderson, I. and Christian, J. (2003) 'Causes of homelessness in the UK: A dynamic analysis.' *Journal of Community and Applied Social Psychology 13*, 105–118.

Arad-Davidson, B. and Benbenishty, R. (2008) 'The role of workers' attitudes and parent and child wishes in child protection workers' assessments and recommendation regarding removal and reunification.' *Children and Youth Services Review 30*, 107-121.

Armstrong, T. A. (2003) 'The effect of moral reconation therapy on the recidivism of youthful offenders: a randomized experiment.' *Criminal Justice and Behaviour* 30, 668–687.

Autti-Rämö, I. (2002) 'Foetal alcohol syndrome - a multifaceted condition.' *Developmental Medicine & Child Neurology 44*, 141–144.

Baker, B.L., McIntyre, L.L., Blacher, J., Crnic, K., Edelbrock, C. and Low, C. (2003) 'Preschool children with and without developmental delay: behaviour problems and parenting stress over time.' *Journal of Intellectual Disability Research* 47, 4/5, 217-230.

Barber, J. and Delfabbro, P. (2005) 'Children's adjustment to long-term foster care.' *Children and Youth Services Review 27*, 3, 329–340.

Barlow, J., Davis, H., Mcintosh, E., Jarrett, P., Mockford, C. and Stewart-Brown, S. (2007) 'Role of home visiting in improving parenting and health in families at risk of abuse and neglect: result of a multicentre randomised controlled trial and economic evaluation.' *Archives of Disease in Childhood* 92, 3, 229-233.

Barlow. J., Davis, H., McIntiosh, E., Jarrett, P., Mockford, C. and Stewart-Brown, S. (2008) *The Oxfordshire Home Visiting Study: Three Year Follow-Up.* Warwick: University of Warwick.

Barlow, J., Fisher, J.D. and Jones, D. (2012) *Systematic Review of Models of Analysing Significant Harm.* London: Department for Education.

Barlow, J., Johnston, I., Kendrick, D., Polnay, L. and Stewart-Brown, S. (2008) 'Individual and group-based parenting programmes for the treatment of physical child abuse and neglect (Review).' *The Cochrane Library*, Issue 4.

Barlow, J. and Scott, J. (2010) *Safeguarding in the 21st Century – Where to Now?* Dartington: Research in Practice.

Barlow, J. and Schrader MacMillan, A. (2010) *Safeguarding Children from Emotional Maltreatment: What Works.* London: Jessica Kingsley Publishers.

Barlow, J., Schrader McMillan, A., Kirkpatrick, S., Ghate, D., Barnes, J. and Smith, M. (2010) 'Health-led interventions in the early years to enhance infant and maternal mental health: a review of reviews'. *Child and Adolescent Mental Health* 15, 4, 178-185.

Barlow, J., Simkiss, D. and Stewart-Brown, S. (2006) 'Interventions to prevent or ameliorate child physical abuse and neglect: findings from a systematic review.' *Journal of Children's Services 1, 3, 6-28.*

Barnard, M. (2007) Drug Addiction and Families. London: Jessica Kingsley Publishers.

Barnard, M. and Barlow, J. (2003) 'Discovering parental drug dependence: silence and disclosure.' *Children and Society 17*, 1, 45–56.

Barnard, M. and McKeganey, N. (2004) 'The impact of parental problem drug use on children: what is the problem and what can be done to help?' *Addiction 99*, 552–559.

Barnsdale, L. and Walker, M. (2007) *Examining the Use and Impact of Family Group Conferencing*. Edinburgh: Scottish Executive.

Barth, R. (2009) 'Preventing child abuse and neglect with parent training: evidence and opportunities.' *The Future of Children 19*, 2, 95-115.

Bateman, A. and Fonagy, P. (2010) 'Mentalisation treatment for borderline personality disorder.' *World Psychiatry* 9, 11-15.

Baynes, P., Jones, D., Bowyer, S. (2013) Assessing Risk of Further Maltreatment: a research-based approach. Available at: www.rip.org.uk/riskassessmenttool last accessed 06.02.14.

Beckwith, L., Howard, J., Espinosa, M. and Tyler, R. (1999) 'Psychopathology,mother-child interaction, and infant development: Substance-abusing mothers and their offspring.' *Development and Psychopathology 11*, 4, 715–725.

Benoit, D., Parker, K.C.H. and Zeanah, C.H. (1997) 'Mothers' representations of their infants assessed prenatally: stability and association with infants' attachment classifications.' *Journal of Child Psychology and Psychiatry and Allied Disciplines 38*, 307–313.

Bentovim, A. and Elliot, J. (2014) 'Hope for children and families – targeting abusive parenting and the associated impairment of children.' *Journal of Clinical Child and Adolescent Psychology 43*, 2, 270-285.

Berzin, S.C. (2006) 'Using sibling data to understand the impact of family group decision-making on child welfare outcomes.' *Child and Youth Services Review 28*, 1449-14.

Bessant, J. (2003) 'Pregnancy in a brotherhood bin: housing and drug-treatment options for pregnant young women.' *Australian Social Work 56*, 3, 234-246.

Bhardwaj, A. (2001) 'Growing up young, Asian and female in Britain: a report on self-harm and suicide.' *Feminist Review 68*, 52-67.

Biehal, N., Ellison, S., Baker, C. and Sinclair, I. (2010) *Belonging and Permanence: Outcomes in Long-term Foster Care and Adoption.* London: BAAF.

Biringen, Z. (2000) 'Emotional availability: conceptulization and research findings.' *American Journal of Orthopsychiatry* 70, 1, 104-114.

Blair, C. (2002) 'School readiness: integrating cognition and emotion in a neurobiological conceptualization of children's functioning at school entry.' *American Psychologist 57*, 2,111-127.

Bolger, K.E. and Patterson, C.J. (2001) 'Developmental patheays from child maltreatment to peer rejection'. *Child Development 72*, 2, 549-568.

Booth, T. and Booth, W. (1993) 'Parenting with learning difficulties: lessons for practitioners.' *British Journal of Social Work 23*, 5, 459-480.

Booth, T. and Booth, W. (2004) 'A family at risk: multiple perspectives on parenting and child protection.' *British Journal of Learning Disabilities* 32, 1, 9-15.

Bowen, E. and Gilchrist, E. (2004) 'Do court- and self-referred domestic violence offenders share the same characteristics? A preliminary comparison of motivation to change, locus of control and anger.' *Legal and Criminological Psychology* 9, 2, 279-294.

Bradley, A.R. and Wood, J.M. (1996) 'How do children tell? 'The disclosure process in child sexual abuse.' *Child Abuse and Neglect 20*, 9, 881-891.

Brandon, M., Bailey, S., Belderson, P., Gardner, R., Sidebotham, P., Dodsworth, J., Warren, C. and Black, J. (2009) *Understanding Serious Case Reviews and their Impact: A Biennial Analysis of Serious Case Reviews 2005-7*. Research Report DCSF-RR129. London: Department for Children Schools and Families.

Brandon, M., Bailey, S., Belderson, P. and Larsson, B. (2013) *Neglect and Serious Case Reviews*. London: NSPCC.

Brandon, M., Belderson, P., Warren, C., Howe, D., Gardner, R., Dodsworth, J. and Black, J. (2008) 'Analysing Child Deaths and Serious Injury Through Abuse or Neglect; What Can We Learn. A Biennial Analysis of Serious Case Reviews 2003–2005.' Research Report DCSF RR023. London: Department for Children, Schools and Families.

Brandon, M., Sidebotham, P., Bailey, S., Belderson, P., Hawley, C., Ellis, C. and Megson, M. (2012) *New Learning from Serious Case Reviews: A two year report for 2009-2011.*Research Report DFE-RR226. London: Department for Education.

Brandon, M., Sidebotham, P., Ellis, C., Bailey, S. and Belderson, P. (2011) *Child and Family Practitioners' Understanding of Child Development: Lessons Learnt from a Small Sample of Serious Case Reviews.* Research Report DFE-RR110. London: Department for Education.

Brewer, M., Browne, J. and Joyce, R. (2011) *Child and Working Age Poverty from 2010-2020.* London: Institute for Fiscal Studies.

Bridle, C., Riemsma, R.P., Pattenden, J., Soeden, A.J., Watt, I.S. and Walker, A. (2005) 'Systematic review of the effectiveness of health behavior interventions based on the trans-theoretical model.' *Psychology and Health 20*, 3, 283-301.

Brophy, J., Jhutti-Johal, J. and Owen, C. (2003) 'Assessing and documenting child ill-treatment in minority ethnic households.' *Family Law 33*, 756–764.

Brown, L. (2007) 'The adoption and implementation of a service innovation in a social work setting – a case study of family group conferencing in the UK.' Social Policy and Society 6, 3, 321-332.

Brown, L., Callahan, M., Strega, S., Walmsley, C. and Dominelli, L. (2009) 'Manufacturing ghost fathers: the paradox of father presence and absence in child welfare.' *Child and Family Social Work 14*, 25–34.

Brown, R., Hyde-Dryden, G., Thomas, C. and Ward, H. (forthcoming) 'Infants suffering, or likely to suffer significant harm: a prospective longitudinal study – extension to age 8.' Loughborough: Centre for Child and Family Research, Loughborough University.

Brown, R. and Ward, H. (2012) *Decision-making within a Child's Timeframe: An Overview of Current Research Evidence for Family Justice Professionals Concerning Child Development and the Impact of Maltreatment.* Working Paper 16. London: Childhood Wellbeing Research Centre.

Buckley, H., Holt, S. and Whelan, S. (2007) 'Listen to me! Children's experiences of domestic violence.' *Child Abuse Review 16*, 296-310.

Bullock, R., Little, M. and Milham, S. (1993) *Going Home: The Return of Children Separated from their Birth Families*. Aldershot: Dartmouth Publishing Co Ltd.

Bullock, K., Sarre, S., Tarling, R. and Wilkinson, M. (2010) *The Delivery of Domestic Abuse Programmes: An implementation study of the delivery of domestic abuse programmes in probation areas and her Majesty's Prison Service.* Ministry of Justice Research series 15/10. London: Ministry of Justice.

Burford, G., Connolly, M., Morris, K. and Pennell, J. (2009) *Family Involvement Strategies: An Annotated Bibliography*. American Humane Association, New York, USA. Available at: http://www.americanhumane.org/children/programs/family-group-decision-making/bibliographies/research-and-evaluation/ last accessed February 2014.

Burton, S., Regan, L. and Kelly, L. (1998) *Supporting Women and Challenging Men:* Lessons from the Domestic Violence Intervention Project. Bristol: Policy Press.

Cadsky, O., Hanson, R. K., Crawford, M. and Lalonde, C. (1996) 'Attrition from a male batterer treatment programme: client–treatment congruence and lifestyle instability.' *Violence and Victims 11*, 1, 51–64.

Cafcass/ADCS (undated) *Draft good practice guidance for social work practised in the family courts.* Available at:

https://www.cafcass.gov.uk/media/126324/good_practice_guidance_for_social_work_practised_in_the_family_courtsv5.pdf

Cannella, B.L. (2005) 'Maternal fetal attachment: an integrative review.' *Journal of Advanced Nursing 50*, 60-68.

Carey, T. A., Carey, M., Stalker, K., Mullan, R. J., Murray, L. K. and Spratt, M. B. (2007) 'Psychological change from the inside looking out: a qualitative investigation.' *Counselling and Psychology Research* 7, 3, 178–187.

Carlson, B.E., Matto, H., Smith, C.A. and Eversman, M. (2006) 'A pilot study of reunification following drug abuse treatment: Recovering the mother role.' *Journal of Drug Issues 36*, 877 – 902.

Carpenter, J., Hackett, S., Patsios, D. and Szilassy, E. (2010) *Outcomes of Inter-Agency Training to Safeguard Children. Research Report DCSF-RR209*. London: DCSF

Carpenter, J., Webb, C., Bostock, L. and Coomber, C. (2012) *Effective Supervision in Social Work and Social Care*. Research Brief 43. London: Social Care Institute for Excellence.

Casanueva, C., Martin, S., Runyan, D., Barth, R. and Bradley, R. (2008) 'Parenting services for mothers involved with child protective services: do they change maternal parenting and spanking behaviours with young children?' *Children and Youth Services Review 30*, 8, 861–878.

Casey, P. and Rottman, D.B. (2000) 'Therapeutic jurisprudence in the courts.' *Behavioral Sciences and the Law 18*, 4, 445-457.

Cashmore, J. and Kiely, P. (2000) 'Implementing and evaluating Family Group Conferences: The New South Wales Experience.' pp. 242-252. In G. Burford and J. Hudson (Eds.) *Family Group Conferencing: New Directions in Community-Centred Child and Family Practice*. New York: Aldine de Gruyter.

Chaffin, M., Bonner, B. and Hill, R. (2001) 'Family preservation and family support programs: child maltreatment outcomes across client risk levels and program types.' *Child Abuse and Neglect 25,* 10, 1269-1289.

Chaffin, M., Kelleher, K. and Hollenberg, J. (1996) 'Onset of physical abuse and neglect: psychiatric, substance abuse and social risk factors from prospective community data.' *Child Abuse and Neglect 20*, 191-203.

Charney, D.A., Zikos, E. and Gill, K.J. (2010) 'Early recovery from alcohol dependence: Factors that promote or impede abstinence.' *Journal of Substance Abuse Treatment 38* (2010) 42–50

Cleaver, H. and Freeman, P. (1995) *Parental Perspectives in Cases of Suspected Child Abuse*. London: HMSO.

Cleaver, H. and Nicholson, D. (2007) *Parental Learning Disability and Children's Needs:* Family Experiences and Effective Practice. London: Jessica Kingsley Publishers.

Cleaver, H., Nicholson, D., Tarr, S. and Cleaver, D. (2007) *Child Protection, Domestic Violence and Parental Substance Misuse: Family Experiences and Effective Practice*. London: Jessica Kingsley Publishers.

Cleaver, H., Unell. I. and Aldgate, J. (2011) *Children's Needs – Parenting Capacity. Child Abuse: Parental Mental Illness, Learning Disability, Substance Misuse and Domestic Violence* (2nd Edition). London: The Stationery Office.

Cluss, P., Chang, J., Hawker, L., Scholle, S., Dado, D., Buranosky, R. and Goldstrohm, S. (2006) 'The process of change for victims of intimate partner violence: support for a psychosocial readiness model.' *Women's Health Issues* 16, 5, 262–74.

Cohen, J. (1992) 'A power primer.' Quantitative Methods in Psychology 112,1,155-199.

Cohen, J. (1969) Statistical Power Analysis for the Relational Sciences. New York: Academic Press.

Collins, C.C., Grella, C.E. and Hser, Y. (2003) 'Effects of gender and level of parental involvement among parents in drug treatment.' *The American Journal of Drug and Alcohol Abuse* 29, 2, 237–261.

Collins, S.E., Malone, D.K. and Larimer, M.E. (2012) 'Motivation to change and treatment attendance as predictors of alcohol-use outcomes among project-based Housing First residents.' *Addictive Behaviors* 37, 8, 931-939.

Connell-Carrick, K. (2003) 'A critical review of the empirical literature: identifying correlates of child neglect.' *Child and Adolescent Social Work Journal* 20, 389-425.

Copello, A., Orford, J., Hodgson, R., Tober, G. and Barrett, C. On behalf of the UKATT Research Team (2002) 'Social behaviour and network therapy: Basic principles and early experiences.' *Addictive Behaviors* 27, 345–366.

Copello, A., Templeton, L., Orford, J., Velleman, R., Patel, A., Moore, L., MacLeod, J. and Godfrey, C. (2009) 'The relative efficacy of two levels of a primary care intervention for family members affected by the addiction problem of a close relative: a randomized trial.' *Addiction 104*, 1, 49-58.

Copello, A., Williamson, E., Orford, J. and Day, E. (2006) 'Implementing and evaluating social behaviour and network therapy in drug treatment practice in the UK: A feasibility study.' *Addictive Behaviors* 31, 5, 802–810.

Cordero, A. (2004) When family reunification works: Data-mining foster care records.' Families in Society 85,4, 571–580.

Covell, K. and Howe, B.R. (2009) *Children, Families and Violence: Challenges for Children's Rights.* London: Jessica Kingsley Publishers, 288.

Coventry LSCB (2013) Final Overview Report of Serious Case Review re Daniel Pelka. Coventry: Coventry LSCB.

Crampton, D. (2007) 'Research review: family group decision-making: a promising practice in need of more programme theory and research.' *Child and Family Social Work* 12, 2, 202-209.

Cummings, A.M. Gallop, R.J. and Greenfield, S.F. (2010) 'Self-efficacy and substance use outcomes for women in single-gender *versus* mixed-gender group treatment.' *Journal of Groups in Addiction and Recovery 5*,1, 4-16.

Currie, J. and Widom, C.S. (2010) 'Long-term consequences of child abuse and neglect on adult economic well-being.' *Child Maltreatment 15*, 2, 111-120.

Daly, J. and Pelowski, S. (2000) 'Dropout from groups for men who batter: a review of studies with implications for research and practice.' *Violence and Victims 15*, 2, 137-160.

Dalzell, R. and Sawyer, E. (2007) *Putting Analysis into Assessment: Undertaking Assessments of Need - A Toolkit for Practitioners*. London: National Children's Bureau.

Damashek, A., Doughty, D., Ware, L. and Silovsky, J. (2011) 'Predictors of client engagement and attrition in home-based child maltreatment prevention services.' *Child Maltreatment* 16. 1. 9-20.

Dagenais, C., Begin, J., Bouchard, C. and Fortin, D. (2004) 'Impact of intensive family support programs (IFPS): A synthesis of evaluation studies.' *Children and Youth Services Review 26*, 249–263.

Daniel, B., Taylor, J. and Scott, J. (2011) *Noticing and Helping the Neglected Child: Evidence-Based Practice for Assessment and Intervention*. London: Jessica Kingsley Publishers.

Danziger, S., Levav, J. and Avnaim-Pesso, L. (2011) 'Extraneous factors in judicial decisions.' *Proceedings of the National Academy of Sciences 108,17,* 6889-6892.

Darbyshire, L.V. and Kroese, B.S. (2012) 'Psychological well-being and social support for parents with intellectual disabilities: risk factors and interventions.' *Journal of Policy and Practice in Intellectual Disabilities* 9, 1, 40–52.

David, M. and Sutton, C.D. (2011) *Social Research: An Introduction*. (2nd Edition) London: Sage.

Davies, C. and Ward, H. (2012) Safeguarding Children Across Services: Messages from Research. London: Jessica Kingsley Publishers.

Davis, H., Dusoir, T., Papadopoulou, K., Dimitrakaki, C. *et al.* (2005) 'Child and family outcomes of the European Early Promotion Project.' *International Journal of Mental Health Promotion* 7, 1, 63-81.

Dawe, S. and Harnett, P. H. (2007) 'Improving family functioning in methadone maintained families: results from a randomised controlled trial.' *Journal of Substance Abuse Treatment 32*, 381-390.

Dawes, R. M., Faust, D. and Meehl, P. E. (1989) 'Clinical *versus* actuarial judgment.' *Science 243*, 4899, 1668–1674.

Dawson, K. and Berry, M. (2002) 'Engaging families in child welfare services: an evidence-based approach to best practice.' *Child Welfare 81*, 2, 293–317.

Day, C., Carey, M. and Surgenor, T. (2006) 'Children's key concerns: Piloting a qualitative approach to understanding their experience of mental health care.' *Clinical Child Psychology and Psychiatry 11*, 1, 139-155.

Day, E., Copello, A., Seddon, J.L., Christie, M., Bamber, D., Powell, C., George, S., Ball, A., Frew, E. and Freemantle, N. (2013) 'Pilot study of a social network intervention for heroin users in opiate substitution treatment: study protocol for a randomized controlled trial.' *Trials* 14, 1, 264.

De Bellis, M.D. (2005) 'The psychobiology of neglect.' Child Maltreatment 10, 2, 150-172.

De Boer, C. and Coady, N. (2007) 'Good helping relationships in child welfare: learning from stories of success.' *Child and Family Social Work 12*, 1, 32-42.

DeGarmo, D.S., Patterson, G.R. and Forgatch, M.S. (2004) 'How do outcomes in a specified parent training intervention maintain or wane over time?' *Prevention Science 5*, 2, 73-89.

DeHart, D. D., Kennerly, R. J., Burke, L. K. and Follingstadt, D. R. (1999) 'Predictors of attrition in a treatment programme for battering men.' *Journal of Family Violence 14*, 1, 19–34.

DePanfilis, D. and Girvin, H. (2005) 'Investigating child maltreatment in out-of-home care: Barriers to good decision-making.' *Children and Youth Services Review 27*, 353-374.

Department for Education (2011) *The Munro Review of Child Protection: A Child Centred System. Final Report.* London: Department for Education.

Department for Education (2012) *Training resources on childhood neglect*. Available at: http://education.gov.uk/childrenandyoungpeople/safeguardingchildren/childhoodneglect/b http://education.gov.uk/childrenandyoungpeople/safeguardingchildren/childhoodneglect/b http://education.gov.uk/childrenandyoungpeople/safeguardingchildren/childhoodneglect/b http://education.gov.uk/childrenandyoungpeople/safeguardingchildren/childhoodneglect/b http://education.gov.uk/childrenandyoungpeople/safeguardingchildren/childhoodneglect/b http://education.gov.uk/childhood-neglect/b http://education.gov.uk/childhood-neglect/b http://education.gov.uk/childhood-neglect/b http://education.gov.uk/childhood-neglect/b http://education.gov.uk/childhood.neglect/b http://education.gov.uk/childhood.neglect/b http://education.gov.uk/childhood.neglect/b http://education.gov.uk/childhood.neglect/b http://education.gov.uk/childhood.neglect/b http://education.gov.uk/childhood.neglect/b</

Department for Education (2013a) *Improving permanence and stability for vulnerable children and families: implementing evidence-based interventions and expanding the foster care workforce.* Available at:

http://www.education.gov.uk/childrenandyoungpeople/families/childrenincare/news/a0075 693/prospectus

Department for Education (2013b) *Children looked after in England (including adoption and care leavers) year ending 31 March 2013*. London: Department for Education. https://www.gov.uk/government/uploads/system/uploads/attachment_data/file/244872/SF
R36_2013.pdf accessed Jan 2014

<u>Department for Education (2014a)</u> <u>Statutory Guidance on Court Orders and Preproceedings for Local Authorities.</u> <u>London: Department for Education. Available at:</u>
https://www.gov.uk/government/uploads/system/uploads/attachment_data/file/281452/St
atutory Guidance on court orders and pre-proceedings 18.2.pdf

Department for Education (2014b) *Characteristics of Children in Need in England*. London: Department for Education.

Department of Health (1989) Introduction to the Children Act 1989. London. HMSO.

Department of Health, Department for Education and Employment and Home Office (2000) *Framework for the Assessment of Children in Need and their Families*. London: The Stationery Office.

(http://webarchive.nationalarchives.gov.uk/20130401151715/https://www.education.gov.uk/publications/eOrderingDownload/Framework%20for%20the%20assessment%20of%20children%20in%20need%20and%20their%20families.pdf)

DiClemente, C.C., Nidecker, M. and Bellack, A.S. (2008) 'Motivation and the stages of change among individuals with severe mental illness and substance abuse disorders.' *Journal of Substance Abuse Treatment 34*, 1, 25-35.

DiClemente, C.C. and Prochaska, J.O. (1982) 'Self change and therapy change of smoking behavior: a comparison of processes of change in cessation and maintenance.' *Addictive Behaviors 7, 2,* 133–142.

DiLalla, D.L. and Crittenden, P.M. (1990) 'Dimensions of maltreated children's home behaviour: A factor analytic approach.' *Infant Behavior and Development 13*, 439-460.

Dimitrova, N., Pierrehumbert, B., Glatz, N., Torrisi, R., Heinrichs, M., Halfon, O. and Chouchena, O. (2010) 'Closeness in relationships as a mediator between sexual abuse in childhood or adolescence and psychopathological outcome in adulthood.' *Clinical Psychology and Psychotherapy 17*, 183–195.

Dixon, L., Browne, K.D., Hamilton-Giacritsis, C. (2005) 'Risk factors of parents abused as children: a mediational analysis of the intergenerational continuity of child maltreatment (Part 1).' *Journal of Psychology and Psychiatry 46*, 1, 47–57.

Drieschner, K.H., Lammers, S.M.M. and Van der Staak, C.P.F. (2004) 'Treatment motivation: an attempt for clarification of an ambiguous concept.' *Clinical Psychology Review 23*, 8, 1115-1137.

Duffy, P. and Baldwin, H. (2013) 'Recovery post treatment: plans, barriers and motivators.' *Substance Abuse Treatment, Prevention, and Policy 8*, 6.

Dumbrill, G. (2006) 'Parental experience of child protection intervention: a qualitative study.' *Child Abuse and Neglect 30*, 1, 27-37.

Dumaret and Coppel-Batsch (1998) 'Effects in adulthood of separations and long-term foster care: a French research study.' *Adoption & Fostering 22*, 1, 31-39.

Eckenrode, J., Ganzel, B., Henderson, Jr, C.R., Smith; E., Olds, D.L., Powers, J., Cole, R., Kitzman, H. and Sidora, K. (2000) 'Preventing child abuse and neglect with a program of nurse home visitation: the limiting effects of domestic violence.' *Journal of the American Medicine Association 284*, 11, 1385-91.

Edmonson, P. and Schneider, E. (2001) 'Supporting parents with learning difficulties.' Disability, Pregnancy and Parenthood International 36, 9–10.

Edwards, M., Tinworth, K., Burford, G. and Pennell, J. (2007) Family team meeting (FTM) process, outcome, and impact evaluation phase II report. American Humane Association: Englewood, CO.

Egeland, B., Stroufe, A. and Erickson, M. (1983) 'The developmental consequences of different patterns of maltreatment.' *Child Abuse & Neglect* 7, 4, 459-469.

Ekendahl, M. (2007) 'Time to change - an exploratory study of motivation among untreated and treated substance abusers.' *Addiction Research & Theory 15*, 3, 247-261.

Eshel, N., Daelmans, B., Cabral de Mello, M. and Martines, J. (2006) 'Responsive parenting: interventions and outcomes.' *Bulletin of the World Health Organization 84*, 12, 992-999.

Etherington, K. (2007) 'Creation as transformation: parenting as a turning point in drug users' lives.' *Counselling and Psychotherapy Research: Linking research with practice 7*, 2, 71-78.

Falkin, G.P. and Strauss, S.M (2003) 'Social supporters and drug use enablers. A dilemma for women in recovery.' *Addictive Behaviors 28*, 141–155.

Farmer, E. (2010) 'What factors relate to good placement outcomes in kinship care?' *British Journal of Social Work 40*, 426–444.

Farmer, E. and Owen, M. (1995) *Child Protection Practice: Private Risks and Public Remedies.* London: HMSO.

Farmer, E., Sturgess, W., O'Neill, T. and Wijedasa, D (2011) *Achieving Successful Returns from Care: What Makes Reunification Work?* London: BAAF.

Farmer, E. and Lutman, E. (2012) Working Effectively with Neglected Children and their Families – Understanding their Experienced and Long-term Outcomes. London: Jessica Kingsley Publishers.

Fauth, R., Jelicic, H., Hart, D., Burton, S., Shemmings, D., Bergeron, C., White, K. and Morris, M. (2010) *Effective Practice to Protect Children Living in 'Highly Resistant' Families*. London: Centre for Excellence and Outcomes in Children and Young People's Services.

Featherstone, B. and Fraser, C. (2012) 'I'm just a mother. I'm nothing special, they're all professionals': parental advocacy as an aid to parental engagement.' *Child and Family Social Work* 17, 2, 244–253.

Featherstone, B., Fraser, C., Ashley, C. and Ledward, P. (2011) 'Advocacy for parents and carers involved with children's services: making a difference to working in partnership?' *Child and Family Social Work 16*, 3, 266–275.

Feldman, M., Case, L. Rincover, A., Towns, F. and Betel, J. (1989) 'Parent education project III: increasing affection and responsivity in developmentally handicapped mothers.' *Journal of Applied Behavior Analysis* 22, 2, 211-22.

Fernandez, E. (2007) 'Supporting children and responding to their families: capturing the evidence on family support.' *Children and Youth Services Review 29*, 10, 1368–1394.

Field, C.A., Duncan, J., Washington, K. and Adinoff, B. (2007) 'Association of baseline characteristics and motivation to change among patients seeking treatment for substance dependence.' *Drug and Alcohol Dependence 91*, 1, 77-84.

Finney, A. (2004) Alcohol and Intimate Partner Violence: Key Findings from the Research. London: Home Office

Fisher, D., Beech, A. and Browne, K. D. (1998) 'Locus of control and its relationship to treatment change and abuse history in child sexual abusers.' *Legal and Criminological Psychology* 3, 1, 1–12.

Fixsen, D., Naoom, S.F., Blase, K.A., Friedman, R.M. and Wallace, F. (2005) Implementation research: A Synthesis of the Literature. Tampa, FL: University of South Florida, Louis de la Parte Florida Mental Health Institute, The National implementation Research Network.(FMHI Publication #231).

Fluke, J., Yuan, Y. and Edwards, M. (1999) 'Recurrence of maltreatment: an application of the National Child Abuse and Neglect Data System (NCANDS).' *Child Abuse &Neglect* 23, 633-50.

Forrester, D. (forthcoming) 'Engaging parents and protecting children? A randomized controlled trial of motivational interviewing in child protection work.' Further information available at: http://www.beds.ac.uk/goldbergcentre/research/current-research

Forrester, D., Cocker, C., Goodman, K., Binnie, C. and Jensch, G. (2009) 'What is the impact of public care on children's welfare? A review of research findings and their policy implications.' *Journal of Social Policy* 38, 3, 439-456.

Forrester, D. and Harwin, J. (2008) 'Parental substance misuse and child welfare: outcomes for children two years after referral.' *British Journal of Social Work 38*, 8, 1518–1535.

Forrester, D. and Harwin, J. (2011) *Parents Who Misuse Drugs and Alcohol: Effective Interventions in Social Work and Child Protection*. Chichester: Wiley-Blackwell.

Forrester, D., Kershaw, S., Moss, H. and Hughes, L. (2008) 'Communication skills in child protection: how do social workers talk to parents?' *Child & Family Social Work 13*, 1, 41–51.

Forrester, D., Westlake, D. and Glynn, G. (2012) 'Parental resistance and social worker skills: towards a theory of motivational social work.' *Child & Family Social Work 17*, 2, 118-129.

Fraser, C., McIntyre, A. and Manby, M. (2009) 'Exploring the impact of parental drug/alcohol problems on children and parents in a midlands county in 2005/06.' *British Journal of Social Work 39*, 846–866 doi:10.1093.

Frasier, P.Y., Slatt, L., Kowlowitz, V. and Glowa, P.T. (2001) 'Using the stages of change model to counsel victims of intimate partner violence.' *Patient Education and Counselling* 43, 2, 211–217.

Freyer, J., Scott Tonigan, J., Keller, S., Rumpf, H., John, U. and Hapke, U. (2005) 'Readiness for change and readiness for help-seeking: a composite assessment of client motivation.' *Alcohol & Alcoholism 40*, 6, 540–544.

Giesen-Bloo, J., Van Dyck, R., Spinhoven, P., Van Tilburg, W., Dirksen, C., Van Asselt, V., Kremers, I., Nadort, M. and Arntz, A. (2006) 'Outpatient psychotherapy for borderline personality disorder: randomized trial of Schema-Focused Therapy vs Transference-Focused Psychotherapy.' *Archives of General Psychiatry* 63, 6, 649-658.

Ginsburg, J.I.D., Mann, R.E., Rotgers, F. and Weekes, J.R. (2002) 'Motivational Interviewing with criminal justice populations.' In W. R. Miller and S. Rollnick (eds.) *Motivational Interviewing: Preparing People to Change Addictive Behavior* (2nd Edition). New York: Guilford Press.

Girvin, H. (2004) 'Beyond 'stages of change:' using readiness for change and caregiver-reported problems to identify meaningful subgroups in a child welfare sample.' *Children and Youth Services Review 26*, 10, 897–917.

Gladstone, J., Dumbrill, G., Leslie, B., Koster, A., Young, M. and Ismaila, A. (2012) 'Looking at engagement and outcome from the perspectives of child protection workers and parents.' *Children and Youth Services Review 34*, 1, 112-118.

Glazemakers, I. and Deboutte, D. (2013) 'Modifying the 'Positive Parenting Program' for parents with intellectual disabilities.' *Journal of Intellectual Disability Research 57, 7,* 616–626.

Gondolf, E.W. (1987) 'Evaluating programs for men who batter: problems and prospects.' *Journal of Family Violence* 2, 1, 95-108.

Gondolf, E.W. (2002) *Batterer Intervention Systems: Issues, Outcomes and Recommendations*. London: Sage.

Gondolf, E.W. (2004) 'Evaluating batterer counselling programs: a difficult task showing some effects and implications.' *Aggression and Violent Behaviour 9*, 6, 605-631.

Gondolf, E.W. (2011) 'The weak evidence for batterer program alternatives.' *Aggression and Violent Behaviour 16*, 347-353.

Gossop, M., Stewart, D. and Marsden, J. (2006) 'Readiness for change and drug use outcomes after treatment.' *Addiction 102*, 2, 301-308.

Grant, T., Ernst, C.C., Pagalilauan, G. and Streissguth, A. (2003) 'Post program follow-up effects of paraprofessional intervention with high-risk women who abused alcohol and drugs during pregnancy.' *Journal of Community Psychology 31*, 3, 211–222

Green, J. and Goldwyn, R. (2002) 'Annotation: attachment disorganisation and psychopathology: new findings in attachment research and their potential implications for developmental psychopathology in childhood.' *Journal of Child Psychology and Psychiatry and Allied Disciplines 43*, 7, 835-846.

Gregoire, T.K. and Burke, A.C. (2004) 'The relationship of legal coercion to readiness to change among adults with alcohol and other drug problems.' *Journal of Substance Abuse Treatment 26*, 1, 35-41.

Gunderson, K., Cahn, K. and Wirth, J. (2003) 'The Washington State long-term outcome study.' *Protecting Children 18*,1 & 2, 42–47.

Hahn, E., Simpson, M. and Kidd, P. (1996) 'Cues to parent involvement in drug prevention and school activities.' *Journal of School Health 66*, 5, 165–171.

Hammond, K. (1996) *Human Judgement and Social Policy: Irreducible Uncertainty, Inevitable Error, Unavoidable Injustice.* Oxford: Oxford University Press.

Hannett, M.J. (2007) 'Lessening the sting of ASFA: the rehabilitation-relapse dilemma brought about by drug addiction and termination of parental rights.' *Family Court Review* 45, 3, 524–537.

Haresnape, S. and Brown, L. (2013) *Development and Trial of an Accreditation Framework for Family Group Conference Services.* London: Family Rights Group.

Haringey Local Safeguarding Children Board (2010) Second Serious Case Review Overview Report Relating to Peter Connelly. London: Department for Education.

Harnett, P.H. (2007) 'A procedure for assessing parents' capacity for change in child protection cases.' *Children and Youth Services Review 29*, 9, 1179–1188.

Harnett, P. and Dawe, S. (2007) 'Reducing potential for child abuse among methadone-maintained parents: Results from a randomized controlled trial.' *Journal of Substance Abuse Treatment* 32, 381-390.

Harnett, P. and Dawe, S. (2008) 'Reducing child abuse potential in families identified by social services: implications for assessment and treatment.' *Brief Treatment and Crisis Intervention* 8, 3, 226-235.

Harnett, P. and Day, C. (2008) 'Developing pathways to assist parents to exit the child protection system in Australia.' *Clinical Psychologist 12*, 3, 79-95.

Harwin, J., Alrouh, B., Ryan, M. and Tunnard, J. (2013) 'Strengthening prospects for safe and lasting family reunification: can a Family Drug and Alcohol Court make a contribution?' *Journal of Social Welfare and Family Law 35*, 4, 459-474.

Harwin, J., Alrouh, B., Ryan, M. and Tunnard, J. (2014) Changing Lifestyles, Keeping Children Safe: an evaluation of the first Family Drug and Alcohol Court (FDAC) in care proceedings. London: Brunel University.

Harwin, J. and Ryan, M. (2008) 'The role of the court in cases concerning parental substance misuse and children at risk of harm.' *Journal of Social Welfare and Family Law* 29, 3-4, 277-292.

Harwin, J., Ryan, M., Tunnard, J., Alrouh, B., Matias, C., Momenian-Schneider, S. and Pokhrel, S. (2011) *FDAC Evaluation Final Report*. London: Brunel University.

Hastings, A., Bailey, N., Besemer, K., Bramley, G., Gannon, M. and Watkins, D. (2013) *Coping with Cuts? Local Government and Poorer Communities.* York: Joseph Rowntree Foundation.

Haugaard, J.J., Wojslawowicz, J.C. and Palmer, M. (1999) 'Outcomes in adolescent and older-child adoptions.' *Adoption Quarterly* 3, 1, 61-69.

Healy, K. and Darlington, Y. (2009) 'Service user participation in diverse child protection contexts: principles for practice.' *Child and Family Social Work 14*, 4, 420–430.

Health Scotland *What is the Starting Well Project?* accessed at : http://www.healthscotland.com/uploads/documents/3495-SWSection%201.pdf January 2014

Hegarty, K., O'Doherty, L., Gunn, J., Pierce, D. and Taft, A. (2008) 'A brief counselling intervention by health professionals utilising the "readiness to change" concept for women experiencing intimate partner abuse: the Weave Project.' *Journal of Family Studies 14*, 2–3, 376–388.

Heptinstall, E. (2000) 'Gaining access to looked after children for research purposes: lessons learned.' *British Journal of Social Work 30*, 6, 867-872.

Herrenkohl, E.C.H., Herrenkohl, R.C., Egolf, B.P. and Russo, M.J. (1998) 'The relationship between early maltreatment and teenage parenthood.' *Journal of Adolescence 21*, 291–303.

Hess, P. and Folaron, G. (1991) 'Ambivalences: a challenge to permanency for children.' Child Welfare 70, 402–425.

Hesse, E. and Main, M. (2000) 'Disorganized infant, child, and adult attachment: collapse in behavioral and attentional strategies.' *Journal of the American Psychoanalytic Association 48*, 1097-1127.

Hester, M. (2013) 'Who does what to whom? Gender and domestic violence perpetrators in English police records.' *European Journal of Criminology* 10, 5, 623-637.

Hester, M., Westmarland, N., Gangoli, G., Wilkinson, M., O'Kelly, C., Kent, A. and Diamond, A. (2006) *Domestic Violence Perpetrators: Identifying Needs to Inform Early Intervention*. Bristol: University of Bristol in association with the Northern Rock Foundation and the Home Office.

Hettema, J., Steele, J. and Miller, W.R. (2005) 'Motivational Interviewing.' *Annual Review of Clinical Psychology 1*, 91-111.

Hibbert, L.J. and Best, D.W. (2011) 'Assessing recovery and functioning in former problem drinkers at different stages of their recovery journeys.' *Drug and Alcohol Review,* 30, 12 – 20.

Higginson, S. and Mansell, W. (2008) 'What is the mechanism of psychological change? A qualitative analysis of six individuals who experienced personal change and recovery.' *Psychology and Psychotherapy: Theory, Research and Practice 81*, 3, 309-328.

Hildyard, K.L. and Wolfe, D.A. (2002) 'Child neglect: developmental issues and outcomes.' *Child Abuse & Neglect 26*, 679–695.

Hiller, M.L., Knight, K., Broome, K.M. and Simpson, D.D. (1998) 'Legal pressure and treatment retention in a national sample of long-term residential programs.' *Criminal Justice and Behavior 25*, 4, 463-481.

Hindley, N., Ramchandani, P. and Jones, D. (2006) 'Risk factors for recurrence of maltreatment: a systematic review.' *Archives of Disease in Childhood 91*, 9, 744-752.

Hinshaw, S.P. (2005) 'The stigmatization of mental illness in children and parents: developmental issues, family concerns, and research needs.' *Journal of Child Psychology and Psychiatry 46*, 7, 714–734.

HM Government (2013) Working Together to Safeguard Children: A Guide to Inter-Agency Working to Safeguard and Promote the Welfare of Children. London: Department for Education.

Hofer, M.A. (1995) 'Hidden regulators: implications for a new understanding of attachment, separation, and loss.' In S. Goldberg, R. Muir and J. Kerr (eds) *Attachment Theory: Social, Developmental, and Clinical Perspectives*. Hillsdale, NJ: Analytical Press.

Holt, K., Broadhurst, K., Doherty, P. and Kelly, N. (2013) 'Access to justice for families? Legal advocacy for parents where children are on the 'edge of care': an English case study.' *Journal of Social Welfare and Family Law 35*, 2, 163-177.

Holland, S. (1999) 'Discourses of decision-making in child protection: conducting comprehensive assessments in Britain.' *International Journal of Social Welfare 8*, 4, 276-286.

Holland, S. (2000) 'The assessment relationship: interactions between social workers and parents in child protection assessments.' *British Journal of Social Work 30*, 149-163.

Holland, S., Scourfield, J. O'Neill, S. and Pithouse, A. (2005) 'Democratising the family and the state? The case of Family Group Conferences in child welfare.' *Journal of Social Policy 34*, 1, 59-77.

Holt, S., Buckley, H. and Whelan, H. (2008) 'The impact of exposure to domestic violence on children and young people: A review of the literature.' *Child Abuse & Neglect 32*, 797–810.

Horwath, J. and Morrison, T. (2001) 'Assessment of parental motivation to change.' In J. Horwath (Ed) *The Child's World. Assessing Children in Need.* London: Jessica Kingsley Publishers.

Horwitz, S. M., Balestracci, K. M. B. and Simms, M. D. (2001) 'Foster care placement improves children's functioning.' *Archives of Paediatric and Adolescent Medicine 155*, 11, 1255–60.

House of Commons Children, Schools and Families Committee (2009) *Children, Schools and Families Committee - Third Report Looked-after Children.* Available at: http://www.publications.parliament.uk/pa/cm200809/cmselect/cmchilsch/111/11102.

Howe, D. (2006) 'Developmental attachment psychotherapy with fostered and adopted children.' *Child and Adolescent Mental Health 11*, 3, 128-134.

Hser, Y., Hoffman, V., Grella, C.E. and Anglin, D. (2001) 'A 33-Year follow-up of narcotics addicts.' *Archive of General Psychiatry 58*, 5, 503-508.

Hser, Y., Longshore, D. and Anglin, M.D. (2007) 'The life course perspective on drug use: A conceptual framework for understanding drug use trajectory and its management.' *Evaluation Review 31*, 6, 515–547.

Humphreys, C., Thiara, R.K. and Skamballis, A. (2011) 'Readiness to change: mother-child relationship and domestic violence intervention.' *British Journal of Social Work 41*, 1, 166-184.

Hunt, J., Waterhouse, S. and Lutman, E. (2008) *Keeping them in the Family : Children Placed in Kinship Care through Care Proceedings*. London: BAAF.

Hutchings, J., Bywater, T., Daley, D., Gardner, F., Whitaker, C., Jones, K., Eames, C. and Edwards, R.T. (2007) 'Parenting intervention in Sure Start services for children at risk of developing conduct disorder: pragmatic randomised controlled trial.' *British Medical Journal* 334,678.

Hutchings, J., Bywater, T., Williams, M.E., Whitaker, C., Lane, E. and Shakespeare, K (2011) 'The extended school aged Incredible Years parent programme.' *Child and Adolescent Mental Health Volume 16*, 3, 136–143.

Jack, G. and Gill, O. (2003) The Missing Side of the Triangle - Assessing the Importance of Family and Environmental Factors in the Lives of Children. Barkingside: Barnardo's.

Jack, G. and Gill, O. (2010) 'The role of communities in safeguarding children and young people.' *Child Abuse Review 19*, 82–96.

Jack. G.and Gill, O. (2013) 'Developing cultural competence for social work with families living in poverty.' *European Journal of Social Work 16*, 2, 220-234.

Jack, G. and Jordan, B. (1999) 'Social capital and child welfare.' *Children and Society 13*, 242-256.

James, H. (2004) 'Promoting effective working with parents with learning disabilities.' *Child Abuse Review 13*, 1, 31-41.

Janis, I.L. and Mann, L. (1977) *Decision-Making: a Psychological Analysis of Conflict, Choice and Commitment*. London: Cassell and Collier MacMillan.

Jarrett, C. (2008) 'When therapy causes harm.' Psychologist 21, 1, 10-12.

Joe, G., Simpson, D. and Broome, K. (1999) 'Retention and patient engagement models of different treatment modalities in DATOS.' *Drug and Alcohol Dependence, 57*, 2, 113-125.

Jones, T.L. and Prinz, R.J. (2005) 'Potential roles of parental self-efficacy in parent and child adjustment: A review.' *Clinical Psychology Review 25*, 341–363.

Jones, D. (2009) 'Assessment of parenting.' In J. Howarth (ed) *The Child's World: The Comprehensive Guide to Assessing Children in Need* (2nd Edition). London: Jessica Kingsley Publishers.

Jones, K., Daley, D., Hutchings, J., Bywater, T. and Eames, C. (2007) *Efficacy of the Incredible Years basic parent training programme as an early intervention for children with conduct problems and ADHD*. Bangor: School of Psychology, University of Wales.

Jones, D., Hindley, N. and Ramchandani, P. (2006) 'Making plans: assessment, intervention and evaluating outcomes.' In J. Aldgate, D. Jones and C. Jeffery (eds) *The Developing World of the Child.* London: Jessica Kingsley Publishers.

Jung-Beeman, M., Bowden, E.M., Haberman, J., Frymiare, J.L., Arambel-Liu, S., Greenblatt, R., Reber, P.J. and Kounios, J. (2004) 'Neural activity when people solve verbal problems with insight.' *PLOS Biology* 2, 4, 0500-0510.

Kahneman, D. (2011) Thinking, Fast and Slow. London: Allen Lane

Kahneman D., Slovic, P. and Tversky, A. (1990) *Judgements under uncertainty: Heuristics and biases*. Cambridge: Cambridge University Press.

Keating, D., Fowler, C. and Briggs, C. (2008) 'Evaluating the Family Partnership Model (FPM) program and implementation in practice in New South Wales, Australia.' *Australian Journal of Advanced Nursing* 25, 2, 28-35.

Kerr, D.C.R., Capaldi, D.M., Owen, L.D., Wiesner, M. and Pears, K.C. (2011) 'Changes in at-risk American men's crime and substance use trajectories following fatherhood.' *Journal of Marriage and Family 73*, 5, 1101–1116.

Keyes, M.A., Sharma, A., Elkins, I.J., Iacono, W.G. and McGue, M. (2008) 'The mental health of US adolescents adopted in infancy.' *Archives of Pediatrics and Adolescent Medicine* 162, 5, 419-425.

Kirkman, E. and Melrose, K. (2014) *Clinical Judgement and Decision-Making in Children's Social Work: An analysis of the 'front door' system, DFE RR37*. London: Department for Education.

Kirkpatrick, S., Barlow, J., Stewart-Brown, S. and Davis, H. (2007) 'Working in partnership: user perceptions of intensive home visiting.' *Child Abuse Review 16*, 1, 32-46.

Kim, K.L., Davis, M.I., Jason, L.A. and Ferrari, J.R. (2006) 'Structural social support: impact on adult substance use and recovery attempts.' *Journal of Prevention & Intervention in the Community 31*, 1-2, 85-94.

King, K.E., Ross, L.E., Bruno, T.L. and Erickson, P.G. (2009) 'Identity work among street-involved young mothers.' *Journal of Youth Studies 12*, 2, 139-149.

Kiresuk, T., Smith, A. and Cardillo, J. (1994) *Goal Attainment Scaling: Applications, Theory and Measurement*. Hillsdale, NJ, England: Lawrence Erlbaum Associates Inc.

Knis-Matthews, L. (2007) 'The role of spouses and extended family members as primary caretakers of children during a parent's drug addiction.' *Occupational Therapy in Mental Health* 23,1, 1-19.

Knudsen, E.I. (2004) 'Sensitive periods in the development of the brain and behavior.' *Journal of Cognitive Neuroscience 16,* 8, 1412-1425.

Koenig, A.L., Cicchetti, D. and Rogosch, F.A. (2004) 'Moral development: the association between maltreatment and young children's prosocial behaviors and moral transgressions.' *Social Development 13*, 1, 87-106.

Kohl, P.L., Jonson-Reid, M. and Drake, B. (2011) 'Maternal mental illness and the safety and stability of maltreated children.' *Child Abuse & Neglect 35*, 309–318.

Koriat, A., Lichenstein, S. and Fischoff, B. (1980) 'Reasons for confidence.' *Journal of Experimental Psychology: Human Learning and Memory 6*, 107-118.

Kreager, D.A., Matsueda, R.L. and Erosheva, E.A. (2010) 'Motherhood and criminal desistance in disadvantaged neighborhoods.' *Criminology 48*, 1, 221-258.

Kroll, B. and Taylor, A. (2003) *Parental Substance Misuse and Child Welfare*. London: Jessica Kingsley Publishers.

La France, M. and Hecht, M. A. (1995) 'Why smiles generate leniency.' *Personality and Social Psychology Bulletin 21*, 207–214.

Laming (2003) The Victoria Climbie Inquiry: Report of an Inquiry by Lord Laming. London: HMSO.

Latham, G.P., Erez, M. and Locke, E. (1988) 'Resolving scientific disputes by the joint design of crucial experiments by the antagonists: application to the Erez–Latham dispute regarding participation in goal setting.' *Journal of Applied Psychology* 73, 4, 753–772.

Latham, G.P. and Locke, E. (2007) 'New developments in and directions for goal-setting research.' *European Psychologist 12*, 4, 290–300.

Laudet, A.B. and White, W.L. (2008) 'Recovery capital as prospective predictor of sustained recovery, life satisfaction, and stress among former poly-substance users.' Substance Use and Misuse 43,1, 27-54.

Laudet, A.B. and White, W.L. (2010) 'What are your priorities right now? Identifying service needs across recovery stages to inform service development.' *Journal of Substance Abuse Treatment 38*, 51–59.

Lawrence, C., Carlson, E.A. and Egeland, B. (2006). 'The impact of foster care on development.' *Development and Psychopathology* 28, 1, 57-76.

Leschied, A.W., Chiodo, D., Whitehead, P.C., Hurley, D. and Marshall, L. (2003) 'The empirical basis of risk assessment in child welfare: the accuracy of risk assessment and clinical judgement.' *Child Welfare 82*, 5, 527-540.

Levendosky, A.A., Huth-Bocks, A. and Semel, M.A. (2002) 'Adolescent peer relationships and mental health functioning in families with domestic violence.' *Journal of Clinical Child & Adolescent Psychology 31*, 2, 206-218.

Lewis, M., Feiring, C., McGuffog, C. and Jaskir, J. (1984) 'Predicting psychopathology in six year olds from early social relations.' *Child Development 55*, 1, 123-36.

Liau, A. K., Shively, R., Horn, M., *et al.* (2004) 'Effects of psycho-education for offenders in a community correctional facility.' *Journal of Community Psychology 32*, 543–558.

Lilienfeld, S.O. (2007) 'Psychological treatments that cause harm.' *Perspectives on Psychological Science 2*, 1, 53–70.

Linehan, M.M (2000) The empirical basis of dialectical behavior therapy: Development of new treatments *versus* evaluation of existing treatments. *Clinical Psychology: Science and Practice* 7, 1, 113-119.

Linehan, M.M., Dimeff, L.A., Reynolds, S.K., Comtois, K.A., Shaw Welch, S., Heagerty, P. and Kivlahan, D.R. (2002) 'Dialectical behavior therapy *versus* comprehensive validation therapy plus 12-step for the treatment of opioid dependent women meeting criteria for borderline personality disorder.' *Drug and Alcohol Dependence* 67,13, 26.

Littell, J.H. and Girvin, H. (2005) 'Caregivers' readiness for change: predictive validity in a child welfare sample.' *Child Abuse & Neglect: The International Journal* 29, 1, 59-80.

Littell, J.H. and Girvin, H. (2006) 'Correlates of problem recognition and intentions to change among caregivers of abused and neglected children.' *Child Abuse & Neglect 30*, 12, 1381–1399.

Littell, J.H. and Tajima, E.A. (2000) 'A multilevel model of client participation in intensive family preservation services.' *Social Service Review 74*, 405-435.

Little, M., Berry, V., Morpeth, L., Blower, S., Axford, N., Taylor, R., Bywater, T., Lehtonen, M. and Tobin, K. (2012) 'The impact of three evidence-based programmes delivered in public systems in Birmingham, UK.' *International Journal of Conflict and Violence 6*, 2, 260-272.

Llewellyn, G., McConnell, D., Honey, A., Mayes, R. and Russo, D. (2003) 'Promoting health and home safety for children of parents with intellectual disability: a randomized controlled trial.' *Research in Developmental Disabilities 24*, 6, 405–431.

Lloyd, C.E. and Barth, R.P. (2011) 'Developmental outcomes after five years for foster children returned home, remaining in care, or adopted' *Children and Youth Services Review* 33, 8, 1383–1391.

Locke, E.A. and Latham, G.P. (2002) 'Building a practically useful theory of goal setting and task motivation: A 35-Year odyssey.' *American Psychologist 57*, 9, 705-717.

Losel, F. and Schmucker, M. (2005) 'The effectiveness of treatment for sexual offenders: a comprehensive meta-analysis.' *Journal of Experimental Criminology* 1, 1, 117–146.

Lupton, C. and Nixon, P. (1999) *Empowering Practice? A Critical Appraisal of the Family Group Conference Approach*. Bristol: The Policy Press.

MacInnes, T., Aldridge, H., Bushe, S., Kenway, P. and Tinson, A. (2013) *Monitoring Poverty and Social Exclusion*. York: Joseph Rowntree Foundation.

MacIntyre, G. and Stewart, A. (2011) 'For the record: the lived experience of parents with a learning disability – a pilot study examining the Scottish perspective.' *British Journal of Learning Disabilities 40*, 1, 5–14.

MacMillan, H.L., Wathen, C.N., Barlow, J., Fergusson, D.M., Leventhal, J.M. and Taussig, H.N. (2009) 'Child maltreatment 3: Interventions to prevent child maltreatment and associated impairment.' *The Lancet 373*, 9659, 250–266.

Magura, S. and Laudet, A.B. (1996) 'Parental substance abuse and child maltreatment: Review and implications for intervention.' *Children and Youth Services Review* 18, 3, 193-220.

Marganska, A., Gallagher, M. and Miranda, R. (2013) 'Adult attachment, emotion dysregulation and symptoms of depression and generalized anxiety disorder.' *American Journal of Orthopsychiatry* 83, 131–141.

Masson, J. Dickens, J., Bader, K and Young, J. (2013) *Partnership by Law? The pre-proceedings process for families on the edge of care proceedings. Report to ESRC,* School of Law University of Bristol and Centre for Research on Children and Families, University of East Anglia available at http://www.bristol.ac.uk/law/research/ researchpublications/2013/partnershipbylaw.pdf

Masson, J., Pearce, J., Bader, K., Joyner, O., Marsden, J. and Westlake, D. (2008) *Care Profiling Study*. London: Ministry of Justice.

McConnell, D., Feldman, M., Aunos, M. and Prasad, N. (2011) 'Child maltreatment investigations involving parents with cognitive impairments in Canada.' *Child Maltreatment 16*, 1, 21-32.

McCrory, E., De Brito, S.A. and Viding, E. (2012) 'The link between child abuse and psychopathology: a review of neurobiological and genetic research.' *Journal of the Royal Society of Medicine 105*, 4, 151-156.

McCurdy, K. and Daro, D. (2001) 'Parent involvement in family support programs: an integrated theory.' *Family Relations 50*, 2, 113-121.

McGaw, S. (2000) What Works for Parents with Learning Disabilities? London: Barnardos

McGaw, S., Ball, K. and Clark, A. (2002) 'The effect of group intervention on the relationships of parents with intellectual disabilities.' *Journal of Applied Research in Intellectual Disabilities* 15, 354–366.

McGaw, S. and Sturmey, P. (1994) 'Assessing parents with learning disabilities: The parental skills model'. *Child Abuse Review*, 3, 36-51

McKeigue, B. and Beckett, C. (2010) 'Squeezing the toothpaste tube: will tackling court delay result in pre-court delay in its place?' *British Journal of Social Work 40*, 1, 154–169.

Merritt, D.H. (2009) 'Child abuse potential: correlates with child maltreatment rates and structural measures of neighborhoods.' *Children and Youth Services Review 31*, 8, 927–934.

Miller, W. R. and Rollnick, S. (2013) *Motivational Interviewing: Helping People Change*. (3rd ed.) New York: Guilford Press.

Ministry of Justice (2011) Family Justice Review. London: Ministry of Justice.

Ministry of Justice (2014) *Public Law Outline*. London: Ministry of Justice. Available at: http://www.justice.gov.uk/protecting-the-vulnerable/care-proceedings-reform

Moe, V. and Slinning, K. (2001) 'Children prenatally exposed to substances: Gender-related differences in outcome from infancy to 3 years of age.' *Infant Mental Health Journal* 22, 3, 334 – 350.

Montgomery, P., Gardner, F., Bjornstad, G. and Ramchandani, P. (2009) *Systematic Reviews of Interventions following Physical Abuse: Helping Practitioners and Expert Witnesses Improve the Outcomes of Child Abuse.* Report to Department for Children, Schools and Families: University of Oxford (Department for Education Research Brief DCSF-RBX-09-08A).

Moos, R.H. and Moos, B.S. (2006) 'Rates and predictors of relapse after natural and treated remission from alcohol use disorders.' *Addiction 101*, 2, 212–222.

Morgan, R. (2008) Children's Care Monitor 2008: Children's Views on How Care is Doing. London: Ofsted.

Morris, K. and Connolly, M. (2012) 'Family decision making in child welfare: Challenges in developing a knowledge base for practice.' *Child Abuse Review 21*, 41-52.

Morrison, T. (2010) 'Assessing parental motivation for change.' In J. Horwath (ed) *The Child's World: The Comprehensive Guide to Assessing Children in Need*. London: Jessica Kingsley Publishers.

Morse J, Hill JW, Pilkonis P, Yaggi K, Broyden N, Stepp S, Reed L, Feske U. (2009) 'Anger, preoccupied attachment, and domain disorganization in borderline personality disorder.' *Journal of Personality Disorder* 23, 3, 240-57.

Mowbray, C., Oyserman, D., Bybee, D. and MacFarlane, P. (2002) 'Parenting of mothers with a serious mental illness: differential effects of diagnosis, clinical history, and other mental health variables.' *Social Work Research* 26, 4, 225-240.

Munby, J. (2013a) View from the President's Chambers [2] The Process of Reform: The Revised PLO and the Local Authority. London: Ministry of Justice.

Munby, J (2013b) Annex to Practice Direction 36c: Pilot Practice Direction 12A Care, Supervision and Other Part 4 Proceedings: Guide to Case Management. London: Ministry of Justice

Munro, E. (1999) 'Common errors of reasoning in child protection work.' *Child Abuse & Neglect*, 23, 8, 745-758.

Munro, Eileen (2005) 'A systems approach to investigating child abuse deaths.' *British Journal of Social Work 35*, 4, 531-546.

National Collaborating Centre for Mental Health (2010) *Antisocial Personality Disorder: Treatment, Management and Prevention. National Clinical Practice Guideline Number* 77. London: The British Psychological Society and The Royal College of Psychiatrists.

National Institute for Health and Care Excellence (2007a) *Drug Misuse: Psychosocial interventions (CG51).* Available at: http://guidance.nice.org.uk/CG51

National Institute for Health and Care Excellence (2007b) *Drug Misuse – Opioid detoxification (CG52)*. Available at: http://publications.nice.org.uk/drug-misuse-opioid-detoxification-cg52

National Institute for Health and Care Excellence (2009) *Antisocial Personality Disorder: Treatment, management and prevention (CG 77)*. Available at: http://publications.nice.org.uk/antisocial-personality-disorder-cg77

National Institute for Health and Care Excellence (2011a) *Alcohol-use Disorders: Diagnosis, assessment and management of harmful drinking and alcohol dependence* (CG115). Available at: http://publications.nice.org.uk/alcohol-use-disorders-diagnosis-assessment-and-management-of-harmful-drinking-and-alcohol-cg115

National Institute for Health and Care Excellence (2011b) *Common Mental Health Disorders: Identification and pathways to care (CG123)*. Available at: http://publications.nice.org.uk/common-mental-health-disorders-cg123

National Institute for Health and Care Excellence (2011c) *Psychosis with Coexisting Substance Misuse: Assessment and management in adults and young people (CG120).* Available at: http://publications.nice.org.uk/psychosis-with-coexisting-substance-misuse-cq120

National Institute for Health and Care Excellence (2014) *Domestic Violence and Abuse:* How social care, health services and those they work with can identify, prevent and reduce domestic violence and abuse. London: National Institute for Health and Care Excellence.

Naughton, A.M., Maguire, S.A., Mann, M.K., Lumb, R.C., Tempest, V., Gracias, S. and Kemp, A.M. (2013) 'Emotional, behavioral, and developmental features indicative of neglect or emotional abuse in preschool children: a systematic review.' *JAMA Pediatrics* 167, 8, 769-75.

Neil, E. (2000) 'The reasons why young children are placed for adoption: findings from a recently placed sample and implications for future identity issues.' *Child and Family Social Work 5*, 4, 303-316.

Neil, E. (2010) 'The benefits and challenges of direct post-adoption contact: perspectives from adoptive parents and birth relatives.' *Aloma: Revista de Psicologia, Ciències de l'Educació i de l'Esport, 27*, 89-115.

Newton, R., Litrownik, A. and Landsverk, J. (2000) 'Children and youth in foster care: disentangling the relationship between problem behaviours and number of placements.' *Child Abuse & Neglect 24*, 1363-1374.

Norman, R.E., Byambaa, M., De, R., Butchart, A., Scott, J. and Vos, T. (2012) 'The long-term health consequences of child physical abuse, emotional abuse, and neglect: a systematic review and meta-analysis.' *PLoS Medicine 9*, 11, e1001349.

NSPCC (undated) Seeing and Hearing the Child. London: NSPCC Publications.

NSPCC (undated) *Children's Voices: Living with Parental Substance Misuse*. London: NSPCC Publications.

Ofsted (2012) Right on Time: Exploring Delays in Adoption. Manchester: OFSTED

Ogles, B.M., Lambert, M.J. and Fields, S.A. (2002) *Essentials of Outcome Assessment*. New York: John Wiley and Sons Inc.

Olds, D. L. (2006) 'The nurse–family partnership: An evidence-based preventive intervention.' *Infant Mental Health Journal 27*, 5–25.

Olds, D.L., Henderson Jr, C.R., Chamberlin, R. and Tatelbaum, R. (1986) 'Preventing child abuse and neglect: a randomized trial of nurse home visitation.' Pediatrics 78, 1, 65-78.

Olds, D., Henderson Jr., C.R., Cole, R., Eckenrode, J., Kitzman, H., Luckey, D., Pettitt, L., Sidora, K., Morris, P. and Powers, J. (1998) 'Long-term effects of nurse home visitation on children's criminal and antisocial behavior: fifteen-year follow-up of a randomized controlled trial.' Journal of American Medicine Association 280, 14, 1238-1244.

O'Reilly, R., Beale, B. and Gillies, D. (2010) 'Screening and intervention for domestic violence during pregnancy care: a systematic review.' *Trauma Violence Abuse 11*, 4, 190-201.

Orford, J., Copello, A., Velleman, R. and Templeton, L. (2010) 'Family members affected by a close relative's addiction: The stress-strain-coping-support model.' *Drugs: Education, Prevention and Policy 17*, 1, 36–43.

Orford, J., Templeton, L., Velleman, R. and Copello, A. (2005) 'Family members of relatives with alcohol, drug and gambling problems: a set of standardized questionnaires for assessing stress, coping and strain.' *Addiction 100,* 1611-1624.

O'Shaughnessy, R., Collins, C. and Fatimilehin, I. (2010) 'Building bridges in Liverpool: exploring the use of family group conferences for black and minority ethnic children and their families.' *British Journal of Social Work 40*, 7, 2034-2049.

O'Toole, P.T., Pollini, R.A., Ford, D. E. and Bigelow, G. (2006) 'Physical health as a motivator for substance abuse treatment among medically ill adults: is it enough to keep them in treatment?' *Journal of Substance Abuse Treatment 31*, 2, 143-150.

O'Toole, P.T., Pollini, R.A., Ford, D.E. and Bigelow, G. (2008) 'The health encounter as a treatable moment for homeless substance-using adults: the role of homelessness, health seeking behavior, readiness for behavior change and motivation for treatment.' *Addictive Behaviors* 33, 9, 1239-1243.

Oyserman, F., Mowbray, C.T., Allen-Meares, P. and Firminger, K. (2000) 'Parenting among mothers with a serious mental illness.' *American Journal of Orthopsychiatry 70*, 296–315.

Pajulo, M., Suchman, N., Kalland, M. and Mayes, L. (2006) 'Enhancing the effectiveness of residential treatment for substance abusing pregnant and parenting women: focus on maternal reflective functioning and mother-child relationship.' *Infant Mental Health Journal* 27, 5, 448-465.

Perez, C.M. and Widom, C.S. Childhood (1994) 'Victimization and long-term intellectual and academic outcomes.' *Child Abuse & Neglect 18*, 8, 617-633.

Peters, J. (2005) 'True ambivalence: child welfare workers' thoughts, feelings and beliefs about kinship foster care.' *Children and Youth Services Review 27*, 6, 595-614.

Peterson, S., Robinson, E. and Littman, I. (1983) 'Parent-child interaction training for parents with a history of mental retardation.' *Applied Research in Mental Retardation 4*, 329-42.

Pinto, S.M., Dodd, S., Walkinshaw, S.A. Siney, C., Kakkar, P. and Mousa, H.A.(2010) 'Substance abuse during pregnancy: effect on pregnancy outcomes.' *European Journal of Obstetrics and Gynecology and Reproductive Biology 150*, 137–141.

Platt, D. (2008) 'Care or control? The effects of investigations and initial assessments on the social worker-parent relationship.' *Journal of Social Work Practice: Psychotherapeutic Approaches in Health, Welfare and the Community* 22, 3, 301-315.

Platt, D. (2012) 'Understanding parental engagement with child welfare services: an integrated model.' *Child & Family Social Work 17*, 2, 138-148.

Pollak, S.D., Klorman, R., Thatcher, J.E. and Ciccetti, D. (2001) 'P3b reflects maltreated children's reactions to facial displays of emotion.' *Psychophysiology 38*, 267–274.

Porowskia, A.W., Burgdorfa, K., Herrell, J.M. (2004) 'Effectiveness and sustainability of residential substance abuse treatment programs for pregnant and parenting women.' *Evaluation and Program Planning* 27,191–198.

Porporino, F. J. and Robinson, D. (1995) 'An evaluation of the Reasoning and Rehabilitation programme with Canadian federal offenders.' In *Thinking Straight: The Reasoning and Rehabilitation Program for Delinquency Prevention and Offender*

Rehabilitation (R. R. Ross & R. D. Ross eds), Ontario, Canada: Air Training and Publications. pp. 155–192.

Povey, D. (2004) *Crime 2003: Supplementary Volume 1 – Homicide and Gun Crime.* London: Home Office.

Powis, B., Gossop, M., Bury, C., Payne, K. and Griffiths, P. (2000) 'Drug-using mothers: social, psychological and substance use problems of women opiate users with children.' *Drug and Alcohol Review 19*, 171–180.

Prinz, R.J., Sanders, M.R., Shapiro, C.J., Whitaker, D.J. and Lutzker, J.R. (2009) 'Population—based prevention of child maltreatment: the US Triple P system population trial.' *Prevention Science 10*, 1, 1-12.

Prochaska, J.O. and DiClemente, C.C. (1982) 'Trans-theoretical therapy: toward a more integrative model of change.' *Psychotherapy: Theory, Research, and Practice* 19, 3, 276–288.

Prochaska, J.O. and DiClemente, C.C. (1983) 'Stages and processes of self-change of smoking: toward an integrative model of change.' *Journal of Consulting and Clinical Psychology 51*, 3, 390–395.

Prochaska, J.O., DiClemente, C.C. and Norcross, J.C. (1992) 'In search of how people change: applications to addictive behaviors.' *American Psychologist 47, 9,* 1102–1114.

Prochaska, J.M. and Prochaska, J.O. (2002) 'Trans-Theoretical Model Guidelines for families with child abuse and neglect.' In A.R Roberts and G.J. Greene (eds) *Social Workers' Desk Reference*. New York: Oxford University Press.

Prochaska, J., Velicer, W., Rossi, J., Goldstein, M., Marcus, B., Rakowski, W., Fiore, C., HJarlow, L., Redding, C., Rosenbloom, D. and Rossi, S. (1994) 'Stages of change and decisional balance for 12 problem behaviours.' *Health Psychology* 13, 39-46.

Puura, K., Davis, H., Mantymaa, M., Tamminen, T. *et al.* (2005) 'The outcome of the European Early Promotion Project: Mother – Child Interaction.' *International Journal of Mental Health Promotion* 7, 1, 82-94.

Qureshi, T., Berridge, D. and Wenman, H. (2000) Family Support for South Asian Communities: A Case Study. London: NCB.

Raistrick, D., Heather, N. and Godfrey, C. (2006) *Review of the Effectiveness of Treatment for Alcohol Problems*. London: National Treatment Agency for Substance Misuse.

Randolph, K.A., Fincham, F. and Radey, M. (2009) 'A framework for engaging parents in prevention.' *Journal of Family Social Work 12*, 1, 56-72.

Redelmeier, D. A. and Shafir, E. (1995) 'Medical decision making in situations that offer multiple alternatives.' *JAMA 273*, 302–305.

Reder, P., Duncan, S. and Lucey, C. (2003) *Studies in the Assessment of Parenting.* East Sussex: Routledge.

Redmond, B. and Richardson, V. (2003) 'Just getting on with it: exploring the service needs of mothers who care for young children with severe/profound and life-threatening intellectual disability.' *Journal of Applied Research in Intellectual Disabilities* 16, 205-218.

Rhule, D. (2005) 'Take care to do no harm': harmful interventions for youth problem behaviour.' *Professional Psychology 36*, 6, 618–625.

Richards, L. (2004) *Getting Away With It: A Strategic Overview of Domestic Violence, Sexual Assault and Serious Incident Analysis.* London: Metropolitan Police.

Riggs, S.A. and Kaminski, P. (2010) 'Childhood emotional abuse, adult attachment, and depression as predictors of relational adjustment and psychological aggression.' *Journal of Aggression, Maltreatment & Trauma 19*, 1, 75-104.

Romer, D. (2010) 'Adolescent risk taking, impulsivity, and brain development: implications for prevention.' *Developmental Psychobiology 52*, 263–276.

Ross, R. R., Fabiano, E. A. and Ewles, C. D. (1988) 'Reasoning and rehabilitation. *International Journal of Offender Therapy and Comparative Criminology 32*, 29–35.

Rossman, B.B.R. and Ho, J. (2008) 'Posttraumatic response and children exposed to parental violence.' *Journal of Aggression, Maltreatment and Trauma 3*, 1, 85 – 106.

Rushton, A. (2003) 'Support for adoptive families: a review of current evidence on problems, needs and effectiveness.' *Adoption and Fostering 27*, 3, 41-50.

Rutter, M., Beckett, C., Castle, J., Colvert, E., Kreppner, J. Mehta, M., Stevens, S. and Sonuga-Barke, E. (2007) 'Effects of profound early institutional deprivation: An overview of findings from a UK longitudinal study of Romanian adoptees.' *Europoean Journal of Developmental Psychology 4*, 3, 332 -350.

Saint-Jacques, M.C., Drapeau, S., Lessard, G. and Beaudoin, A. (2006) 'Parent involvement practices in child protection: a matter of know-how and attitude.' *Child and Adolescent Social Work Journal* 23, 2, 196-215.

Sanders, M.R., Kirby, J.N., Tellegen, C. L. and Day, J.J. (2014) 'The Triple P-Positive Parenting Program: A systematic review and meta-analysis of a multi-level system of parenting support.' *Clinical Psychology Review 34, 337-357.*

Sanders, M.R., Pidgeon, A.M., Gravestock, F., Connors, M.D., Brown, S. and Young, R.W. (2004) 'Does parental attributional retraining and anger management enhance the

effects of the Triple-P Positive Parenting Programme with parents at risk of child maltreatment?' *Behavioural Therapy* 3, 5, 513 – 535.

Sanders, M.R., Ralph, A., Sofronoff, K., Gardiner, P., Thompson, R., Dwyer, S. and Bidwell, K. (2008) 'Every family: a population approach to reducing behavioral and emotional problems in children making the transition to school.' *Journal of Primary Prevention* 29, 197-222.

Scaramella, L. V., Conger, R. D., Spoth, R. and Simons, R. L. (2002) 'Evaluation of a social contextual model of delinquency: a cross-study replication.' *Child Development 73*, 175–195.

Scott, S., Spender, Q., Doolan, M., Jacobs, B. and Aspland, H. (2001) *Multicentre Controlled Trial of Parenting Groups for Childhood Antisocial Behaviour in Clinical Practice*. London: Institute of Psychiatry, King's College London,

Scott, K.L. (2004) 'Predictors of change among male batterers: application of theories and review of empirical findings.' *Trauma, Violence Abuse 5*, 3, 260-284.

Scott, C.K., Foss, M.A. and Dennis, M.L. (2005) 'Pathways in the relapse—treatment—recovery cycle over 3 years.' *Journal of Substance Abuse Treatment 28*, 63–72.

Scourfield, J. (2006) 'The challenge of engaging fathers in the child protection process.' *Critical Social Policy 26*, 440-449.

Sedlak, A.J., Mettenburg, J., Basena, M., Petta, I., McPherson, K., Greene, A. and Li, S. (2010) Fourth National Incidence Study of Child Abuse and Neglect (NIS–4): Report to Congress. Washington, DC: U.S. Department of Health and Human Services, Administration for Children and Families.

Selwyn, J., Sturgess, W., Quinton, D. and Baxter, C. (2006) Costs and Outcomes of Non-Infant Adoptions. London: British Association for Adoption and Fostering.

Selwyn, J.T., Farmer, E., Meakings, S.J. and Vaisey, P. (2013) *The Poor Relations? Children and Informal Kinship Carers Speak Out: A Summary Research Report.* School for Policy Studies, University of Bristol.

Selwyn, J., Wijedasa, D. and Meakings, S. (2014) *Beyond the Adoption Order: Adoption Disruption and Families in Crisis.* London: Department for Education.

Sergeant, H. (2006) *Handle with Care: An Investigation into the Care System*. London: Centre for Policy Studies.

Shean, G. and Meyer, J. (2009) 'Symptoms of schizophrenia and social cognition.' *Psychiatry Research 170*, 2–3, 157–160.

Sheehan, K. A., Thakor, S. and Stewart, D.E. (2012) 'Turning points for perpetrators of intimate partner violence.' *Trauma, Violence, and Abuse 13*, 1, 30-40.

Shlonsky, A. and Wagner, D. (2005) 'The next step: Integrating actuarial risk assessment and clinical judgment into an evidence-based practice framework in CPS case management.' *Children and Youth Services Review 27*, 409-427.

Shonkoff, J.P. and Garner, A.S. and the Committee on Psychosocial Aspects of Child and Family Health, Committee on Early Childhood, Adoption, and Dependent Care, and Section on Developmental and Behavioural Pediatrics (2012) 'The lifelong effects of early childhood adversity and toxic stress.' *American Academy of Pediatrics* 129, e232-e246.

Sinclair, I., Baker, C., Lee, J. and Gibbs, I. (2007) *The Pursuit of Permanence: A Study of the English Care System.* London: Jessica Kingsley Publishers.

Skinner, M.L., Haggerty, K.P., Fleming, C.B., Catalano, R.F. and Gainey, R.R. (2010) 'Opiate-addicted parents in methadone treatment: Long-term recovery, health, and family' relationships.' *Journal of Addictive Diseases 30*, 1, 17-26.

Skuse, T. and Ward, H. (2003) *Outcomes for Looked After Children: Children's Views of Care and Accommodation*. Interim Report to the Department of Health, Loughborough University: Centre for Child and Family Research.

Slack, K.S. Holl, J.L., McDaniel, M., Yoo, J. and Bolger, K. (2004) 'Understanding the risks of child neglect: an exploration of poverty and parenting characteristics.' *Child Maltreatment 9*, 395-408.

Smith, M. (2004) 'Parental mental health: disruptions to parenting and outcomes for children.' *Child and Family Social Work* 9, 1, 3–11.

Smith, D.W. and Brodzinsky, D.M. (2002) 'Coping with birthparent loss in adopted children.' *Journal of Child Psychology and Psychiatry 43*, 2, 213–223.

Snow, M.G., Prochaska, J.O. and Rossi, J.S. (1994) 'Processes of change in Alcoholics Anonymous: maintenance factors in long-term sobriety.' *Journal of Studies on Alcohol 55*, 3, 362-371.

Sobell, L.C., Cunningham, J.A. and Sobell, M.B. (1996) 'Recovery from alcohol problems with and without treatment: prevalence in two population surveys.' *American Journal of Public Health 86*, 7, 966-972.

Somers, V. (2007) 'Schizophrenia: the impact of parental illness on children.' *British Journal of Social Work* 37, 8, 1319–1334.

Sperry, L. (2005) 'Case conceptualization: a strategy for incorporating individual, couple and family dynamics in the treatment process.' *The American Journal of Family Therapy* 33, 5, 353-364.

Spratt, T. and Callan, J. (2004) 'Parents' views on social work interventions in child welfare cases.' *British Journal of Social Work 34*, 2, 199–224.

Springer, K. W., Sheridan, J., Kuo, D. and Carnes, M. (2007) 'Long-term physical and mental health consequences of childhood physical abuse: Results from a large population-based sample of men and women.' *Child Abuse & Neglect 31*, 517-530.

Stallard, P., Norman, P., Huline-Dickens, S., Salter, E. and Cribb, J. (2004) 'The effects of parental mental illness upon children: a descriptive study of the views of parents and children.' *Clinical Child Psychology and Psychiatry 9*, 39-52.

Stanley, N., Cleaver, H. and Hart, D. (2009) 'The impact of domestic violence, parental mental health problems, substance misuse and learning disability on parenting capacity.' In J. Horwath (ed.) *The Child's World: The Comprehensive Guide to Assessing Children in Need.* 2nd edition. London: Jessica Kingsley Publishers.

Stanley, N., Graham-Kevan, N. and Borthwick, R. (2012a) 'Fathers and domestic violence: building motivation for change through perpetrator programmes.' *Child Abuse Review 21*, 4, 264-274.

Stanley, N., Miller, P. and Richardson Foster, H. (2012b) 'Engaging with children's and parents' perspectives on domestic violence.' *Child and Family Social Work 17*, 192–201.

Stanley, N., Miller, P., Richardson Foster, H. and Thomson, G. (2011) 'A stop-start response: social services' interventions with children and families notified following domestic violence incidents.' *British Journal of Social Work 41*, 2, 296-313.

Stanley, N., Penhale, B., Riordan, D., Barbour R. and Holden, S. (2003) *Child Protection and Mental Health Services: Inter-professional responses to the needs of mothers.* Bristol: The Policy Press.

Staples, J. (2007) Knowle West Family Group Conference Project: Evaluation Report. Bristol: Barnardo's/Knowle West Neighborhood Renewal.

Styron, T. and Janoff-Bulman, R. (1997) 'Childhood attachment and abuse: long-term effects on adult attachment, depression, and conflict resolution.' *Child Abuse & Neglect 21*, 10, 1015–1023.

Sudbery, J (2002) 'Key features of therapeutic social work: the use of relationship.' *Journal of Social Work Practice* 16, 2, 149-162.

Sullivan, C.M. and Bybee, D.I. (1999) 'Reducing violence using community-based advocacy for women with abusive partners.' *Journal of Consulting and Clinical Psychology* 67, 1, 43-53.

Sullivan, P. and Knutson, J. (2000) 'Maltreatment and disabilities: a population-based epidemiological study.' *Child Abuse & Neglect 24*, 10, 1257-1273.

Sundell, K. and Vinnerljung, B. (2004) 'Outcomes of family group conferencing in Sweden; a 3-year follow-up.' *Child Abuse & Neglect 28*, 3, 267-287.

Sutherland, S. (1992) Irrationality: The Enemy Within. London: Constable.

Swenson, C., Penman, J., Hengeller, S. and Rowland, M. (2010a) *Multi-Systemic Therapy for Child Abuse and Neglect*. Charleston, SC: Family Services Research Center, MUSC.

Swenson, C., Penman, J., Hengeller, S., Fadowski, R. and Mayhew, A. (2010b) 'Multi-Systemic Therapy for Child Abuse and Neglect: A randomised controlled effectiveness trial.' *Journal of Family Psychology 24*, 4, 497-507.

Sword, W., Jack, S., Niccols, A., Milligan, K., Henderson, J. and Thabane, L. (2009) 'Integrated programs for women with substance use issues and their children: a qualitative meta-synthesis of processes and outcomes.' *Harm Reduction Journal* 6, 32.

Tarullo, A.R. and Gunnar, M.R. (2006) 'Child maltreatment and the developing HPA axis: hormones and behaviour.' *Science Direct 50*, 4, 632-639.

Taussig, H., Clyman, R. and Landsverk, J. (2001) 'Children who return home from foster care: a 6-year prospective study of behavioural health outcomes in adolescence.' *Journal of American Academy of Pediatrics* 7, 5, 419–40.

Taylor, A., Toner, P., Templeton, L. and Velleman, R. (2008) 'Parental alcohol misuse in complex families: the implications for engagement.' *British Journal of Social Work 38*, 5, 843-864.

Tellegen, C. and Sanders, M. (2013) 'Stepping stones: Triple P- Positive Parenting Program for children with disability: a systematic review and meta-analysis.' *Research in Developmental Disabilities* 31, 5, 1556-1571.

Thoburn, J. (2002) 'Adoption and Permanence for Children who Cannot Live Safely with Birth Parents or Relatives.' Quality Protects Research Briefing 5: Department of Health/Making Research Count/Research in Practice.

Thoburn, J., Chand, A. and Procter, J. (2005) *Child Welfare Services for Minority Ethnic Families: The Research Reviewed.* London: Jessica Kingsley Publishers.

Thomas, R. and Zimmer-Gembeck, M.J. (2007) 'Behavioral outcomes of Parent-Child Interaction Therapy and Triple P—Positive Parenting Program: a review and meta-analysis.' *Journal of Abnormal Child Psychology* 35,475–495.

Thomas, R. and Zimmer-Gembeck, M.J. (2011) 'Accumulating evidence for Parent–Child Interaction Therapy in the prevention of child maltreatment.' *Child Development 82*, 1, 177 – 192.

Titcomb. A. and LeCroy, C. (2005) 'Outcomes of Arizona's family group decision making program.' *Protecting Children 19, 4, 47–53.*

Tober, G. and Raistrick, D. (eds.) (2007) *Motivational Dialogue: Preparing Addiction Professionals for Motivational Interviewing Practice*. London: Routledge.

Tompsett, H., Ashworth, M., Atkins, C., Bell, L., Gallagher, A., Morgan, M. and Wainwright, P. (2009) *The Child, the Family and the GP: Tensions and Conflicts of Interest in Safeguarding Children.* DCSF-RBX-09-05-ES. London: Department for Children, Schools and Families.

Trevithick, P. (2003) 'Effective relationship-based practice: a theoretical exploration.' *Journal of Social Work Practice* 17, 2, 163-76.

Trickett, P.K. and McBride-Chang, C. (1995) 'The developmental impact of different forms of child abuse and neglect.' *Developmental Review* 15, 311-337.

Tronick, E.Z. (2007) *The Neuro-behavioural and Socio-Emotional Development of Infants and Children.* London: WW Norton and Company.

Tuck, V. (2004) 'Analysing risk in child protection: A model for assessment.' In V. White and J. Harris (eds.) *Developing Good Practice in Children's Services*. London: Jessica Kingsley Publishers.

Turnell, A. and Edwards, S. (1997) 'Aspiring to partnership: the signs of safety approach to child protection.' *Child Abuse Review 6*, 3, 179-190.

Turney, D., Platt, D., Selwyn, J. and Farmer, E. (2011) *Social Work Assessment of Children in Need: What do we Know?Messages from research.* London: Jessica Kingsley Publishers.

Tversky, A.and Kahneman, D. (1973) 'Availability: A heuristic for judging frequency and probability.' *Cognitive Psychology 5*, 207-232.

UKATT Research Team (2005a) 'Effectiveness of treatment for alcohol problems: findings of the randomised UK alcohol treatment trial.' *BMJ* 331, 541.

UKATT Research Team (2005b) 'Cost effectiveness of treatment for alcohol problems: findings of the randomised UK alcohol treatment trial.' *BMJ 331, 544.*

Ulman, A. and Straus, M.A. (2003) 'Violence by children against mothers in relation to violence between parents and corporal punishment by parents.' *Journal of Comparative Family Studies 34*, 41-60.

Uman, L.S. (2011) 'Systematic reviews and meta-analyses.' *Journal of the Canadian Academy of Child and Adolescent Psychiatry 20*, 1, 57-59.

Van den Dries, L., Juffer, F., Van IJzendoorn, M.H. and Bakermans-Kranenburg, M.J. (2009) 'Fostering security? A meta-analysis of attachment in adopted children.' *Children and Youth Services Review 31*, 3, 410-421.

Van de Mark, N.R. (2007) 'Policy on reintegration of women with histories of substance abuse: A mixed methods study of predictors of relapse and facilitators of recovery.' *Substance Abuse Treatment* 2, 28.

Van IJzendoorn, M.H., Schuengel, C. and Bakermans-Kranenburg, M.J. (1999) 'Disorganized attachment in early childhood: meta-analysis of precursors, concomitants, and sequelae.' *Development and Psychopathology 11*, 2, 225-249.

Van Voorhis, P. V., Spruance, L. M., Ritchey, P. N., *et al.* (2004) 'The Georgia cognitive skills experiment: a replication of Reasoning and Rehabilitation.' *Criminal Justice and Behavior 31*, 282–305.

Velleman, R., Templeton, L., Reuber, D., Klein, M. and Moesgen, D. (2008) 'Domestic abuse experienced by young people living in families with alcohol problems: results from a cross-European study.' *Child Abuse Review 17*, 387-409.

Wade, J., Biehal, N., Farrelly, N. and Sinclair, I. (2011) *Caring for Abused and Neglected Children: Making the Right Decisions for Reunification or Long-Term Care.* London: Jessica Kingsley Publishers.

Ward, H. (2009) 'Patterns of instability: Moves within the English care system, their reasons, contexts and consequences.' *Child and Youth Services Review, 31, pp 1113-1118*

Ward, H., Brown, R. and Maskell-Graham, D. (2012) *Young Children Suffering, or Likely to Suffer Significant Harm: Experiences on Entering Education*. London: Department for Education.

Ward, H., Brown, R. and Westlake, D. (2012) *Safeguarding Babies and Very Young Children from Abuse and Neglect*. London: Jessica Kingsley Publishers.

Ward, H., Holmes, L., Soper, J. (2008) *Costs and Consequences of Placing Children in Care*. London: Jessica Kingsley Publishers.

Ward, H., Munro, E.R. and Dearden, C. (2006) *Babies and Young Children in Care: Life Pathways, Decision-making and Practice*. London: Jessica Kingsley Publishers.

Ward, T., Day, A., Howells, K. and Birgden, A. (2004) 'The multifactor offender readiness model.' *Aggression and Violent Behaviour 9*, 6, 645-673.

Wathen, C. and MacMillan, H. (2003) 'Interventions for violence against women: scientific review.' *The Journal of the American Medical Association 289*, 5, 589-600.

Webster-Stratton, C., Rinaldi, J. and Reid, J.M (2011) 'Long-term outcomes of Incredible Years Parenting Program: predictors of adolescent adjustment.' *Child and Adolescent Mental Health 16*,1, 38–46.

West, R. (2011) Theory of Addiction. Blackwell Publishing Limited: Oxford.

Wexler, D.B. (1993) 'Therapeutic jurisprudence and changing conceptions of legal scholarship.' *Behavioral Sciences and the Law 11*, 1, 17-29.

White, O. G., Hindley, N. and Jones, D.P.H (in press) 'Risk factors for child maltreatment recurrence: an updated systematic review.' *Medicine, Science and the Law.*

Whitelaw, S., Baldwin, S., Bunton, R. and Flynn, D. (2000) 'The status of evidence and outcomes in stages of change research.' *Health Education Research 15*, 6, 707–718.

Widom, C.S., Czaja, S.J. and Dutton, M.A. (2008) 'Childhood victimization and lifetime revictimization.' *Child Abuse & Neglect 32*, 8, 785–796.

Wiggins, M., Austerberry, A. and Ward, H. (2012) *Implementing Evidence-Based Programmes in Children's Services: Key Issues for Success*. (DFE-RR245). London: Department for Education.

Wilkinson, R. and Pickett, K. (2010) *The Spirit Level: Why Equality is Better for Everyone*. London: Penguin.

Wilson, P., Rush, R., Hussey, S., Puckering, C., Sim, F., Allely, C.S., Doku, P., McConnachie, A. and Gillberg, C. (2012) 'How evidence-based is an 'evidence-based parenting program'? A PRISMA systematic review and meta-analysis of Triple P. *BMC Medicine 10*, 130.

Winokur, M., Holtan, A. and Valentine, D. (2009) 'Kinship care for the safety, permanency, and well-being of children removed from the home for maltreatment.' *Campbell Systematic Reviews 2009,1.*

Woodcock J. and Sheppard M. (2002) 'Double trouble: Maternal depression and alcohol dependence as combined factors in child and family social work.' *Children and Society* 2002, 16, 232-45.

Worcel, S. D., Furrer, C.J., Green, B.L., Burrus, S. W. M. and Finigan, M.W. (2008) 'Effects of family treatment drug courts on substance abuse and child welfare outcomes.' *Child Abuse Review 17*, 427–443.

Yatchmenoff, D.K. (2005) 'Measuring client engagement from the client's perspective in nonvoluntary child protective services.' *Research on Social Work Practice 15*, 2, 84-96.

Zalmanowitz, S.J., Babins-Wagner, R., Rodgers, S., Corbett, B.A. and Leschied, A. (2013) 'The association of readiness to change and motivational interviewing with treatment outcomes in males involved in domestic violence group therapy.' *Journal of Interpersonal Violence* 28, 5, 956-974.

List of Cases

L (Children) [2006] EWCA Civ 1282

Re BS (Children) [2013] EWCA Civ 1146

R (on the application of L and others) v. Manchester City Council; R (on the application of R and another) v. Same [2001] EWHC 707, CO/3954/2000, CO/965/2001.

Re L (Care: Threshold Criteria) [2007] 1 FLR 2050



© Loughborough University

Reference: DfE RR 369

ISBN: 978-1-78105-396-6

The views expressed in this report are the authors' and do not necessarily reflect those of the Department of Education

Any enquiries regarding this publication should be sent to us at: richard.white@education.gsi.gov.uk or www.education.gov.uk/contactus

This document is available for download at www.gov.uk/government/publications